INTERNET GOLD – GOLDEN LINES LTD. 2 Dov Friedman Street Ramat Gan 5250301, Israel

SUPPLEMENT TO PROXY STATEMENT
FOR

2019 ANNUAL AND EXTRAORDINARY GENERAL MEETING OF SHAREHOLDERS

TO BE HELD ON AUGUST 8, 2019

This Proxy Statement Supplement (the "Supplement") supplements and amends the original proxy statement of Internet Gold – Golden Lines Ltd. (the "Company," we," or "our"), dated July 4, 2019 (the "Proxy Statement") for the Company's 2019 Annual and Extraordinary Meeting of Shareholders (the "Meeting") to (i) add a new Proposal to the Proxy Statement that provides for approval of an arrangement pursuant to Section 350 of the Israeli Companies Law, and (ii) update the Notice of Annual and Extraordinary General Meeting to add the new Proposal (the "Updated Notice"). As previously disclosed in the Proxy Statement, the Meeting will be held at 10:30 a.m. on Thursday, August 8, 2019, at our offices at 2 Dov Friedman Street, Ramat Gan 5250301.

This Supplement relates to the new Proposal to be considered by shareholders at the Meeting and does not provide all of the information that is important to your decisions with respect to voting on all of the proposals that are being presented to shareholders for their vote at the Meeting. Additional information is contained in the Proxy Statement. To the extent that the information in this Supplement differs from, updates or conflicts with the information contained in the Proxy Statement, the information in this Supplement shall amend and supersede the information in the Proxy Statement. Except as so amended or superseded, all information set forth in the Proxy Statement remains unchanged and important for you to review. Accordingly, we urge you to read this Supplement carefully and in its entirety together with the Proxy Statement.

This Supplement relates to the solicitation of proxies by our Board of Directors (the "Board") for use at the Meeting. This Proxy Statement Supplement, the Updated Notice attached hereto as Appendix A and a revised proxy card (the "Revised Proxy Card") are being made available on or about July 16, 2019 to all stockholders entitled to vote at the Meeting.

We urge shareholders of record to vote on the proposals by submitting a Revised Proxy Card.

If you have already voted and do not submit a Revised Proxy Card, your previously submitted proxy will be voted at the Meeting but will not be counted in determining the outcome of the new Proposal.

PLEASE NOTE THAT IF YOU SUBMIT A REVISED PROXY CARD IT WILL REVOKE ALL PROXY CARDS PREVIOUSLY SUBMITTED, SO IT IS IMPORTANT TO INDICATE YOUR VOTE ON EACH PROPOSAL ON THE REVISED PROXY CARD.

PROPOSALS TO BE VOTED UPON BY SHAREHOLDERS

Information contained in this Supplement relates to the new Proposal (item 4 on the Revised Proxy Card) that will be presented to shareholders at the Meeting. Information regarding Proposals 1, 2 and 3 that will be presented to shareholders at the Meeting can be found in the Proxy Statement as originally filed with the SEC on July 5, 2019.

Please note that we strongly encourage you to read this Supplement and to vote on the new Proposal (item 4 on the Revised Proxy Card).

Vote Required

Pursuant to Israeli law requirements, a majority of the participants in the vote, other than abstentions, who jointly hold three fourths of the shares represented at the vote, is required to approve the Arrangement.

Pursuant to the provisions of Section 350 (13) of the Israeli Companies Law, the Court may impose a debt arrangement even in the event that the majority is not obtained in one of the final votes of the creditors or shareholders' meetings, subject to the Court's discretion and the terms set under Section 350 (13) of the Israeli Companies Law.

Cautionary Statement Concerning Forward-Looking Statements

Except for the historical information contained in this Proxy Statement, the statements contained in this Proxy Statement are "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933, as amended, or the Securities Act, and Section 21E of the Securities Exchange Act of 1934, as amended, or the Exchange Act, and the Private Securities Litigation Reform Act of 1995, as amended, with respect to our business, financial condition and results of operations. Such forward-looking statements reflect our current view with respect to future events and financial results. We urge you to consider that statements which use the terms "anticipate," "believe," "do not believe," "expect," "plan," "intend," "estimate," "anticipate" and similar expressions are intended to identify forward-looking statements. We remind readers that forward-looking statements are merely predictions and therefore inherently subject to uncertainties and other factors and involve known and unknown risks that could cause the actual results, performance, levels of activity, or our achievements, or industry results, to be materially different from any future results, performance, levels of activity, or our achievements expressed or implied by such forward-looking statements. Readers are cautioned not to place undue reliance on these forward-looking statements, which speak only as of the date hereof. Except as required by applicable law, including the securities laws of the United States, we undertake no obligation to publicly release any update or revision to any forward-looking statements to reflect new information, future events or circumstances, or otherwise after the date hereof. We have attempted to identify significant uncertainties and other factors affecting forward-looking statements in the "Risk Factors" section that appears below.

NEW PROPOSAL (item 4 on the Revised Proxy Card)

The description in this Proxy Statement of the Arrangement (as defined below) and the Searchlight Transaction (as defined below) is subject to, and is qualified in its entirety by reference to, the Arrangement documents

including the share purchase agreement (the "SPA"). We have attached a copy of a translation of the full Arrangement (as filed with the Court) to this Proxy Statement as <u>Exhibit A</u> and we recommend that you read it carefully in its entirety. Our shareholders are not third-party beneficiaries under the Arrangement and should not rely on the representations and warranties or any descriptions thereof or on the covenants contained in the Arrangement, the SPA signed with the Purchasers or the descriptions thereof as reflections of the actual condition of the Purchasers or us or as an undertaking towards them. The Arrangement was filed with the Israeli court in the Hebrew language. In the event of any conflict between the Hebrew and the English translation on the meaning or interpretation of a word, phrase or, the Hebrew language version shall prevail.

In the event of any conflict between this proxy statement and the full Arrangement and the SPA, the Arrangement and the SPA (and any other provisions of the Arrangement documents) shall prevail. You should not rely solely on the description in this proxy and the shareholders are being asked to go through and read the full Arrangement documents attached before voting. The following description is presented for convenience and does not constitute a full and binding description.

Background

For more than a year, our management and Board of Directors have contemplated and reviewed various acquisition opportunities in an attempt to improve our financial condition. The Company has a substantial amount of existing debt, restricting our operating and financing flexibility.

On January 16, 2019, the Company announced that its Board of Directors resolved to open an immediate dialogue with the Company's debenture holders (Series C and D) and that considering the liquidity balances of the Company and the need to maintain the value of its assets, the Company intends to withhold any payments to its debenture holders until further notice.

As indicated in the Company's announcement dated January 16, 2019, the Company's Board of Directors' resolution was obtained in light of the outcome of the competitive sale process conducted by the Company for the sale of its holdings in BCOM in which the Company did not receive any binding offers to purchase its holdings in BCOM and the fact that the sale of the Company's holdings in BCOM at the price reflected in an indication received by the Company by the final submission date, would not have enabled the full redemption of the Company's liabilities to its creditors.

The Company's Board continued to explore the possibility of selling its holdings in BCOM including discussions with the potential bidders. During the last few months, the Company received offers from Searchlight Capital Partners, L.P. ("Searchlight") to purchase its holdings in BCOM as well as an offer from TRYMG International Communications Ltd. (formerly Zeevi International Communications Ltd.) to invest capital in the Company in consideration for 100% of the Company's share capital.

The discussions were affected, *among else* by the business and financial condition of Bezeq – the Israel Telecommunication Company Ltd. ("**Bezeq**") and the fact that on March 20, 2019, BCOM announced that it intends to withhold any payments to its debenture holders until further notice.

On June 24, 2019 we announced that we have signed a definitive agreement (an "SPA") with a purchaser group led by Searchlight and the Fuhrer Family (together with Searchlight, the "Purchasers" and the "Searchlight Transaction"). The SPA was filed as Exhibit 99.1 to a Form 6-K filed by the Company on EDGAR dated June 21, 2019 and is part of the Arrangement documents. It is attached as Exhibit B hereto.

The transaction will include, *among else*, the sale by the Company of its entire holdings in BCOM for an aggregate amount of NIS 225 million and an investment by the Company in BCOM of NIS 345 million (comprised of the amount to be paid by the Purchasers and an additional amount of NIS 120 million). Pursuant to the transaction, the Company will receive NIS 310 million par value of Series C debentures of BCOM as well as 8,383,234 ordinary shares of BCOM.

Upon the closing of the transaction, the Purchasers will also invest an additional amount of NIS 260 million in BCOM in consideration for 62,275,450 ordinary shares of BCOM. In addition, BCOM will offer its shareholders (other than the Company) the right to subscribe for additional ordinary shares in an aggregate amount of up to NIS 35 million (the "Aggregate Offering Amount"). Any portion of the Aggregate Offering Amount not subscribed by BCOM's shareholders shall be subscribed by Searchlight SPC.

BCOM will repay it Series B debentures in their entirety and will repay an amount of NIS 614 million to its Series C debentures (other than the Series C debentures issued to the Company upon closing of the Searchlight Transaction). Certain amendments of BCOM's Series C debentures will become effective and BCOM, including amendment to the financial covenants therein.

The SPA includes, *among else*, representations by the Company, including with respect of authorization to enter into the transaction, required consents, certain representations relating to BCOM and representations relating to undisclosed proceedings. The agreement further provides for certain covenants by the parties during the period between signing and closing, including, exclusivity, making the required filings with the Ministry of Communications, obtaining of a court approval and the approval of the shareholders of BCOM, as well as certain restricted actions by the Company, BCOM or Bezeq.

The SPA includes certain conditions to closing, including the receipt of a court approval, the issuance of a control permit to the Purchasers from the Ministry of Communications, antitrust approvals, absence of material adverse effect, designation of board members on behalf of the Purchasers on the boards of directors of BCOM and Bezeq, and minimum liquidity and maximum indebtedness of BCOM upon Closing. There is no certainty that the transaction will be consummated.

For full description of the Arrangement (translated to English for convenience only), including a copy of the SPA, shareholders are advised to review Exhibit A. The SPA is attached as Exhibit B.

If the Transaction is consummated, the Company will not be able to fully pay its debts, and the Company intends to initiate an additional creditors' arrangement which will be submitted to the applicable Israeli court pursuant to Section 350 of the Israeli Companies Law. The Company anticipates that pursuant to such arrangement, the existing shares of the Company will be nullified and that the creditors of the Company will receive, among else, 100% of the shares of the Company.

The Court Process

Under the terms of the Searchlight Transaction, one of the conditions to closing is the receipt of the Israeli court's approval pursuant to Section 350 of the Israeli Companies Law, 5759-1999 (the "Arrangement"). On July 9, 2019, the Tel-Aviv District Court (the "Court") (Case no. PRK: 21556-07-19) approved our petition and ordered the convening of meetings of the Company's creditors and Shareholders for the purpose of approving the Arrangement.

A copy of the Arrangement, as filed with the Court (translated to English for convenience only) is attached as Exhibit A.

Risk Factors

In addition to the other information included in this Proxy Statement, including the matters addressed under "Cautionary Statement Concerning Forward-Looking Statements," you should carefully consider the following risk factors in determining how to vote at the Meeting.

Failure to complete the Arrangement could negatively impact our share price, business, financial

condition, results of operations or prospects.

The Arrangement is subject to the satisfaction or waiver of certain closing conditions described above. No assurance can be given that each of the conditions will be satisfied. If the conditions are not satisfied or waived in a timely manner and the Arrangement is delayed, our debenture holders and creditors may bring the Company to immediate liquidation which may also result in delisting and cessation of operations.

We and BCOM are subject to the Control Permit for holding Bezeq shares. Failure to comply with this permit or other regulatory provisions relating to the Control Permit of Bezeq may result in the revocation of the Control Permit and our rights with respect to our Bezeq interest would be adversely impacted, which would materially and adversely affect our business and financial position.

Pursuant to the Communications Order, we were required to obtain the prior written consent of the Ministers of Communications and Finance, or the Ministers, in order to obtain a permit to acquire Bezeg.

According to the Control Permit, the control must be held by an "Israeli Party," as defined in the Communications Order. Currently, BCOM and we are deemed to be "Israeli Parties," so long BCOM and we are controlled by a citizen and resident of Israel. Persons with certain criminal convictions may also not hold a Control Permit.

There is no assurance that the Purchasers will be granted a new Control Permit and this is another condition precedent to the Arrangement.

Conflicts of interest may arise between BCOM, Purchasers and us that could be resolved in a manner unfavorable to us.

The interest of BCOM in the Arrangement may not align with the interest of the Purchasers and with our own interest in dealing with our respective debenture holders and shareholders. There is no guarantee that BCOM will approve the Arrangement.

Exclusivity period may prevent better prospects.

The SPA prohibits us from entering into, soliciting or engaging in negotiations with respect to acquisition proposals or other transactions. These provisions could discourage other companies from proposing alternative transactions that may be more favorable to our shareholders than the Arrangement.

If the Transaction is consummated, the Company will not be able to fully pay its debts, and the Company intends to initiate an additional creditors' arrangement which will be submitted to the applicable Israeli court pursuant to Section 350 of the Israeli Companies Law. The Company anticipates that pursuant to such arrangement, the existing shares of the Company will be nullified and that the creditors of the Company will receive, among else, 100% of the shares of the Company.

Recommendation of our Board of Directors

Our Board of Directors has reviewed and considered the terms and conditions of the Arrangement, including the SPA and unanimously approved the Arrangement, the SPA and the underlying transactions contemplated by the Arrangement. The Board of Directors took into consideration, *among else*, the fact that the Searchlight Transaction is the outcome of a competitive and comprehensive sale process conducted by the Company and that the transaction was approved by the Company's debenture holders (in a preliminary vote) which are the Company's most substantial creditor.

An English translation of the Arrangement is attached as <u>Exhibit A</u> to this Proxy Statement. The SPA was filed as Exhibit 99.1 to a Form 6-K filed by the Company on EDGAR dated June 21, 2019 and is attached as <u>Exhibit</u>

B hereto.

Required Vote

Pursuant to Israeli law requirements, a majority of the participants in the vote, other than abstentions, who jointly hold three fourths of the shares represented at the vote, is required to approve the Arrangement.

Pursuant to the provisions of Section 350 (13) of the Israeli Companies Law, the Court may impose a debt arrangement even in the event that the majority is not obtained in one of the final votes of the creditors or shareholders' meetings, subject to the Court's discretion and the terms set under Section 350 (13) of the Israeli Companies Law.

Proposed Resolution

"RESOLVED, that the Arrangement, attached as <u>Exhibit A</u>, including the Arrangement Documents (as defined in the Arrangement) and the other transactions contemplated by, or related to, the Arrangement, and they hereby are, approved and adopted in all respects."

The Board of Directors recommends a vote FOR the foregoing resolution.

IF YOU HAVE ALREADY VOTED, YOUR VOTES WILL BE COUNTED WITH RESPECT TO THE ELECTION OF DIRECTORS IN PROPOSAL 1, AND PROPOSALS 2 AND 3. HOWEVER, WE URGE YOU TO CAST YOUR VOTE ON ALL FOUR PROPOSALS USING THE REVISED PROXY CARD EVEN IF YOU HAVE PREVIOUSLY CAST YOUR VOTE ON PROPOSALS 1, 2 AND 3 AS DESCRIBED IN OUR PROXY STATEMENT DATED JULY 4, 2019.

Appendix A

UPDATED NOTICE OF 2019 ANNUAL AND EXTRAORDINARY GENERAL MEETING OF SHAREHOLDERS

Dear Shareholders:

We cordially invite you to the 2019 Annual and Extraordinary General Meeting of Shareholders to be held on Thursday, August 8, 2019 at 10:30 a.m. (Israel time) at our offices at 2 Dov Friedman Street, Ramat Gan 5250301, Israel. At the Meeting, shareholders will be asked to adopt the following resolutions, as further detailed in the attached proxy statement:

- 1. To elect two directors, to hold office until our next annual general meeting of shareholders, as separate vote for each director will be taken.
- 2. To authorize our Board of Directors to effect a reverse share split of the Company's ordinary shares at a ratio not to exceed one-for-one hundred and to approve related amendments to the Company's Memorandum and Articles of Association; and
- 3. To ratify and approve the reappointment of Somekh Chaikin, registered public accounting firm, a member of KPMG International, as our independent registered public accountants for the year ending December 31, 2019, and to authorize our Board of Directors and our audit committee (under their authority in accordance with the Israeli Companies Law), to fix the compensation of such independent registered public accountants in accordance with the volume and nature of their services.
- 4. To approve an arrangement pursuant to Section 350 of the Israeli Companies Law.

In addition, our auditor's report and consolidated financial statements for the year ended December 31, 2018 will be reviewed and discussed at the Meeting.

Shareholders of record at the close of business on July 9, 2019 are entitled to notice of and to vote at the meeting. You can vote either by mailing in your proxy or in person by attending the meeting. If voting by mail, the proxy must be received by our transfer agent or at our registered office in Israel at least forty-eight (48) hours prior to the appointed time of the meeting to be validly included in the tally of ordinary shares voted at the meeting. If you attend the meeting, you may vote in person and your proxy will not be used. Alternatively, shares held via a TASE member may be voted electronically via the ISA's electronic voting system, up to six hours before the time fixed for the Meeting. Shareholders should receive instructions about electronic voting from the TASE member through which they hold their shares. Detailed proxy voting instructions are provided both in the Proxy Statement and on the enclosed proxy card.

Sincerely,

Doron Turgeman Chief Executive Officer

July 16, 2019

SETTLEMENT PLAN

A. GENERAL

- 1. This document describes the main terms of the settlement proposal formulated by B Communications Ltd. ("BComm"), Searchlight II BZQ L.P. ("Searchlight") and T.N.R. Investments Ltd. (hereafter, each of them the "Purchasers"), Internet Gold Golden Lines Ltd. ("Internet Gold"), and in accordance with the decision of the debenture holders to support approval of the Settlement Documents as part of the settlement pursuant to section 350 of the Companies Law, as reflected in the Settlement Documents ("the Settlement Plan").
- 2. It should be clarified that we refer to the main points of the provisions of the settlement only, and that in any case of contradiction and/or incompatibility between the Settlement Plan or any other provisions in this motion, and the provisions set forth in the Settlement Documents, the provisions of the Settlement Documents (as defined below) shall prevail; according to the provisions of the deeds of trust for the Series C, D and E Debentures, commencing on the Closing Date (as defined below), as part of the relations between BComm and the holders of the Series C, D and E Debentures, in case of contradiction and/or incompatibility among the various Settlement Documents, the provisions of the Series C, D and E deeds of trust shall prevail; the Settlement Documents are attached as Appendix A to the Settlement Plan and will be brought in their entirety for approval of the meetings; reading this Settlement Plan is not a substitute for reading the referenced documents.¹
- 3. The Settlement Plan does not include reference to all the provisions of the Investment Agreement and the other Settlement Documents.
- 4. Furthermore, in everything relating to the deed of trust of Series C Debentures of BComm, this Settlement Plan relates only to material amendments that will be made in the amended deed of trust, which will replace the original deed of trust for the Series C Debentures and which will enter into force on the Closing Date (the "Amended Series C Deed of Trust").

B. <u>DEFINITIONS</u>

- 5. In the Settlement Plan, the following terms have the meaning ascribed to them below:
 - 5.1 "Series B Debentures" The Debentures (Series B) issued by BComm on September 21, 2010, including their amendments from time to time.
 - 5.2 "Series C Debentures" The Debentures (Series C) issued by BComm on September 18, 2016, January 16, 2017 and January 23, 2018.
 - 5.3 "Series D Debentures" The Debentures (Series D) that will be issued by BComm on the Closing Date in a total amount of NIS 58 million par value.
 - 5.4 "Series E Debentures" The Debentures (Series E) insofar as will be issued by BComm on the Closing Date for raising Additional Debt (as defined below), the issuance of which is subject to raising the entire amount of the Additional Debt.

General remark: Some of the component parts of the amended settlement proposal are under audit by the Stock Exchange and the Securities Authority, as well as the Ministry of Communications, and some adjustments may have to be made in order to comply with their demands.

- 5.5 "Bezeq" Bezeq Israel Communication Corp. Ltd.
- 5.6 "Court" Tel Aviv District Court.
- 5.7 "Internet Gold" Internet Gold Golden Lines Ltd.
- 5.8 "TASE" The Tel Aviv Stock Exchange Ltd.
- 5.9 The "Settlement" The Settlement among BComm, Internet Gold, the Purchasers, the Series B and C debenture holders and the shareholders of BComm, as set forth in this Settlement Plan.
- 5.10 "BComm" B Communications Ltd.
- 5.11 The "Law" The Companies Law, 1999.
- 5.12 The "Investment Agreement" The Investment Agreement signed among BComm, Internet Gold and the Purchasers on June 24, 2019, in the wording attached to this Settlement Plan as part of Appendix A.
- 5.13 The "Purchasers" Searchlight II BZQ L.P. ("Searchlight"), and T.N.R. Investments Ltd. (the "Israeli Entity"), pro rata, according to the relative part of the Israeli Entity that will be set by Searchlight at least 3 business days prior to the Closing Date, and which shall be not less than 13% and not more than 19% of the transaction.
- 5.14 The "BComm Companies" BComm, SP1 and SP2 (as defined below).
- 5.15 The "Closing Date" The date commencing 13 business days after fulfillment of all the conditions precedent set forth in the Investment Agreement, on which the actions and transactions pursuant to the provisions of the Settlement will be carried out.
- 5.16 The "Transaction Share Price" NIS 4.175 per BComm share.
- 5.17 The "Settlement Documents" The Investment Agreement, the deeds of trust of the Series C, D and E² Debentures, the Financing Commitments of the Purchasers, letters of Guarantee of the Purchasers and the Bezeq Share Lien Agreements, attached as <u>Appendix A</u> to this Settlement Plan.
- 5.18 "LTV" The ratio of (1) the balance of all the debts of BComm net of cash and cash equivalents of BComm in banking corporations and recognized financial institutions (including those which are limited or pledged and bank guarantees for securing debts) on the date of the relevant audit, to (2) the value of all the Bezeq shares owned by BComm, directly or indirectly, on that same audit date.
- 5.19 "SP1" B Communications (SP1) Ltd.
- 5.20 "SP2" B Communications (SP2) Ltd.
- 5.21 "Value of Bezeq Shares" means the adjusted average closing price of the Bezeq share in the 90 days ended on the date of the relevant audit.

Insofar as Series E Debentures will be issued.

C. THE TRANSACTION

6. Purchase of controlling shares by the Purchasers

The Purchasers will purchase from Internet Gold all of its holdings in BComm, accounting for 51.95% of the issued and outstanding capital of BComm, in consideration of NIS 225 million.

7. Investments in the Company

- 7.1 On the Closing Date, the Purchasers will invest in the Company the sum of NIS 260 million, against an allotment of 62,275,450 ordinary shares of BComm, at the Transaction Share Price.
- 7.2 BComm shares will be issued as described in section 8 below.
- 7.3 On the Closing Date, Internet Gold will be allotted Series C Debentures of NIS 310 million par value in consideration of an injection of NIS 310 million to the Company.
- 7.4 Similarly, in the event that the balance of cash in BComm will be an amount between NIS 695 million and a minimum of NIS 680 million ("the Minimum Balance"), additional shares will be issued to the Purchasers without additional payment in an amount equivalent to the difference between NIS 695 million and the Minimum Balance divided by the share price of the transaction.8. Capital Issue
 - 8.1 On the Closing Date, BComm will make a capital issue of NIS 70 million to the BComm shareholders, at the Transaction Share Price, as described in sections 8.2 and 8.3 below (the "Capital Issue").
 - 8.2 No later than 20 business days prior to the Closing Date, BComm will offer its shareholders (excluding Internet Gold) participation in half of the Capital Issue, i.e. NIS 35 million, at the Transaction Share Price. Searchlight has undertaken to purchase the balance of those shares, amounting up to NIS 35 million, if the other shareholders in the Company do not respond to the offer.
 - P.3 Internet Gold has undertaken to purchase, in the Capital issue, half of the Capital Issue, i.e. NIS 35 million, subject to Searchlight's right (but not its obligation) to purchase the shares instead of Internet Gold, if Internet Gold fails to meet its undertaking to purchase the shares as aforesaid. The issue will be made at the Transaction Share Price.

9. Repayment of Series B Debentures

On the Closing Date, BComm will repay in full those amounts of principal and interest (but without arrearage interest and/or penalties and/or other payments) still unrepaid in respect of BComm's Series B Debentures as of that repayment date.

10. Series C Debentures

10.1 Early repayment of principal and interest of the Series C Debentures

On the Closing Date, BComm will make the following payments only to those who were registered on the Closing Date as holders of the Series C Debentures, and for the removal of doubt, except in respect of the debentures that will be issued on the Closing Date to the controlling shareholder in the Company, Internet Gold:

(a) Payment on account of the interest on the Series C Debentures which has accrued up to the Closing Date and has not yet been paid (at an interest rate of 3.85%, without any additional interest and/or arrearage interest and/or penalties and/or other payments):³

It is clarified that in accordance with the provisions of the original deed of trust, the interest rate commencing July 4, 2019 is 4.85% (the "Interest Increase") but if the Settlement is completed, the rate of the interest accrued by the Closing Date and not yet paid will remain at 3.85%.

(b) BComm will make early repayment of NIS 614 million par value principal of the Series C Debentures in respect of principal payments scheduled to be made in the years 2020-2022 and in respect of partial repayment (65%) of the interest payment in 2023, The balance of the payment of the principal for 2023 will be replaced by Series D Debentures, which will be issued to the existing debenture holders on the Closing Date, as described in section 11 below.

10.2 The new loan schedule of the Series C Debentures after the Closing Date

The Series C Debentures will fall due (principal) in one payment (100%), to be paid on November 30, 2024.

Commencing on the Closing Date (if it takes place), the interest on the unrepaid balance, as may be from time to time, of the principal of the Series C Debentures will be paid commencing November 2019, and twice a year on May 31 and November 30 of each of the years 2020 to 2024.

The final principal and interest payments will be made on November 30, 2024.

10.3 Interest

The Series C Debentures will bear annual shekel interest of 3.85%.

For failure to comply with the LTV Ratio, the annual interest rate will increase by 0.25% in respect of the period commencing on the date of the second consecutive audit in which BComm failed to comply with the LTV Ratio, until the earlier of (1) the date of ending the deviation; (2) the date of full repayment of the balance of the unrepaid principal of the Series C Debentures, and all in accordance with the provisions of section 5.15 of the Amended Series C deed of trust.

10.4 Series expansion

On the Closing Date, BComm will allot Series C Debentures of NIS 310 million par value to Internet Gold by way of a series expansion. It is clarified that the series expansion will be implemented only after early repayment as described in section 10.1(b) above.

10.5 Raising financial debt and senior debt

BComm may not assume additional financial debt other than in one of the following cases: (a) the entire net debt consideration (less the expenses involved in raising the additional financial debt) is used for repayment of BComm's debts in respect of the Series C and D Debentures (and in such a case BComm's debt to the Series D Debentures will be repaid first, and the balance remaining, if any, after repayment of BComm's entire debt to the Series D Debentures will be used for repayment of BComm's debt to the Series C Debentures); (b) for generating liquidity for unforeseen operating expense needs of BComm (and it is clarified that this sum shall not be used for making a distribution to BComm's shareholders) (the "Additional Debt"), provided that the amount of the Additional Debt does not exceed the lower of (a) NIS 100 million (one hundred million shekels) and (b) 1.3% of the Value of Bezeq Shares as defined above.

⁴ The first interest payment on November 30 2019 will be made in respect of the period starting on the first trading day after the Closing Date and ending on that payment date, and will be calculated according to the number of days in that period and on the basis of 365 days in a year at the annual interest noted in section 10.3 below.

BComm will act to raise the Additional Debt by issuing Series E Debentures of BComm, which will be offered to the holders of Series C Debentures of BComm only (including after the series expansion for Internet Gold as described in section 7.3 above). If BComm does not receive orders for the entire amount of the Additional Debt from the issuance of the Series E Debentures, the issuance will be cancelled and BComm may implement a series expansion of Series C Debentures by means of a private placement to classified investors.

If BComm approached fifteen classified investors and did not receive orders for the entire amount of the Additional Debt from the aforementioned private placement, the series expansion of Series C Debentures will be cancelled and BComm may raise debt from time to time in an amount that does not exceed the amount of the Additional Debt from one or more third parties, including by means of issuing a new series of debentures on terms and at an interest rate at BComm's discretion and subject to provisions relating to creation of a senior lien as set forth in the amended Series C deed of trust and in section 10.5.

10.6 Amendment of the financial criteria and causes for immediate repayment

- (a) The financial criterion concerning the ratio of separate equity to total separate balance sheet, as in the Series C deed of trust, is voided on the Closing Date. Commencing 24 months after the Closing Date, a criterion relating to the LTV Ratio will come into force, whereby the LTV Ratio shall not exceed, during two consecutive quarters (1) 80% by November 30, 2023, and (2) 75% commencing December 1, 2023 and to the date of full and final repayment of the Series C Debentures (the "LTV Ratio").
- (b) In addition, immediate repayment events relating to compliance with equity of NIS 650 million, termination or lowering of the rating of the debentures have been deleted, and cause of immediate repayment in respect of registration of a going concern has been suspended for a period of 24 months from the Closing Date. Other immediate repayment events have been added, relating to breach of BComm's duties in the amended Series C deed of trust with regard to BComm's compliance with the LTV Ratio and assuming financial debt.

11. Issuance of Series D

- 11.1 On the Closing Date, BComm will allot Series D Debentures to whoever were registered as holders of Series C Debentures on the date of record for the Closing Date, and at the same time it will delete the same amount of NIS 58 million par value from the principal of the Series C Debentures
- 11.2 The Series D Debentures will be issued on terms that are mostly similar to the terms of the Series C Debentures, except for: (a) BComm's obligation for **full** early repayment if dividends are received from Bezeq in which 90% of the amount will suffice to cover the entire principal of the Series D Debentures; (b) if BComm issues new series of debentures or BComm sells any of its assets (including the encumbered shares), the proceeds of an issuance or the sale, as the case may be, shall be used first and foremost for early repayment of the Series D Debentures; and (c) no series expansion is permitted for the Series D Debentures.

12. <u>Issuance of Series E</u>

- 12.1 BComm will act to raise the Additional Debt first by way of an issuance of Series E Debentures of BComm as described in section 10.5 above.
- 12.2 The Series E Debentures will be issued on terms mostly similar to the terms of the Series C Debentures except for: (a) the date of the first interest payment, which will be on May 20, 2020; and (b) no series expansion is permitted for the Series E Debentures.
- 12.3 If BComm issues new series of debentures or BComm sells any of its assets (including the Pledged Bezeq Shares), the proceeds of an issuance or the sale, as the case may be, shall be used first for early repayment of the Series D Debentures. The proceeds remaining after full repayment of the Series D Debentures will be used for full repayment of the Series C Debentures, and the proceeds then remaining, if any, will be used for early repayment (full or partial) of the Series E Debentures. In the event of a sale of the Pledged Bezeq Shares, the balance of the sale will be used first and foremost for the early repayment of the Series E Debentures in full.
- 12.4 The Series E Debentures will be secured by a first and senior lien on the Pledged Bezeq Shares and the pledged shares account, and accordingly, the liens securing the Series C and D Debentures will become second and junior liens.

13. Lien on Bezeq shares

- 13.1 The Purchasers, BComm and Internet Gold, undertake to take all necessary actions and use reasonable commercial means in good faith, to obtain the approval of the Ministry of Communications and any other approval that may be required under the Bezeq license (the "Approvals"), for imposition of a first lien on the Bezeq shares as described below. It is clarified that non-receipt of approval for the lien does not constitute a breach of the Settlement Documents and/or cause for cancellation of the Settlement.
- 13.2 On the Closing Date, and subject to receipt of the Approvals, first liens will be provided for the Series C Debenture holders and for the Series D Debenture holders, pro rata, on all the Bezeq shares held directly or indirectly by BComm on the Closing Date (the "Pledges Bezeq Shares"), as well as on the rights in the Bezeq accounts through which the shares are held (the "Pledged Assets").

- 13.3 Grant of the liens is subject to receipt of the approval of the Ministry of Communications for providing the liens, and any other approval that may be required under the Bezeq license and in accordance with the Restrictions on the Liens and their Exercise (as defined below).
- 13.4 Until creation of the liens as described above, BComm has undertaken a negative pledge on all the Bezeq shares held by BComm and by SP2.
- 13.5 Creation and exercise of the liens is subject to the restrictions laid down in law concerning exercise of the liens, including restrictions laid down in the Communications Law and in the Communications (Telecommunications and Broadcasts) (Determination of an Essential Service Provided by Bezeq, BComm Israel Communications Ltd.) Order, 1997 (the "Communications Order"); restrictions set forth in the control permit and approval of the lien, including receipt of prior written approval of the entities named in the lien approval, for any transfer of means of control, as required under the Communications Law and the Communications Order; and restrictions set in the Bezeq license and in the additional licenses, including receipt of the prior written approval of the Minister for Communications or of the head of the Civil Administration in the Administered Territories (as the case may be), insofar as such approval may be required and as set forth in the additional licenses (the "Restrictions on the Liens and their Exercise").
- 13.6 Until the date of exercise of the Pledged Assets, BComm and SP2 may vote by virtue of the Pledges Bezeq Shares (and the bonus shares) at Bezeq meetings on any subject whatsoever, at their exclusive discretion.
- 13.7 The liens in favor of the debenture holders are subject to an option to create a senior lien, in a case where BComm raises the Additional Debt, all in accordance with the provisions set forth in the deeds of trust and as described in section 10.5 above. In such case, the liens referred to above will become second liens until repayment of the Additional Debt.
- 13.8 Any sum received in respect of exercise of the liens will be divided among the holders of the Series C and D Debentures, pro rata among them on account of the balance of BComm's debt to them.
- 13.9 If on the date of exercise of the Restricted Assets the Additional Debt exists and in respect thereof liens that are senior to the existing liens are provided for the Series C and D Debenture holders, the proceeds from such exercise will be used first for early redemption, full or partial, of the Additional Debt, and the balance remaining from the proceeds of the exercise will be used for early redemption, full or partial, of the Series C Debentures and the Series D Debentures, pro rata among them.

⁵ This means licenses that were or will be granted by the Ministry of Communications or the Civil Administration in the Administered Territories, to Bezeq and the companies it holds, except for the General License for the provision of domestic fixed-line telecommunications services that was granted to Bezeq.

- 14. In the event of sale of the Pledged Bezeq Shares, BComm will use the entire proceeds it receives from the sale of the sold asset, after taxes, expenses and deductions involved in such sale of the shares, for early redemption, full or partial, of the Series D Debentures in accordance with their terms, and the balance remaining, if any, will be used for early redemption, full or partial, of the Series C Debentures.
- 15. If on the date of sale of the Pledged Bezeq Shares the Additional Debt (as defined in section 10.5 above) exists, the proceeds from the sale in respect of sale of the Pledged Bezeq Shares will be used first for repayment of the Additional Debt in full, and thereafter, first for repayment of BComm's debt in respect of the Series D Debentures and the balance remaining, if any, for early redemption, full or partial, of the Series C Debentures.

16. Sale of the Pledged Bezeg Shares

- 16.1 BComm may sell Pledged Bezeq Shares only if on the date of the sale there is no cause for the Series C Debentures or the Series D Debentures or the Additional Debt, if it exists, being called for immediate repayment and also, after such sale BComm is in compliance with the LTV Ratio required in the deed of trust, if required, or alternatively, in the event of the sale of all the Restricted Bezeq Shares, the proceeds of the sale plus BComm's Available Cash are equal at least to the sum required for full repayment of BComm's debt in respect of the Additional Debt, if it exists, in respect of the Series D Debentures and in respect of the Series C Debentures in accordance with the provisions of the deeds of trust.
- 16.2 The proceeds from such sale of shares will be used first for early redemption, full or partial, of the Series D Debentures in accordance with their terms, and the balance remaining, if any after full repayment of BComm's debt in respect of the Series D Debentures, will be used for early redemption, full or partial, of the Series C Debentures.
- 16.3 Notwithstanding the foregoing in this section 16, in the event of sale of the Pledged Bezeq Shares, if Additional Debt exists on the Closing Date, the proceeds of the sale in respect of sale of the Pledged Bezeq Shares will be used first for repayment of the Additional Debt in full and thereafter, for early repayment in accordance with their terms of BComm's debt in respect of the Series D Debentures and after full repayment, for early redemption, full or partial, of the Series C Debentures as aforesaid.

17. Restrictions on distribution

BComm undertakes not to distribute a dividend to its shareholders and/or to buy back its shares and/or any other distribution as defined in the Companies Law, unless all of the terms listed below obtain:

- 17.1 On the date of a resolution for a distribution, there is no cause for immediate repayment and there is no such cause as a result of making the distribution.
- 17.2 The LTV does not exceed 65% immediately after making the distribution.
- 17.3 For a distribution to be made on a date after November 30, 2023 if the Series D Debentures have been repaid in full.
- 17.4 BComm holds, immediately after making the distribution, a cash balance equal to the amount of the upcoming interest payment in respect of the Series C and D Debentures and in respect of the Series E Debentures, if they exist (including by means of the interest cushion as described in the deeds of trust of the Series C, D and E Debentures).

18. Restrictions on management fees

The Purchasers undertake that as long as the Series C Debentures are in circulation, management fees will not be collected from BComm, except reimbursement of expenses and remuneration for its directors, in an amount that does not materially deviate from what is permissible under BComm's current compensation policy.

19. Conditions precedent

The Investment Agreement includes the following list of conditions precedent for closing the transactions (and all subject to and as set forth in the Agreement):

- 19.1 Receipt of a control permit from the Ministry of Communications for the Purchaser, as set forth in the Investment Agreement,
- 19.2 Unconditional approval of the Court that does not include conditions and is not subject to any further consent of any third party other than the Ministry of Communications, for transactions that are included in the Investment Agreement without amendments, changes or provisos, and exemption for the officers at Internet Gold and BComm in connection with any act or omission relating to the Investment Agreement (and the transactions included in it).

As part of the motion for approval of the Court for the Settlement, release will be sought of any lock-up restrictions (חַסימה) that might be applicable to the securities being issued by BComm under the Investment Agreement at the Closing and the Securities Authority will be added as a party to the motion for approval, and all without receipt of such approval being a condition for completion of the Settlement.

- 19.3 Approval of the Antitrust Commissioner for transactions included as part of the Settlement in accordance with the Restrictive Trade Practices Law, 1988 and its amendments from time to time, and any rules, regulations or guidelines promulgated thereunder.
- 19.4 Approval of the TASE for issuance of the securities issued as part of the Settlement, and including the relevant approval for the Series C Debentures in accordance with the terms of the amended Series C deed of trust, amendment of the Series C deed of trust as part of the Settlement.
- 19.5 No injunction The absence of any injunction or other order or other legal requirement which prevents, necessitates or restricts execution of the transactions included in the Settlement.

- 19.6 Replacement of the directors in the Company and Bezeq, as described in section 22 below.
- 19.7 No breach of the undertakings and/or representations of BComm and Internet Gold under the Investment Agreement and no Material Adverse Effect as defined in section 28 below.
- 19.8 No suspensions of trading, delisting, notices of suspensions of trading of BComm or Bezeq by the TASE or Nasdaq.
- 19.9 Bezeq shares held by BComm and the BComm Companies were not sold and at the Closing they will account for at least 25.1% of the issued and outstanding share capital of Bezeq.
- 19.10 BComm will hold the requisite cash balances in its treasury as part of the Settlement.
- 19.11 Approval of the amendments required to BComm's debentures as part of the Settlement.
- 19.12 The approval of the shareholders has been received or the approval of the Court has been received for the Settlement and the date for appeal of the approval has elapsed (and no appeal relating to the approval of the Court for the Settlement is pending).

20. Control permit

The Investment Agreement includes detailed provisions relating to the Purchasers filing a motion for receipt of a control permit for the Bezeq shares, and the Parties are obligated to make every effort in good faith and to act for the speedy filing of all requisite information.

The Purchasers will be ready to receive a control permit (on terms substantially the same as the existing control permit and will not make the control permit conditional upon additional or preferential terms to those contained in the existing permit. The Purchasers are not obligated to agree to any Burdensome Condition (as defined in the Investment Agreement, which is not included in the existing permit) on their activities, their businesses (or those of entities related to any of the units of the Purchasers) or the relations between the units of the Purchasers.

21. Liquidity and debt conditions

On the transaction Closing Date, BComm will fulfill the following undertakings:

- 21.1 The Available Cash at BComm (after Settlement expenses and without deduction of the costs of any legal proceeding settled with the prior consent of the Purchasers at their absolute discretion) exceeds the minimum balance. If the amounts exceed the minimum balance but are less than NIS 695 million, shares will be allotted to the Purchaser as set forth in the Investment Agreement; and
- 21.2 The amount of BComm's consolidated debt does not exceed NIS 2,466,048,333 par value; and
- 21.3 BComm, SP2 or SP1 have not changed the terms of their debt except as provided in the Investment Agreement.

22. Change in composition of directors in BComm and Bezeq

- 22.1 On the Closing Date and pursuant to the decision of the Purchasers, the tenure of directors in the Company will be terminated and new directors will be appointed in accordance with the wishes of the Purchasers and the requirements of the law insofar as concerns external directors, provided that they meet the qualification criteria laid down in the legal requirements applicable to members of the Board of Directors of BComm and subject to retention of the composition of the Board of Directors as required by the provisions of the law, and provided that the directors selected by the Purchasers constitute the majority of the members of the Board of Directors of BComm.
- 22.2 On the Closing Date, the tenure of 2 directors out of the 9 members of the Board of Directors of Bezeq will be terminated and new directors will be appointed in accordance with the wishes of the Purchasers and the requirements of the law insofar as concerns external directors, provided that they meet the qualification criteria laid down in the legal requirements applicable to members of the Board of Directors of Bezeq, and in any case at least two of the members of the Board of Directors of Bezeq will be those selected by the Purchasers.

23. Representations

- 23.1 The Agreement contains basic representations from all the Parties to the transaction, the breach of which could invoke compensation with limitations as set forth in the Investment Agreement and subject to its provisions.
- 23.2 The representations in the Investment Agreement expire within 4 months of the Closing Date.
- 23.3 As emphasized, the representations made as part of the Investment Agreement by BComm and/or the Controlling Shareholder and the liability for them, is not a joint and several liability.

24. Reliefs, compensation and limitations on compensation amounts

- 24.1 Except in a case of fraud, the compensation that the Parties to the Agreement might owe is limited to NIS 26 million for the Company, NIS 22.5 million for Internet Gold and NIS 30 million for the Purchasers, divided pro rata among the Purchasers according to the percentage of their part in the transaction.
- 24.2 Breach of undertakings or representations will also entitle the other Parties to the transaction to terminate the Agreement subject to its provisions.

25. Financial commitments and guarantee

For compliance with the undertakings of the Purchasers, each of the controlling shareholders in the Purchasers provided Financial Commitments relating to the provision of the part of each of them in the injection that the Purchasers must make on the occasion of Closing the transaction, and letters of guarantee relating to the Financial Commitments of the Purchasers if the Agreement is terminated.

The Financial Commitments and the guarantee are part of the Settlement Documents that were attached as an appendix to this Settlement Plan.

26. Provisions for the Interim Period

Commencing on the date of execution of the Agreement and until the Closing Date, each of the BComm Companies shall continue the normal course of its business, shall not transfer or encumber the Bezeq shares it owns. Furthermore, each of Internet Gold and the BComm Companies shall use (subject to duties of trust applicable under the applicable legal requirements) all its voting rights in the Company and in Bezeq,, respectively (and if such matters are brought to a shareholders' vote) so that the BComm Companies and Bezeq and its subsidiaries do not do, effect or allow to be done (or commit or agree to take, do, effect or allow) even one of the Restricted Matters as defined in the Investment Agreement.

The Restricted Matters include a list of actions, including grant or issue of capital rights in the companies, making a distribution, amendment of documents of association (except with regard to a capital increase in BComm, which was completed on June 30, 2019), structural changes and the like, settlement in any derivative claim or class action (provided that BComm may settle derivative claim 47621-07-16 Horev v. B Communications Ltd. et al, which is pending in the Tel Aviv District Court if this does not impose any liabilities or obligation on the BComm Companies as noted in the Investment Agreement), and more. In addition, during the Interim Period, Internet Gold shall not transfer or encumber the BComm shares it owns, and each of the BComm Companies shall not transfer or encumber the shares of the BComm Companies and Bezeq, respectively, that it owns.

27. Exclusivity

Commencing on the date of execution of the Agreement and until the Closing Date, Internet Gold and the BComm Companies shall not initiate, lobby for or negotiate any Acquisition Proposal relating to their securities or principal assets, and shall use reasonable efforts to ensure that their officers, employees, advisers or anyone acting on their behalf, do not act in such manner.

It is clarified that this undertaking does not apply to representatives of the debenture holders of Internet Gold and of BComm unless they are acting on the instruction of Internet Gold or BComm.

28. Material Adverse Effect

Subject to the provisions of the Investment Agreement, the Purchaser is entitled not to close the transaction if a Material Adverse Effect occurred on the business, on the situation (financial or other) or on the results of operations of the BComm Companies and of their subsidiaries (including Bezeq and its subsidiaries) together. The definition of Material Adverse Effect excludes a long list of events and circumstances that will not be cause for non-closure of the transaction, including general changes in the terms affecting the industries in which Bezeq and its subsidiaries operate, any change in the market price or trading volume of the ordinary shares or shares of BComm or of Bezeq on Nasdaq or the TASE (of themselves), and more.

A Material Adverse Effect event does not constitute (of itself) cause for a demand for compensation under the Investment Agreement.

29. Exemption from legal claims

Approval of the Settlement includes exemption and release from liability in relation to any act or omission concerning this Settlement and the transactions pursuant to it by the directors and officers of BComm and Internet Gold, and the holders of the debentures of BComm and Internet Gold, the trustees of the debenture holders, the representatives of the debenture holders, the financial advisers and legal advisers of the debenture holders, in each of the cases, in their capacity as directors and officers.

30. Term of the Settlement Plan

Non-fulfillment of the conditions precedent for the transaction in a period of up to five months from the date of execution of the Agreement, shall constitute cause for termination of the Agreement. Furthermore, if the approval of the Court as described in section 19.2 above is not received within 60 days of the date of execution of the Investment Agreement, Searchlight will be entitled to bring the Agreement to an end.

* * *

SHARE PURCHASE AGREEMENT

by and among:

INTERNET GOLD – GOLDEN LINES LTD. AS SELLER

B COMMUNICATIONS LTD.
AS COMPANY

and

SEARCHLIGHT II BZQ, L.P. AS PURCHASER A

T.N.R. INVESTMENTS LTD. AS PURCHASER B

Dated as of June 24, 2019

Table of Contents

			Page			
1.	DEFINITIO	NS				
2.	SALE AND PURCHASE OF SHARES; SUBSCRIPTION; CONTRI					
3.	REPRESENTATIONS AND WARRANTIES OF THE SELLER AND T					
4.	REPRESEN	TATI	ONS AND WARRANTIES OF EACH PURCHASER 18			
5.	COVENANT	ΓS				
6.	CONDITIO	NS PF	RECEDENT28			
7.	CLOSING	•••••	31			
8.	TERMINAT	ION.				
9.	MISCELLA	NEO	US35			
	<u>EXHIBITS</u>					
	Exhibit A-1	_	Commitment Letter			
	Exhibit A-2	_	Commitment Letter			
	Exhibit A-3	-	Sponsor Guarantee			
	Exhibit A-4	-	Sponsor Guarantee			
	Exhibit B	_	Company Debt Modifications			

SHARE PURCHASE AGREEMENT

This Share Purchase Agreement (this "<u>Agreement</u>") is entered into as of June 24, 2019, by and among **Internet Gold – Golden Lines Ltd.**, a company organized under the laws of the State of Israel, having its registered office at 2 Dov Friedman St., Ramat Gan, Israel (the "<u>Seller</u>"); **B** Communications Ltd., a company organized and existing under the laws of the State of Israel, with registration number 51-283274-2, having its registered office at 2 Dov Friedman St., Ramat Gan, Israel (the "<u>Company</u>"); **Searchlight II BZQ, L.P.**, a Cayman Islands exempt limited partnership ("<u>Purchaser A</u>"); and **T.N.R. Investments Ltd.**, a company organized and existing under the laws of the State of Israel ("<u>Purchaser B</u>", and together with Purchaser A, the "<u>Purchasers</u>"; each of them a "<u>Purchaser</u>") (each of the Seller, the Company and the Purchasers are referred to herein as a "<u>Party</u>" and together as the "<u>Parties</u>").

RECITALS

WHEREAS, the Seller is the beneficial and record owner of 19,363,396 ordinary shares (as may be increased pursuant and subject to Section 2.2(c), the "Purchased Shares"), par value NIS 0.1 per share of the Company ("Ordinary Shares"), which Ordinary Shares are listed for trading on the Nasdaq Global Select Market (the "Nasdaq") and on the Tel-Aviv Stock Exchange (the "TASE") and which Ordinary Shares are held by the Seller in accordance with a control permit issued by the Israeli Minister of Communications and the Israeli Finance Minister (the "Ministers"), dated April 13, 2010 (the "Current Control Permit");

WHEREAS, the Purchasers desire to purchase from the Seller all of the Purchased Shares, which represent, as of the date hereof, approximately 51.95% of the Issued and Outstanding Share Capital of the Company (as defined herein), for an aggregate purchase price equal to the Seller Purchase Price (as defined below), upon the terms and subject to the conditions set forth in this Agreement;

WHEREAS, the Seller desires to sell the Purchased Shares to the Purchasers upon the terms and subject to the conditions set forth in this Agreement;

WHEREAS, the Purchasers wish to subscribe for the Purchaser Subscribed Shares (as defined herein) for an aggregate purchase price of NIS 260,000,000 (the "Company Purchase Price") upon the terms and subject to the conditions set forth in this Agreement;

WHEREAS, the Seller wishes to subscribe for the Seller Subscribed Shares (as defined herein) for an aggregate purchase price of NIS 35,000,000 (as may be reduced pursuant to the proviso in <u>Section 2.4(a)</u>, the "<u>Share Contribution Amount</u>") upon the terms and subject to the conditions set forth in this Agreement;

WHEREAS, the Seller wishes to subscribe for the Seller Subscribed Debentures (as defined herein) for an aggregate purchase price of NIS 310,000,000 (the "<u>Debenture Contribution Amount</u>") upon the terms and subject to the conditions set forth in this Agreement;

WHEREAS, concurrently with the execution and delivery of this Agreement, each of T.N.R. Real Estate Properties Ltd. (an entity controlled by the Fuhrer family) ("<u>Local Sponsor</u>") and Searchlight Capital II, L.P. and Searchlight Capital II PV, L.P. (each, a "<u>Sponsor</u>") has provided to Purchaser A and Purchase B, respectively, the Commitment Letters (as defined below);

WHEREAS, concurrently with the execution and delivery of this Agreement, each Sponsor has provided the Company and the Seller the limited guarantee, attached hereto as <u>Exhibit A-3</u> and <u>Exhibit A-4</u>, respectively (the "Sponsors' Guarantee"); and

WHEREAS, prior to entering into this Agreement, representatives of the Purchasers have approached the Israeli Ministry of Communication, *inter alia*, for the purpose of commencing discussions with respect to the aforementioned sale and purchase of the Purchased Shares.

NOW, THEREFORE, in consideration of the representations, warranties, covenants and agreements herein contained and other good and valuable consideration (the receipt and sufficiency of which are hereby acknowledged) and intending to be legally bound hereby, the Parties agree and undertake as follows:

1. **DEFINITIONS**

1.1 For purposes of this Agreement, the following terms have the meanings set forth below, which shall be equally applicable to both the singular and plural forms.

"Acquisition Proposal" means any proposal, inquiry or offer with respect to (i) a direct or indirect sale, transfer, exchange, pledge, investment in, acquisition of, or other transaction involving the Purchased Shares or any part thereof or any other Equity Securities or debt securities of any of the BComm Companies (including reorganization, debt restructuring, debt refinancing, creditors arrangement or other similar transaction), including as a result of a tender offer with respect to Equity Securities of the Company, (ii) a direct or indirect acquisition or purchase, in a single transaction or a series of transactions, of a material portion of the assets or a business of the Seller or the BComm Companies, (iii) a merger, consolidation, business combination, recapitalization, liquidation, dissolution, share exchange or similar transaction involving the Seller or any of the BComm Companies, or (iv) any combination of the foregoing, in each case other than the transactions contemplated by this Agreement, in each case, whether initiated, sponsored, conducted, supervised, required or mandated by the Seller, the Company, any of their Affiliates or any other Person.

"Additional Company Purchase Price" means an amount in NIS equal to (a) the number of Unsubscribed Shares, <u>multiplied by</u> (b) the Subscription Price Per Share. For clarity (and without limiting anything contained in this Agreement), in no event shall the Additional Company Purchase Price exceed an aggregate amount of NIS 35,000,000.

"Additional Equity Offering" means as defined in Section 5.7(a).

"Additional Purchaser Subscribed Shares" means as defined in Section 2.3(c).

"Additional Subscription Shares" means as defined in Section 5.7(a).

"Adjustment Subscribed Shares" means as defined in Section 2.3(a).

"Affiliate" or "Affiliated" with respect to any specified Person, means a Person that directly or indirectly, through one or more intermediaries, Controls, is Controlled by, or is under common Control with, such specified Person; provided, however, that, for the purposes of this Agreement and the transactions contemplated hereby, in no event shall Purchaser A be considered an "Affiliate" of, or "Affiliated" with, any portfolio company (as such term is customarily used in the private equity industry) of investment funds or investment vehicles advised or managed (directly or indirectly) by Searchlight Capital Partners, L.P.

- "Agreement" means as defined in the Preamble.
- "Antitrust Clearance" means approval of the transactions contemplated by this Agreement by the Competition Authority of the State of Israel, in accordance with the provisions of the Antitrust Law.
 - "Antitrust Law" means the Antitrust Law, 5748-1988, of the State of Israel.
- "Available Cash" means the amount of freely available cash of the Company as of immediately prior to the Closing. Such Cash Amount shall reflect the impact of, be calculated net of, and be reduced by, the aggregate amount of all Transaction Expenses incurred or otherwise payable by any of the BComm Companies (to the extent not paid prior to the Closing) whether or not due as of the Closing; provided, however, that the calculation of Available Cash shall not be reduced by the amount paid in settlement of any Proceeding that (and solely to the extent such settlement) was expressly approved in advance and in writing by Purchaser A (in its sole and absolute discretion).
- "BComm Companies" means the Company, SP1 and SP2 together; and each of them, a "BComm Company".
 - "Bezeq" means as defined in the Recitals.
 - "Bezeq Designated Directors" means as defined in Section 5.5(b).
- "<u>Bezeq Shares</u>" means 728,373,713 ordinary shares of Bezeq, par value NIS 1.00 each, which are owned by the Company, SP1 or SP2 (and no other Persons).
 - "Burdensome Condition" means as defined in Section 5.1(f).
- "Business Day" means any day (other than Saturday and Sunday) on which commercial banks are generally open for business both in the State of Israel and in New York City, New York, USA.
- "Cash Shortfall" means (a) NIS 695,000,000 minus (b) the Available Cash; provided that if the Cash Shortfall is a negative number, for the purposes of this Agreement the Cash Shortfall shall be deemed to equal zero (0).
 - "Closing" means as defined in Section 7.1.
 - "Closing Date" means as defined in Section 7.1.
- "Commitment Letters" commitment letters by each of the Sponsors, enforceable by the Seller and the Company against the Sponsors in accordance with their respective terms, attached hereto as Exhibit A-1 and Exhibit A-2, respectively.
- "Communications Law" means the Communications Law (Telecommunications and Broadcasting), 5742-1982, as amended from time to time.
- "Communications Order" means the Communications Order (Telecommunications and Broadcasts) (Determination of Essential Services Provided by "Bezeq" The Israeli Telecommunications Corp., Ltd.), 5757-1997, as amended from time to time.
 - "Companies Law" means the Companies Law, 5759-1999.

- "Company" means as defined in the Recitals.
- "Company Debt Modifications" means the debt modifications provided in Exhibit B.
- "Company Designated Directors" means as defined in Section 5.5(a).
- "Company Liability Cap" means an amount equal to NIS 26,000,000.
- "Company Owned Securities" means as defined in Section 3.1(c)(ii).
- "Company Purchase Price" means as defined in the Recitals.
- "Company Series B Debenture" means Series B Debentures issued by the Company on September 21, 2010, as amended from time to time,
- "Company Series C Debenture" means Series C Debentures issued by the Company on September 18, 2016, January 16, 2017 and January 23, 2018, respectively, as amended from time to time.
 - "Company Shareholders Meeting" means as defined in Section 5.9.
- "Consent" means any Permit, approval, consent, ratification, waiver, or other authorization of or by, qualification, registration, designation or filing with any Person, including without limitation, any Governmental Body.
- "Contract" means any legally binding agreement, contract, purchase order, lease, option, license, instrument, mortgage, obligation, commitment, arrangement, promise, or undertaking (whether written or oral and whether express or implied), including all exhibits and schedules thereto and any and all amendments and modifications thereto.
- "Control" means the ability, whether directly or indirectly, to direct (or cause the direction of) the activities, management or policies of the relevant entity, including, the holding of (i) more than 50% of the issued and outstanding share capital of the relevant entity, (ii) such share capital as carries directly or indirectly more than 50% of the shareholders' votes in a general meeting of the relevant entity, (iii) the ability to appoint or elect more than 50% of the directors or equivalent body of such entity, or (iv) the right to receive more than 50% of dividends which may be distributed by the relevant entity.
- "Control Permit" means a control permit in Bezeq to be issued by the Israeli Minister of Communications and the Israeli Prime Minister (or a person authorized by the Prime Minister for such purpose) to the Purchasers in accordance with the Communication Law and the Communication Order (including the Council Recommendation).
- "Council Recommendation" means (i) a positive recommendation by the Cable and Satellite Council to the Minister of Communications of the State of Israel to approve the indirect transfer of control and means of control in the wholly owned subsidiary of Bezeq, D.B.S. Satellite Services 1998 Ltd. (also known by its commercial name "Yes"), and (ii) approval issued by the Israeli Minister of Communications to the Purchasers in accordance with the communication regulation (Telecommunications and Broadcasts) (Procedures and Conditions for Granting a Satellite Broadcasting License), 5758-1998.
 - "Court" means the Tel Aviv District Court.

"Court Approval" means unconditional approval by the Court that does not include a provision stating that it is subject to any further consent by any third party other than the Israeli Ministry of Communications, of the transactions contemplated by this Agreement (without any revisions, modifications or qualifications) as a creditors arrangement and, to the extent required, as a shareholders arrangement pursuant to Section 350 of the Companies Law, including (i) the sale and issuance of the Adjustment Subscribed Shares (if any), the Purchaser Subscribed Shares, the Additional Purchaser Subscribed Shares (if any) and the Election Subscribed Shares (if any), (ii) the adoption of the Company Debt Modifications, (iii) the sale and issuance of the Seller Subscribed Shares and the Series D Debentures, (iv) the sale of the Purchased Shares, and (v) a waiver and exemption, with respect to any act or omission taken by any of the Relevant Parties in connection with this Agreement and the transaction contemplated hereby.

"Current Control Permit" means as defined in the recitals.

"Debenture Contribution Amount" means as defined in the recitals.

"Effect" means as defined in the definition of "Material Adverse Effect" in Section 1.1.

"Election Purchase Price" means as defined in Section 2.3(d).

"Election Subscribed Shares" means as defined in Section 2.4(a).

"Eligible Shareholders" means as defined in Section 5.7(a).

"Employee Body" means any labor union, labor association, labor organization, work council, trade union, or other representative of employees.

"Encumbrance" means any encumbrance, lien, mortgage, charge, option, purchase right, pledge, hypothecation, preemption right, security interest, right of first refusal, transfer restriction, or other rights of third parties, other than (i) Encumbrances arising under the provisions of the Communications Law and the Communications Order; (ii) Encumbrances arising under the Current Control Permit or the Control Permit; (iii) Encumbrances included in the Organizational Documents of the Company as of the date of this Agreement; (iv) Encumbrances arising under the Company Series B Debentures or the Company Series C Debentures and (v) any transfer restrictions under applicable securities Legal Requirements.

"End Date" means November 24, 2019; <u>provided</u> that, in the event Purchaser A has provided a notice that it will not consummate the Closing due to the occurrence of a Material Adverse Effect, the End Date shall be December 10, 2019.

"Equity Securities" of any Person (other than an individual) means, as applicable (i) any and all of its shares of capital stock, membership interests or other equity interests or share capital, (ii) any warrants, Contracts or other rights or options to directly or indirectly subscribe for or to purchase any capital stock, membership interests or other equity interests or share capital of such Person, (iii) any and all securities or instruments, directly or indirectly, exchangeable for or convertible or exercisable into, any of the foregoing or with any profit participation features with respect to such Person, or (iv) any share appreciation rights, phantom share rights or other similar rights with respect to such Person or its business.

"Exchange Act" means the United States Securities Exchange Act of 1934, as amended and the rules and regulations promulgated thereunder from time to time.

"<u>Financing Commitment</u>" means the commitment from the Sponsors to provide funds to the Purchasers at the Closing to pay the Seller Purchase Price, the Company Purchase Price, the Additional Company Purchase Price (if payable under this Agreement) and the Election Purchase Price (if payable under this Agreement), as set forth in, and subject to the terms and conditions of, the Commitment Letters.

"Governmental Body" means any Israeli, U.S. or other federal, state, local, municipal, foreign or other governmental body, including without limitation any administrative, executive, judicial, legislative, regulatory or taxing authority of any nature of any jurisdiction in which the BComm Companies, the Purchasers, the Seller or their respective subsidiaries (including Bezeq and its subsidiaries) (as the case may be) have been incorporated or are conducting business or that otherwise have jurisdiction thereon (including without limitation, any governmental agency, board, bureau, branch, department, official, or entity and any court, arbitrator or other tribunal), including self-regulated organizations or other non-governmental regulatory or quasi-governmental authorities and any national securities exchange or interdealer quotation system.

"<u>Horev Claim</u>" means Derivative Action 47621-07-16 Horev v. B Communications Ltd., et. al. pending in the Tel Aviv District Court.

"<u>IFRS</u>" means the International Financial Reporting Standards promulgated by the International Accounting Standards Board as in effect from time to time.

"Interim Period" means as defined in Section 5.2(a).

"ISA" means the Israel Securities Authority.

"<u>Israeli Securities Law</u>" means the Israeli Securities Law 5728-1968, as amended and the rules and regulations promulgated thereunder from time to time.

"<u>Issued and Outstanding Share Capital of the Company</u>" means the total number of issued and outstanding Ordinary Shares (excluding any Ordinary Shares owned by the Company).

"Knowledge of the Company" means the actual knowledge of (i) the chairman of the board of directors of the Company, (ii) the chief executive officer of the Company or (iii) the chief financial officer of the Company, in each case, as such positions are held as of the date hereof, in their capacity as office holders of the Company.

"Knowledge of the Seller" means the actual knowledge of (i) the chairman of the board of directors of the Seller, (ii) the chief executive officer of the Seller or (iii) the chief financial officer of the Seller, in each case, as such positions are held as of the date hereof, in their capacity as office holders of the Seller.

"<u>Legal Requirement</u>" means any Israeli, U.S. or other federal, state, local, municipal, foreign, international, multinational, or other statute, law, code, Order, constitution, rule, regulation, ordinance, principle of common law, treaty or other mandatory requirement of any Governmental Body (and including the applicable rules of any national securities exchange or interdealer quotation system, including the TASE and Nasdaq).

"Local Sponsor" means as defined in the Recitals.

"Losses" means as defined in Section 9.3(b).

"Material Adverse Effect" means any change, event, occurrence, fact, effect or circumstance (each, an "Effect") that, individually or taken together with all other Effects, has had, or will have, a material adverse effect on the business, condition (financial or otherwise) or results of operations of the BComm Companies and their respective subsidiaries (including Bezeq and its subsidiaries), taken as a whole; provided, however, that none of the following shall be taken into account in determining whether a Material Adverse Effect has occurred: (i) any changes in Israeli economic conditions, (ii) any general changes in conditions affecting the industries in which Bezeq or its subsidiaries operate, (iii) any change in the market price or trading volume of the Ordinary Shares or shares of the Company or Bezeg on the Nasdag or TASE (provided that the exception in this clause (iii) shall not prevent or otherwise affect a determination that any Effect underlying such change in the market price or trading volume has resulted in or contributed to, or would reasonably be expected to result in or contribute to, a Material Adverse Effect), (iv) any change in the credit rating or the credit outlook of the Company or Bezeg or its subsidiaries (provided that the exception in this clause (iv) shall not prevent or otherwise affect a determination that any Effect underlying such change in the credit rating has resulted in or contributed to, or would reasonably be expected to result in or contribute to, a Material Adverse Effect), (v) any regulatory, legislative or political conditions or changes, in each case, generally affecting the industries in which Bezeq or its subsidiaries operate in the State of Israel, including actual or anticipated results of elections, (vi) any failure, in and of itself, by any of the BComm Companies or Bezeg or its subsidiaries to meet any internal or published plans, projections, forecasts, estimates or predictions in respect of revenues, earnings or other financial or operating metrics for any period (provided that the exception in this clause (vi) shall not prevent or otherwise affect a determination that any Effect underlying such failure has resulted in or contributed to, or would reasonably be expected to result in or contribute to, a Material Adverse Effect), (vii) any adverse Effect resulting from the negotiation, execution or delivery of this Agreement or the public announcement of this Agreement, (viii) any changes in IFRS or in laws applicable to the Company or Bezeq or its subsidiaries or the repeal, enforcement or interpretation thereof, (ix) accounting reserves or expenses directly in connection with severance payments in respect of workforce reductions, (x) any general geopolitical conditions, the outbreak or escalation of hostilities, any acts of war (whether or not declared), terrorism or military actions, (xi) any change resulting from acts of God, (xii) any Effect resulting from the impairment for writedown of certain tax assets, including the tax asset of Bezeq associated with the activities of D.B.S. Satellite Services (1998) Ltd., in an aggregate amount not to exceed NIS 1,200,000,000, (xiii) any Effect resulting from investigations conducted by a Governmental Body as of the date hereof against the BComm Companies and their subsidiaries (including Bezeq and its subsidiaries) that, in each case of this clause (xiii), were disclosed in the Public Filings (provided that the exception in this clause (xiii) shall not prevent or otherwise affect a determination that any Effect underlying such investigations has resulted in or contributed to, or would reasonably be expected to result in or contribute to, a Material Adverse Effect); or (xiv) organized labor actions undertaken by Employee Bodies of Bezeq or its subsidiaries; provided, however, that any Effect set forth in the foregoing clauses (i), (ii), (v), (viii) or (x) may be taken into account in determining whether there has been, is or could be a Material Adverse Effect if such Effect has a disproportionate adverse effect on the Company and its subsidiaries (including Bezeq and its subsidiaries), taken as a whole, relative to other Persons similarly situated; provided, further, for clarity, that the so-called "structural separation" imposed by the Israeli Ministry of Communications as and to the extent in effect as of the date of this Agreement (expressly excluding any amendments, changes, expansions or other modifications thereto, or changes or in the enforcement or interpretation thereof), in and of itself, shall not constitute a Material Adverse Effect. For clarity, solely for the purpose of this definition of "Material Adverse Effect" (and no other purposes), an Effect that (individually or together with any other Effects) would reasonably be considered by a buyer of a business or enterprise (taken as a whole) as representing a diminution in the value of such business or enterprise (taken as a whole)

of less than twenty five (25%), would not be considered as a material adverse effect for the purposes of this definition of "Material Adverse Effect".

"Ministers" means as defined in the Recitals.

"Nasdaq" means as defined in the Recitals.

"Non-Recourse Parties" means as defined in Section 9.18.

"Order" means any award, decision, injunction, judgment, order, Permit, decree, ruling, subpoena, or verdict entered, issued, made, or rendered by any court, administrative agency, or other Governmental Body.

"Ordinary Shares" means as defined in the Recitals.

"Organizational Documents" means, with respect to any entity, its certificate of incorporation, organization or formation, memorandum of association, articles of association, bylaws, or other charter or governing documents, and any amendments thereto.

"Party" or "Parties" means as defined in the Preamble.

"<u>Permit</u>" means any permit, license, variance, franchise, approval, certificate, consent, waiver, concession, exemption, Order, registration, notice, filing or other authorization issued by, filed with, or obtained from any Governmental Body.

"<u>Person</u>" means any individual, corporation, general or limited partnership, limited or unlimited liability company, joint venture, estate, trust, incorporated or unincorporated association, firm, organization, labor union, or other entity or Governmental Body.

"Pro Rata Share" means (a) with respect to Purchaser B, the percentage specified in a written notice delivered by Purchaser A to the Company and the Seller at least three (3) Business Days prior to the Closing Date (the "Specified Percentage"); provided that (i) the Specified Percentage shall be no less than 13% and not greater than 19%, and (ii) if no such notice has been provided by Purchaser A prior to or on the date that is three (3) Business Days prior to the Closing Date, the Specified Percentage shall equal 19%, or (b) with respect to Purchaser A, a percentage equal to (x) 100% minus (y) the Specified Percentage.

"Proceeding" means any written claim, demand, summon, subpoena, order to show cause, action, arbitration, audit, hearing, investigation, litigation, suit or other proceeding (whether civil, criminal, administrative, regulatory or investigative) commenced, brought, conducted, or heard by or before any court or other Governmental Body.

"Public Filings" means the reports, schedules, forms, statements and other documents filed by the Company or Bezeq with the SEC or the ISA, as applicable, and publically available at least two (2) Business Days prior to the date of this Agreement.

"Purchased Shares" means as defined in the Recitals.

"Purchaser A" means as defined in the Preamble.

"Purchaser B" means as defined in the Preamble.

"Purchaser Election" means as defined in Section 2.4(a).

"Relevant Parties" means the directors and officers of the Seller and the Company, and the bondholders of the Seller and the Company, bondholder's trustees, bondholders representative bodies, bondholder financial consultants and legal counsel, in each case, solely in their capacity as such.

"Restricted Matters" means, with respect to any Person, any of the following actions, effects or things: (a) issuance, sale, disposition of, grant, delivery, Encumbrance, or agreement, authorization, or commitment to the issuance, sale, disposition of, grant, delivery, or Encumbrance by such Person (in each case, whether through the issuance or granting of options, warrants, commitments, subscriptions, rights to purchase or otherwise) of any of its Equity Securities or debt securities (other than, solely in the case of Bezeg, issuance of Equity Securities of Bezeg pursuant to employee benefit plans disclosed in the Public Filings), (b) split, combination, subdivision, reclassification or redemption, repurchase or other acquisition by such Person of, directly or indirectly, any of such Person's Equity Securities, (c) with respect to any BComm Company, declaration, set aside or payment of any dividend or other distribution (whether in cash, shares or property or any combination thereof) in respect of any of such Person's Equity Securities, or any other actual, constructive or deemed distribution in respect of such Person's Equity Securities, (d) amendment, replacement or other modification of such Person's Organizational Documents; provided that the Company may, to the extent approved by the Court (in a decision the execution of which was not stayed) to amend its Organizational Documents only to the extent necessary to increase its authorized share capital and for no other purpose, (e) approval, proposal or adoption of, or other agreement by such Person to, a plan of complete or partial liquidation, dissolution, merger, consolidation, amalgamation, restructuring, recapitalization or other reorganization, (f) settlement of any pending or threatened derivative or class action brought against such Person; provided that the Company, subject to all applicable Legal Requirements, may settle the Horev Claim solely to the extent that such settlement does not impose or would result in any liabilities or obligations (monetary or otherwise, including any direct or indirect payment obligations or liabilities) to or of any of the BComm Companies (whether or not covered by insurance, and including under any D&O or other insurance policies maintained by or for the benefit of any of the BComm Companies), (g) settlement of any other material Proceeding, (h) entry into, amendment or other modification to any Contract with any controlling shareholder of such Person, other than termination of any such Contract without any liability or obligations to any of the BComm Companies or their respective subsidiaries, or (i) with respect to any BComm Company, (1) incurrence or assumption of, or amendment or other modification to, any Contract or other instrument evidencing, any long-term or short-term debt for borrowed monies or issuance of, or amendment or modification to the terms of, any debt securities by such Person (including the Company Debt Modifications), (2) assumption, guarantee or endorsement of, or otherwise becoming liable or responsible (whether directly, contingently or otherwise) for the obligations of any other Person, (3) the making of any loans, advances or capital contributions to or investments in any other Person, or (4) mortgaging or pledging any of such Person's assets, tangible or intangible, or creation any Encumbrances thereupon (other than the Company Debt Modifications).

"RTP Law" means the Israeli Restrictive Trade Practices Law, 5748-1988, as amended from time to time, and any rules, regulations or guidelines promulgated thereunder.

[&]quot;Purchaser Liability Cap" means an amount equal to NIS 30,000,000.

[&]quot;Purchaser Subscribed Shares" means as defined in the Section 2.3(b).

[&]quot;Representative" means as defined in Section 5.3(a).

"Sale Process" means all matters, whether occurring before, on or after the date of this Agreement, relating, directly or indirectly, to the sale of any interest in the Seller or any of the BComm Companies or any of their respective assets, including any arrangement with their respective bondholders and other creditors or other debt restructuring (and review of strategic alternatives with respect to any such transactions), and all activities in connection therewith, including matters relating to: (a) the solicitation of proposals from, and negotiating with, third parties, including the Purchasers, and (b) the drafting and negotiation of any of the provisions of any of the Transaction Documents.

"SCP Entity" means any of the investment funds or investment vehicles advised or managed (directly or indirectly) by Searchlight Capital Partners, L.P. or any of their respective portfolio companies (as such term is customarily used in the private equity industry).

"SEC" means the United States Securities and Exchange Commission.

"Securities Act" means the United States Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder from time to time.

"Securities Laws" means the Israeli Securities Law, the Exchange Act and the Securities Act.

"Seller" means as defined in the Preamble.

"Seller Contribution Amount" means the Share Contribution Amount and the Debenture Contribution Amount.

"Seller Liability Cap" means an amount equal to NIS 22,500,000.

"Seller Purchase Price" means an amount of NIS 225,000,000 (two hundred twenty five million New Israeli Shekels).

"Seller Subscribed Debentures" means NIS 310,000,000 par value of Series C Debentures issued to the Seller by the Company by way of an expansion of the existing Series C Debentures, through a private placement.

"Seller Subscribed Shares" mean as defined in Section 2.4.

"Share Contribution Amount" means an defined in the recitals.

"Share Pledge" means as defined in Section 5.1(g).

"Series D Debentures" means a new series of Company debentures, in an aggregate amount of NIS 58,000,000 par value, to be issued by the Company at the Closing, having identical terms and conditions to those of the Company Series C Debenture, with the following exception: Within seven (7) days following the date on which 90% of the aggregate amount of dividends distributed by Bezeq to (and retained by) the Company following the Closing Date equals at least the principal amount of the Series D Debentures, the Company will issue an immediate report calling for a full prepayment of the Series D Debentures, which prepayment will be made in accordance with the terms of the applicable deed of trust.

"Shareholders Approval" means as defined in Section 5.9.

"SP1" means B Communications (SP1) Ltd., an Israeli limited company.

- "SP1 Bezeq Shares" means as defined in Section 3.1(c)(ii).
- "SP1 Securities" means as defined in Section 3.1(c)(ii).
- "SP2" means B Communications (SP2) Ltd., an Israeli limited company.
- "SP2 Bezeq Shares" means as defined in Section 3.1(c)(ii).
- "SP2 Securities" means as defined in Section 3.1(c)(ii).
- "Sponsor" means as defined in the Recitals.
- "Sponsors' Guarantee" means as defined in the Recitals.
- "Subscription Price Per Share" means NIS 4.175.
- "TASE" means as defined in the Recitals.

"<u>Transaction Documents</u>" means (i) this Agreement, (ii) the Commitment Letters, (iii) the Sponsors' Guarantee, and (iv) the other documents, agreements, exhibits, schedules, statements or certificates being executed and delivered by the Seller, the Company or the Purchasers in connection with this Agreement and the transactions contemplated hereby.

"Transaction Expenses" means any of the BComm Companies' fees, costs and expenses incurred or payable by the BComm Companies (including on behalf of their respective bondholders) in connection with the Sale Process (i,ncluding the consideration, preparation, negotiation execution, and performance of this Agreement, the Company Debt Modifications or the transactions contemplated hereby or thereby) or otherwise pursuant to the Company Series B Debentures or Company Series C Debentures, including all fees, costs and expenses of trustees, bankers, agents, representatives, counsel, accountants, economic advisors, or other advisors or representative, and including any such amounts incurred or payable by the BComm Companies on behalf of any other Person.

"Unsubscribed Shares" means as defined in Section 5.7(c).

"Withholding Tax Certificate" means as defined in Section 2.2(e).

2. SALE AND PURCHASE OF SHARES; SUBSCRIPTION; CONTRIBUTION

- 2.1 <u>Sale and Purchase of Shares</u>. Upon the terms and subject to the conditions of this Agreement, at the Closing, the Seller shall sell, assign, transfer, convey and deliver to each Purchaser, and each Purchaser shall purchase and accept from the Seller, its Pro Rata Share of all (but not part) of the Purchased Shares, free and clear of any Encumbrances.
 - 2.2 Seller Purchase Price.
 - (a) The aggregate purchase price of the Purchased Shares is the Seller Purchase
- Price.
- (b) [RESERVED].
- (c) [RESERVED].
- (d) [RESERVED].

- Each Purchaser shall pay its Pro Rata Share of the Seller Purchase Price in (e) New Israeli Shekels, and shall, from any consideration payable hereunder by the Purchasers, only withhold or deduct any taxes under Israeli law as provided in the valid withholding tax certificate provided by the Seller or the Company, as applicable, to the Purchasers not less than three (3) Business Days prior to the Closing Date (the "Withholding Tax Certificate"); provided, however, that (i) such Withholding Tax Certificate will be in a form reasonably acceptable to Purchaser A, and (ii) if the Withholding Tax Certificate shall not have been provided, expired or ceased to be valid for any other reason, prior to the Closing, then the Purchasers shall deduct and withhold taxes from any such consideration payable by the Purchasers to the Seller hereunder as required by applicable Legal Requirements, unless and to the extent the Seller or the Company, as applicable, provides the Purchasers with a valid certificate issued by the Israel Tax Authority, at least three (3) Business Days prior to the applicable payment date, exempting the Purchasers from withholding tax from payments made under this Agreement or setting a lower withholding tax rate (in which case the Purchasers shall deduct and withhold an amount in accordance therewith). All amounts so deducted and withheld by the Purchasers shall be treated for all purposes of this Agreement as having been paid to the applicable payee.
- 2.3 <u>Purchaser Subscription</u>. Upon the terms and subject to the conditions of this Agreement, at the Closing, the Company shall issue, free and clear of all Encumbrances:
- (a) To the Purchasers (allocated among them pro-rata, based on their respective Pro Rata Share), such number of newly issued Ordinary Shares (the "<u>Adjustment Subscribed Shares</u>") equal to (i) the Cash Shortfall, <u>divided by</u> (ii) Subscription Price Per Share (rounded up to the nearest whole number) at no additional consideration (for the avoidance of doubt, nothing in this <u>Section 2.3(a)</u> of otherwise shall limit or otherwise affect Purchaser A's rights pursuant to <u>Section 6.2(i)</u>);
- (b) To the Purchasers (allocated among them pro-rata, based on their respective Pro Rata Share), 62,275,450 newly issued Ordinary Shares (the "<u>Purchaser Subscribed Shares</u>") in exchange for the Company Purchase Price;
- (c) To Purchaser A, if the amount of the Unsubscribed Shares is greater than zero (0), additional newly issued Ordinary Shares in an amount equal to the Unsubscribed Share (the "<u>Additional Purchaser Subscribed Shares</u>") in exchanged for the Additional Company Purchase Price; and
- (d) To Purchaser A, if Purchaser A has made the Purchaser Election, the Election Subscribed Shares in exchange for NIS 35,000,000 (the "Election Purchase Price")
- 2.4 <u>Seller Contribution</u>. Upon the terms and subject to the conditions of this Agreement, at the Closing:
- (a) The Seller shall pay to the Company the Share Contribution Amount in exchange for the issuance by the Company to the Seller of 8,383,234 newly issued Ordinary Shares (as may be adjusted pursuant to the proviso in this Section 2.4(a), the "Seller Subscribed Shares"), free and clear of all Encumbrances and any transfer restrictions under applicable Legal Requirements; provided, however, that notwithstanding anything herein to the contrary, in the event that the Seller (x) is unable to satisfy its obligations pursuant to this Section 2.4(a) at the Closing if the Closing were then to occur, or (y) otherwise failed to satisfy its obligations pursuant to this Section 2.4(a), then, in each case, Purchaser A may (but shall not be obligated to) elect (in its sole and absolute discretion, the "Purchaser Election"), to subscribed for such 8,383,234 newly issued Ordinary Shares in lieu of the Seller (the "Election Subscribed Shares") for the Election Purchase

Price, and if such Purchaser Election has been made (i) the Seller Subscribed Shares shall equal zero (0) (and the Seller shall have no rights or obligations in respect thereof nor shall any of the Parties have any claim towards the Seller in connection with such related reduction of the Seller Subscribed Share), and (ii) the Share Contribution Amount shall be reduced to zero (0); and

(b) The Seller shall pay to the Company the Debenture Contribution Amount in exchange for the issuance by the Company to the Seller of the Seller Subscribed Debentures, free and clear of all Encumbrances and any transfer restrictions under applicable Legal Requirements.

2.5 [RESERVED].

3. REPRESENTATIONS AND WARRANTIES OF THE SELLER AND THE COMPANY

- 3.1 The Seller represents and warrants to each of the Purchasers that the statements contained in this <u>Section 3.1</u> are true and correct as of the date hereof, and will be true and correct as of the Closing Date as though made as of the Closing Date:
- (a) Organization, Corporate Power and Qualification. The Seller is a public company duly organized and validly existing under the laws of the State of Israel and has all requisite corporate power and authority to (a) conduct its business as and to the extent now conducted and to own, use and lease its assets and properties, and (b) execute and deliver this Agreement and the other Transaction Documents to which it is a party and, subject to the satisfaction of the conditions to Closing, to consummate the transactions and perform its obligations contemplated hereby and thereby (including the right, authority and power to sell, assign and transfer the Purchased Shares to the Purchasers in accordance with the terms of this Agreement). The Company is a public company duly organized and validly existing under the laws of the State of Israel and has all requisite corporate power and authority to conduct its business as and to the extent now conducted and to own, use and lease its assets and properties. Each of SP1 and SP2 is a private limited company duly organized and validly existing under the laws of the State of Israel and has all requisite corporate power and authority to conduct its business as and to the extent now conducted and to own, use and lease its assets and properties.
- (b) <u>Authorization; Binding Agreement.</u> All corporate action on the part of the Seller necessary or required for the authorization, execution, delivery and performance of all its obligations under this Agreement and any other Transaction Document to which Seller is a party have been duly and validly taken, and (subject to obtaining the required approvals under Section 350 of the Companies Law) no other proceedings or Consents (including of Eurocom Communications Ltd., or any creditors or liquidators thereof or any other Governmental Body) on the part of the Seller are necessary or required to authorize the execution, delivery and performance of this Agreement and the other Transaction Documents to which Seller is a party and the consummation by the Seller of the transactions contemplated hereby or thereby. This Agreement is duly and validly executed and delivered by the Seller, and constitutes the valid and legally binding obligations of the Seller, legally enforceable against the Seller in accordance with its terms subject to bankruptcy, insolvency, reorganization and other Legal Requirements of general applicability relating to or affecting creditors' rights and to general equity principles.

(c) Title and Ownership.

(i) The Seller is the holder and sole record, legal and beneficial owner of the Purchased Shares and has good, valid and marketable title to the Purchased Shares; and the Purchased Shares are fully paid, non-assessable and free and clear of all Encumbrances.

There are no Contracts to which the Seller is a party (including subscriptions, options, warrants, rights, calls, puts, or other arrangements of any kind) or Orders which (i) obligate (or would obligate, upon the occurrence of any event) the Seller to sell, transfer, purchase or otherwise dispose of or acquire any Equity Securities of any BComm Company, or (ii) would restrict the Seller from selling, transferring, or otherwise disposing of, directly or indirectly, any Equity Securities of the Company, other than the Current Control Permit.

- The Company is the holder and sole record, legal and beneficial (ii) owner of (i) all of the Equity Securities of SP1 (the "SP1 Securities"), and (ii) 14,204,153 of the Bezeg Shares (the Equity Securities described in the foregoing clauses (i) and (ii), collectively, the "Company Owned Securities"). The Company has good, valid and marketable title to the Company Owned Securities; and the Company Owned Securities are fully paid, non-assessable and free and clear of all Encumbrances (other than Encumbrances expressly contemplated by the Current Control Permit). There are no Contracts to which any of the BComm Companies is a party or by which any of them is bound (including subscriptions, options, warrants, rights, calls, puts, or other arrangements of any kind) or Orders which (i) obligate or would obligate the Company or any of its Affiliates to sell, transfer, purchase, or otherwise dispose of or acquire any Company Owned Securities or any other Equity Securities of Bezeg, SP1 or SP2 or any of their respective subsidiaries, or (ii) would restrict any of the BComm Companies from selling, transferring, purchasing, or otherwise disposing of or acquiring any of the Company Owned Securities or any other Equity Securities of SP1, SP2 or Bezeg, other than the Current Control Permit. SP1 is the holder and sole record, legal and beneficial owner of (1) all of the Equity Securities of SP2 (the "SP2 Securities"), and (2) none of the Bezeq Shares (the "SP1 Bezeq Shares"), and SP2 is the holder and sole record, legal and beneficial owner of 714,169,560 of the Bezeg Shares (the "SP2 Bezeg Shares"). SP1 has good, valid and marketable title to the SP2 Securities and the SP1 Bezeq Shares; and the SP2 Securities and the SP1 Bezeq Shares are fully paid, non-assessable and free and clear of all Encumbrances. SP2 has good, valid and marketable title to the SP2 Bezeg Shares; and the SP2 Bezeg Shares are fully paid, non-assessable and free and clear of all Encumbrances. There are no Contracts to which any of the BComm Companies is a party or by which any of them is bound (including subscriptions, options, warrants, rights, calls, puts, or other arrangements of any kind) or Orders which (I) obligate or would obligate any of SP1 or SP2 to sell, transfer, purchase, or otherwise dispose of or acquire any Equity Securities of its subsidiaries or Equity Securities of Bezeg or any of its subsidiaries, or (II) would restrict any SP1 or SP2 from selling, transferring, purchasing, or otherwise disposing of or acquiring any Equity Securities of its subsidiaries or Equity Securities of Bezeg, other than the Current Control Permit.
- (d) <u>Capitalization</u>. The Purchased Shares represent as of the date hereof approximately 51.95% of the Issued and Outstanding Share Capital of the Company. Except for the Purchased Shares, the SP1 Securities, the SP2 Securities and the Bezeq Shares, (i) neither the Seller nor any of the BComm Companies, directly or indirectly, beneficially or of record, owns any Equity Securities of the Company or any of its subsidiaries (including Bezeq and its subsidiaries), and (ii) the Seller, directly or indirectly, beneficially or of record, does not own any Equity Securities of the BComm Companies.
- (e) <u>Consents and Filings; Non Contravention</u>. Other than as detailed in <u>Schedule 3.1(e)</u> hereof, and subject to the required approvals under Section 350 of the Companies Law the execution, delivery and performance of this Agreement and any other Transaction Document (in accordance with their respective terms) to which Seller is a party, by the Seller and the consummation of the transactions contemplated hereby and thereby will not result in any violation

of or be in conflict with or constitute, with or without the passage of time or giving of notice, a default under, or require any Consent of any Person pursuant to, (a) any provision of the Seller's or any of the BComm Companies' Organizational Documents, (b) any Order, Permit or Contract to which it or any of the BComm Companies is a party or, to the Knowledge of the Seller, by which it is bound, or would result in the loss of any rights, or the imposition or creation of any Encumbrance upon any assets or properties of the Seller or any of the BComm Companies, under any such Order, Permit or Contract or (c) any provisions of any Legal Requirement applicable to it or any of the BComm Companies.

(f) <u>Brokers' Fees</u>. No agent, broker, investment banker, or other Person acting in a similar capacity on behalf of or under the authority of the Seller is or will be entitled to any broker's or finder's fee or any other commission or similar fee, directly or indirectly, in connection with the transactions contemplated by this Agreement (including the sale and transfer of the Purchased Shares as contemplated under this Agreement), other than Oppenheimer & Co. Inc. and Migdal Capital Markets, for which the Seller shall be solely responsible.

(g) Undisclosed Proceeding Matters; Affiliate Transactions.

- (i) Except as set forth on (i) <u>Schedule 3.1(g)(i)</u> or (ii) the Public Filings, to the Knowledge of the Seller, as of the date hereof (and not, for clarity, as of the Closing Date), there exist no event, occurrence or fact materially affecting any material Proceedings in which any of the BComm Companies or Bezeq (or any of their respective subsidiaries) is involved or that would materially and adversely affect any of them.
- (ii) Except as set forth on <u>Schedule 3.1(g)(ii)</u>, there are no Contracts or other arrangements for the provision of any services between the Seller, on the one hand, and any of the BComm Companies, on the other hand; no personnel, assets or properties of the Seller are used in or utilized for the conduct of the business of any of the BComm Companies.
- (h) <u>No Additional Representations</u>. The representations and warranties made by the Seller in this <u>Section 3.1</u> are the exclusive representations and warranties made by it in connection with the transactions contemplated hereby and the Seller hereby disclaims any other express or implied representations or warranties.
- 3.2 The Company represents and warrants to each of the Purchasers (and not any other Party) that the statements contained in this <u>Section 3.2</u> are true and correct as of the date hereof, and will be true and correct as of the Closing Date as though made as of the Closing Date:
- (a) Organization, Corporate Power and Qualification. The Company is a public company duly organized and validly existing under the laws of the State of Israel and has all requisite corporate power and authority to conduct its business as and to the extent now conducted and to own, use and lease its assets and properties. Each of SP1 and SP2 is a private limited liability company duly organized and validly existing under the laws of the State of Israel and has all requisite corporate power and authority to conduct its business as and to the extent now conducted and to own, use and lease its assets and properties. The Company has made available to the Purchasers true, correct and complete copies of all Organizational Documents of each of SP1 and SP2.
- (b) <u>Authorization; Binding Agreement</u>. All corporate action on the part of the Company necessary or required for the authorization, execution, delivery and performance of all its obligations under this Agreement and any other Transaction Document to which the Company is a

party have been duly and validly taken, and (subject to obtaining the required approvals under Section 350 of the Companies Law) no other proceedings or Consents (including any other Governmental Body) on the part of the Company are necessary or required to authorize the execution, delivery and performance of this Agreement and the other Transaction Documents to which the Company is a party and the consummation by the Company of the transactions contemplated hereby or thereby. This Agreement is duly and validly executed and delivered by the Company, and constitutes the valid and legally binding obligations of the Company, legally enforceable against the Company in accordance with its terms subject to bankruptcy, insolvency, reorganization and other Legal Requirements of general applicability relating to or affecting creditors' rights and to general equity principles.

- Title and Ownership. The Company is the holder and sole record, legal and beneficial owner of all of the Company Owned Securities. The Company has good, valid and marketable title to the Company Owned Securities; and the Company Owned Securities are fully paid, non-assessable and free and clear of all Encumbrances. Subject to the fulfillment of the Conditions to Closing, there are no Contracts to which any of the BComm Companies is a party (including subscriptions, options, warrants, rights, calls, puts, or other arrangements of any kind) or Orders which (i) obligate or would obligate the Company upon the occurrence of any event to sell, transfer, purchase, acquire or otherwise dispose of any Company Owned Securities or any other Equity Securities of Bezeq, SP1 or SP2 or any of their respective subsidiaries, or (ii) would restrict (x) any of the BComm Companies from selling, transferring, purchasing, or otherwise disposing of or acquiring any of the Company Owned Securities or any other Equity Securities of SP1, SP2 or Bezeg, other than the Current Control Permit or (y) the Company from issuing the Adjustment Subscribed Shares (if any), the Purchaser Subscribed Shares, the Additional Purchaser Subscribed Shares (if any) or the Election Subscribed Shares (if any). SP1 is the holder and sole record, legal and beneficial owner of all of the SP2 Securities and the SP1 Bezeg Shares, and SP2 is the holder and sole record, legal and beneficial owner of the SP2 Bezeq Shares. SP1 has good, valid and marketable title to the SP2 Securities and the SP1 Bezeg Shares; and the SP2 Securities and the SP1 Bezeg Shares are fully paid, non-assessable and free and clear of all Encumbrances. SP2 has good, valid and marketable title to the SP2 Bezeg Shares; and the SP2 Bezeg Shares are fully paid, nonassessable and free and clear of all Encumbrances. There are no Contracts to which any of the BComm Companies is a party (including subscriptions, options, warrants, rights, calls, puts, or other arrangements of any kind) or Orders which (I) obligate or would obligate upon the occurrence of any event any of SP1 or SP2 to sell, transfer, purchase or otherwise dispose of or acquire any Equity Securities of its subsidiaries or Equity Securities of Bezeq or any of its subsidiaries, or (II) would restrict any SP1 or SP2 from selling, transferring, purchasing or otherwise disposing of or acquiring any Equity Securities of its subsidiaries or Equity Securities of Bezeg, other than the Current Control Permit.
- (d) <u>Capitalization</u>. The Purchased Shares represent as of the date hereof approximately 51.95% of the Issued and Outstanding Share Capital of the Company. Except for the Purchased Shares, the SP1 Securities, the SP2 Securities and the Bezeq Shares, none of the BComm Companies, directly or indirectly, beneficially or of record, owns any Equity Securities of the Company or any of its subsidiaries (including Bezeq and its subsidiaries). (i) Except for the Company Owned Securities, the Company does not own or hold any Equity Securities of any Person, (ii) except for the SP1 Bezeq Shares and the SP2 Securities, SP1 does not own or hold any Equity Securities of any Person, and (iii) except for the SP2 Bezeq Shares, SP2 does not own or hold any Equity Securities of any Person other than as part of its investment portfolio, other than, in the case of the preceding clauses (i) through (iii), Equity Securities that are publicly traded on a recognized national stock exchange and held for short term investment purposes; provided that such

Equity Securities, in the aggregate (with respect to all cases of clauses (i) through (iii)), do not represent more than half a percent (0.5%) of the Equity Securities of any issuer.

- (e) <u>Consents and Filings; Non Contravention</u>. Other than as detailed in <u>Schedule 3.2(e)</u> hereof and subject to the required approvals under Section 350 of the Companies Law, the execution, delivery and, subject to the Shareholders Approval performance, of this Agreement and any other Transaction Document to which it is a party by the Company and the consummation of the transactions contemplated hereby and thereby will not result in any violation of or be in conflict with or constitute, with or without the passage of time or giving of notice, a default under, or require any Consent of any Person pursuant to, (i) any provision of any of the BComm Companies' Organizational Documents, (ii) any Order, Permit or Contract to which it or any of the BComm Companies is a party or by which it is bound, or would result in the loss of any rights, or the imposition or creation of any Encumbrance upon any assets or properties of any of the BComm Companies, under any such Order, Permit or Contract, or (iii) any provisions of any Legal Requirement applicable to it, any of the BComm Companies or Bezeq (or any of its subsidiaries).
- (f) <u>Brokers' Fees; Transaction Expenses</u>. No agent, broker, investment banker, or other Person acting in a similar capacity on behalf of or under the authority of any of the BComm Companies is or will be entitled to any broker's or finder's fee or any other commission or similar fee, directly or indirectly, in connection with the transactions contemplated by this Agreement (including the issuance and delivery of the Adjustment Subscribed Shares (if any), the Purchaser Subscribed Shares, the Additional Purchaser Subscribed Shares (if any) and the Election Subscribed Shares (if any) as contemplated under this Agreement).
- (g) <u>Undisclosed Proceeding Matters</u>. Except as set forth on (i) <u>Schedule 3.2(g)</u> or (ii) the Public Filings, to the Knowledge of the Company, as of the date hereof (and not, for clarity, as of the Closing Date), there exist no event, occurrence or fact materially affecting any material Proceedings in which any of the BComm Companies or Bezeq (or any of their respective subsidiaries) is involved or that would materially and adversely affect any of them.

(h) Subscribed Shares.

- (i) The Adjustment Subscribed Shares, the Purchaser Subscribed Shares, the Additional Purchaser Subscribed Shares and the Election Subscribed Shares have been duly authorized and allotted in accordance with applicable Legal Requirement and the Organizational Documents of the Company, and shall be available for subscription by the Purchaser in the manner and time provided under this Agreement.
- (ii) The Adjustment Subscribed Shares (if any), the Purchaser Subscribed Shares, the Additional Purchaser Subscribed Shares (if any) and the Election Subscribed Shares (if any), when issued at the Closing, shall be duly issued in accordance with applicable Legal Requirements and the Organizational Documents of the Company, fully paid and non-assessable and shall be free and clear of any Encumbrances.
- (iii) The Adjustment Subscribed Shares (if any), the Purchaser Subscribed Shares ,the Additional Purchaser Subscribed Shares (if any) and the Election Subscribed Shares (if any), as of the Closing, will not be subject to avoidance or liquidation in bankruptcy, composition or other solvency proceedings relating to the Company, as the case may be.
- (i) The representations and warranties made by the Company in this <u>Section 3.2</u> or in any other Transaction Document are the exclusive representations and warranties made by it

in connection with the transactions contemplated hereby and the Company hereby disclaims any other express or implied representations or warranties.

3.3 The Company represents and warrants to the Seller that the statements contained in this <u>Section 3.3</u> are true and correct as of the date hereof, and will be true and correct as of the Closing Date as though made as of the Closing Date:

(a) Seller Subscribed Shares and Debentures.

- (i) The Seller Subscribed Shares (if any) and the Seller Subscribed Debentures, when issued at the Closing, shall be duly authorized and allotted in accordance with applicable Legal Requirement and the Organizational Documents of the Company, and shall be available for subscription by the Seller in the manner and time provided under this Agreement.
- (ii) The Seller Subscribed Shares (if any) and the Seller Subscribed Debentures, when issued at the Closing, shall be duly issued in accordance with applicable Legal Requirements and the Organizational Documents of the Company, fully paid and non-assessable and shall be free and clear of any Encumbrances and any transfer restrictions under applicable Legal Requirements and tradeable.
- (iii) The Seller Subscribed Shares (if any) and the Seller Subscribed Debentures, as of the Closing, will not be subject to avoidance or liquidation in bankruptcy, composition or other solvency proceedings relating to the Company, as the case may be.
- (b) <u>No Additional Representations</u>. The representations and warranties made by the Company in this <u>Section 3.3</u> are the exclusive representations and warranties made by it to the Seller in connection with the transactions contemplated hereby and the Company hereby disclaims any other express or implied representations or warranties made to the Seller in connection with this Agreement, the other Transaction Documents or the transactions contemplated hereby and thereby.

4. REPRESENTATIONS AND WARRANTIES OF EACH PURCHASER

Each Purchaser, severally (and not joint and severally), with respect to itself only, represents and warrants to the Seller and the Company that the statements contained in this <u>Section 4</u>, with respect to such Purchaser, are true and correct as of the date hereof, and will be true and correct as of the Closing Date as though made as of the Closing Date:

- 4.1 <u>Organization, Corporate Power and Qualification</u>. In the case of Purchaser A, such Purchaser is a Cayman Islands exempt limited partnership duly organized, validly existing under the laws of the Cayman Islands, or in the case of Purchaser B, an Israeli company duly organized, validly existing under the laws of the State of Israel, and has all requisite corporate power and authority to execute and deliver this Agreement and all other Transaction Documents to which it is a party and to consummate the transactions and perform its obligations contemplated hereby and thereby (including the right, authority and power to purchase and acquire the Purchased Shares and subscribe and acquire the Subscribed Shares in accordance with the terms of this Agreement).
- 4.2 <u>Authorization; Binding Agreement.</u> All corporate action on the part of such Purchaser necessary or required for the authorization, execution, delivery and performance of all its obligations under this Agreement and all other Transaction Documents to which it is a party have been duly and validly taken, and no other proceedings on the part of such Purchaser are necessary or required to authorize the execution, delivery and performance of this Agreement and the other

Transaction Documents to which it is a party and the consummation by such Purchaser of the transactions contemplated hereby or thereby. This Agreement is duly and validly executed and delivered by such Purchaser, and constitutes the valid and legally binding obligations of such Purchaser, legally enforceable against such Purchaser in accordance with its terms subject to bankruptcy, insolvency, reorganization and other Legal Requirements of general applicability relating to or affecting creditors' rights and to general equity principles.

4.3 <u>Consents and Filings; Non Contravention</u>. Subject to obtaining the Control Permit and the Antitrust Clearance, the execution, delivery and performance of this Agreement and any other Transaction Document to which it is a party by such Purchaser and the consummation of the transactions contemplated hereby and thereby will not result in any violation of or be in conflict with or constitute, with or without the passage of time or giving of notice, a default under, or require Consent of any Person pursuant to, any provision of such Purchaser's Organizational Documents or any Order, Permit or Contract to which it is a party or by which it is bound or any provisions of any Legal Requirement applicable to it.

4.4 Financial Capacity.

- (a) Taking into account the Financing Commitment, such Purchaser has, and at the Closing will have, sufficient resources to pay, in cash any and all amounts necessary for it to consummate the transactions contemplated hereby at the Closing, including payment of its Pro Rata Share of the Seller Purchase Price and the Company Purchase Price, and in the case of Purchaser A only, the Additional Company Purchase Price and the Election Purchase Price (to the extent payable hereunder) and all the fees and expenses expressly required to be paid by such Purchaser hereunder without any restrictions to transfer such funds at Closing to the Seller and the Company, as and to the extent required to be paid pursuant to, and subject to the terms of, this Agreement. The Sponsor affiliated with such Purchaser has, and at the Closing will have, sufficient resources to meet its obligations under its Commitment Letter as they become due.
- (b) As of the date of this Agreement, such Purchaser has delivered to the Seller a true, correct and complete copy of the Commitment Letter provided by its affiliated Sponsor, dated as of the date hereof. Such Commitment Letter has not been amended or modified, and the respective commitments contained in such Commitment Letter have not been withdrawn, terminated or rescinded. Such Commitment Letter (i) is in full force and effect, (ii) constitutes the legal, valid and binding obligation of such Purchaser and the Sponsor party thereto, and (iii) is enforceable by the Seller and the Company against such Purchaser and the Sponsor party thereto, in accordance with its terms, subject to bankruptcy, insolvency, reorganization and other Legal Requirements of general applicability relating to or affecting creditors' rights and to general equity principles. There are no side letters or other Contracts related to the funding or investing, as applicable, of the applicable Financing Commitment other than such Commitment Letter. There are no conditions precedent to the consummation of such Financing Commitment other than those set forth in such Commitment Letter. As of the date of this Agreement, the Sponsor affiliated with such Purchaser is not subject to bankruptcy proceedings.
- (c) Notwithstanding anything to the contrary contained herein, in no event shall this <u>Section 4.4</u> be deemed breached (and no condition set forth in <u>Section 6.3</u> shall be deemed to have failed as a result of any actual or alleged breach of this <u>Section 4.4</u>), if (notwithstanding any actual or alleged breach), such Purchaser is willing and able to consummate its obligations at the Closing if and when it is otherwise required to do so under the terms and conditions of this Agreement.

4.5 <u>Compliance</u>.

- (a) Such Purchaser hereby confirms that it is familiar with the requirements of the Communications Law and Communications Order and has conducted independent investigations (including with consultants on its behalf) related thereto and that it has reviewed the Current Control Permit, and such Purchaser is not aware, as of the date of this Agreement, of any facts or circumstances that would prevent it from complying with all such requirements, including with respect of the 'Israeli Entity' (as such term is defined in the Communications Order) under the Communications Order, in each case, as such Legal Requirements are in effect as of the date of this Agreement; provided that it is understood and acknowledge by all Parties that the Purchasers are proposing to effect the transaction contemplated by this Agreement through a contractual arrangement between them pursuant to Section 4(a)(3) of the Communications Orders.
- (b) Neither such Purchaser nor any of its 'Interested Parties' (as such term is defined in the Communications Law) (including the Sponsors) has, on its own behalf or in acting on behalf of any other Person, directly or indirectly, engaged in any transactions, or otherwise had any trade, commercial, financial or other relationships with (i) any Person organized, domiciled, managing business, or located in, or that is a national any of the states listed in <u>Schedule 4.5(b)</u> hereof (each, an "<u>Enemy State</u>"), (ii) a Governmental Body of or within an Enemy State, or (iii) a Person that is supervised by an Enemy State.
- (c) For the avoidance of doubt, such Purchaser has not made and is not making any representations or warranties with respect to, and does not guarantee that, and nothing herein or otherwise shall be construed or interpreted as (or deemed) a representation or warranty with respect to, or a guarantee that, the Control Permit allowing the consummation of the transactions contemplated by this Agreement will be obtained and any deemed representations, warranties or other promises or commitments to the contrary are hereby expressly disclaimed.
- 4.6 <u>No Operations in Israel</u>. Solely with respect to Purchaser A, such Purchaser (and any affiliated entity (other than the Local Sponsor and it affiliated entities), as would be deemed "affiliated" for the purposes of the RTP Law) has no operations or business activities or "sales turnover" (as such term is used in Section 17(a)(2) of the RTP Law) in the State of Israel, except as set forth on Schedule 4.6.
- 4.7 <u>Investment Experience</u>. Without limiting any of the representations, warranties, covenants or agreements of the Seller contained in this Agreement or the other Transaction Documents or fraud claims, such Purchaser has such knowledge and experience in financial and business matters, and is capable of evaluating the merits and risks of the transactions contemplated by this Agreement.
- 4.8 Own Account. such Purchaser is acquiring the Purchased Shares and the Subscribed Shares to be acquired thereby solely for its own account and not as a nominee, agent or trustee of any third party.
- 4.9 <u>Waiver and Exemption</u>. Such Purchaser does not object to any waiver or exemption being granted to the Relevant Parties in connection with the Court Approval.

4.10 No Other Representations or Warranties.

(a) Such Purchaser acknowledges and agrees that, except for the representations and warranties contained in <u>Section 3</u> above and without limiting fraud claims, the transactions contemplated hereby or in any other Transaction Document shall be consummated without reliance

by such Purchaser on any other representations and warranties by the Seller or the Company with respect to the Purchased Shares or the Subscribed Shares, the Company, and its businesses, operations, rights, assets, liabilities or prospects, the condition or legal status of any of the above or otherwise.

- (b) Subject to and without limiting any of the representations and warranties of the Seller or the Company set forth in Section 3.1(g)(i) or Section 3.2(g), such Purchaser acknowledges and agrees that such Purchaser has not relied on, and the Seller and Company do not make nor has any of them made, any representation or warranty, either express or implied, whether written or oral, concerning Bezeq or any of its subsidiaries or any of their respective businesses, operations, assets, liabilities, Proceedings (including by regulatory authorities and other Governmental Bodies in Israel or outside Israel), results of operations, condition (financial or otherwise) or prospects.
- 4.11 <u>Due Diligence</u>; <u>Independent Decision</u>. Subject to and without limiting any of the representations, warranties, covenants or agreements of the Seller or the Company contained in this Agreement or fraud claims, such Purchaser represents that (a) it conducted a due diligence process during which it had an opportunity to ask questions and to receive answers from the Company and the Seller, regarding all Company and Bezeq business, activities and financial situation, (b) it has had access to, and an adequate opportunity to review, financial and other information, documents and materials as it deemed necessary to make its decision to purchase the Purchased Shares and the Subscribed Shares, (c) it has made its own assessment and has satisfied itself concerning all considerations relevant to the transactions contemplated hereunder, and (d) it has made its own independent decisions to enter into the transactions contemplated by this Agreement and as to whether they are appropriate or proper for it, based upon its own judgment and upon advice from such advisers as it has deemed necessary.
- Forward-looking Statements. In connection with the due diligence investigation of the Company and Bezeg by such Purchaser, such Purchaser has received and may continue to receive after the date hereof from the Seller and the Company certain estimates, projections, forecasts and other forward-looking information, as well as certain business plan information, regarding the Company and Bezeq and their respective Affiliates and subsidiaries and their respective business and operations. Such Purchaser hereby acknowledges and agrees (a) that there are uncertainties inherent in attempting to make such estimates, projections, forecasts and other forward-looking statements, as well as business plans, (b) to take full responsibility for making its own evaluation of the adequacy and accuracy of all such estimates, projections, forecasts and other forward-looking statements, as well as such business plans, so furnished to it (including the reasonableness of the assumptions underlying such estimates, projections, forecasts, forwardlooking statements or business plans), and (c) that none of the Company, Bezeq or the Seller, has made or is making any express or implied representation or warranty with respect to such estimates, projections, forecasts, forward-looking statements or business plans (including the reasonableness of the assumptions underlying such estimates, projections, forecasts, forward-looking statements or business plans).
- 4.13 <u>Brokers' Fees</u>. No agent, broker, investment banker, person or firm acting in a similar capacity on behalf of or under the authority of such Purchaser is or will be entitled to any broker's or finder's fee or any other commission or similar fee, directly or indirectly, in connection with the sale and transfer of the Shares as contemplated under this Agreement, other than Citigroup Global Markets Limited and RBC Capital Markets, LLC, for which the Purchasers shall be solely responsible.

5. COVENANTS

5.1 <u>Control Permit; Filings; Cooperation</u>.

- As soon as practicable following the date of this Agreement (and in any event (a) no later than ten (10) Business Days following the date of this Agreement), the Purchasers shall file an application to the Israeli Ministry of Communication for the purpose of obtaining the Control Permit in accordance with the applicable Legal Requirements. Each Party shall take all necessary actions and use its reasonable best efforts in good faith, to file, without any undue delay, all notices, reports and other documents required to be filed by such Party with any Governmental Body in order to obtain the Control Permit and the Antitrust Clearance with respect to the transactions contemplated by this Agreement, to submit, without any undue delay, any additional information requested by any such Governmental Body to effect all applications, notices, petitions and filings and to obtain as promptly as possible the Control Permit and the Antitrust Clearance. limiting the generality of the foregoing, each of the Seller and the Company shall, subject to applicable Legal Requirements, use its reasonable best efforts in good faith to cause Bezeg and its subsidiaries to cooperate with each Party's efforts to obtain the Control Permit, including to promptly answer any questions or inquiries of, and promptly submit any information requested by, any Governmental Body with respect to the transactions contemplated by this Agreement or the Control Permit to be issued to the Purchasers.
- Without derogating from the above, each Party hereby agrees (as applicable) (i) to provide any Governmental Body with any and all information regarding the corporate and holding structure of such Party, the ultimate shareholders of such Party and the various holdings, business affairs and interests of such Party and any affiliate thereof throughout the world as may reasonably be requested by such Governmental Body in connection with the Control Permit, (ii) to waive any confidentiality or other restriction (other than attorney-client or similar privilege) in connection with any examination or investigation by a Governmental Body concerning such Party and any 'Interested Party' (as such term is defined in the Communications Law) thereof as may reasonably be requested by such Governmental Body in connection with the Control Permit, (iii) to reasonably cooperate with the other Parties and allow a counsel of the other Parties to participate in any material meeting with a Governmental Body with respect of the request to obtain a Control Permit to the extent permitted by the applicable Governmental Body, (iv) to allow review and comment and shall consult in good faith with the other Parties on all material filings submitted to a Governmental Body in connection with the request to obtain the Control Permit prior to the submission thereof, provided that such Party may redact competitively sensitive or confidential information from the materials provided to the other Parties for review, (v) to provide the other Parties with a copy of any letters and/or other material correspondence with any Governmental Body in connection with the Control Permit, provided that such Party may redact competitively sensitive or confidential information from the materials provided to the other Parties for review, and (vi) to keep the other Parties reasonably updated, on a timely basis, of the status of matters relating to the discussions with the Governmental Bodies, including, if reasonably requested by the other Parties, on weekly basis, a telephonic update on the status and progress of obtainment of the Control Permit. For the avoidance of doubt, nothing in this Section 5.1 shall in any way limit or restrict the rights of the Parties pursuant to Section 8.1(d).
- (c) Subject to and without limiting <u>Section 5.1(f)</u>, (A) each Purchaser undertakes not to, and further undertakes to use its commercially reasonable efforts to cause any of its 'Interested Parties' (as such term is defined in the Communications Law) (including the Sponsors) not to, engage in or take any action that would reasonably expected to prevent it from complying with the Legal Requirements pertaining to the obtaining of the Control Permit, and to cause any of

its 'Interested Parties' not to engage in any transactions, or otherwise have any trade, commercial, financial or other relationships with (i) any Person organized, domiciled, managing business, or located in, or that is a national of any Enemy State, (ii) a Governmental Body of or within an Enemy State, or (iii) a Person that is supervised by an Enemy State; (B) each Purchaser shall be willing to accept, and shall not assert that it will refuse to accept, for any reason, a Control Permit on substantially the same terms as the Current Control Permit; and (C) shall not condition the acceptance of a Control Permit on terms which are not included in the Current Control Permit or which are more favorable to such Purchaser than those contained in the Current Control Permit.

- (d) Each of the Parties hereto shall, as promptly as practicable and before the expiration of any relevant legal deadline, file with any other Governmental Body, any filings, reports, information and documentation required for the transactions contemplated hereby pursuant to the Antitrust Law. Each of the Parties shall furnish to each other's counsel such necessary information and reasonable assistance as the other may request in connection with its preparation of any filing or submission that is necessary under the Antitrust Law, <u>provided</u> that a Party may redact competitively sensitive or confidential information from the materials provided to the other Party for review. Each of the Parties commits to instruct its counsel to cooperate with each other Party and its counsel and use reasonable best efforts to facilitate and expedite the identification and resolution of any such issues and, consequently, the expiration of the applicable waiting periods under the Antitrust Law at the earliest practicable dates.
- (e) Each of the Parties shall keep the other Parties reasonably apprised, in a timely manner, to the best of its Knowledge, with respect to any discussions between the Purchasers, the Seller, the Company or Bezeq and its subsidiaries, on the one hand, and any Employee Body, on the other hand. Without limiting the generality of the foregoing, each of the Parties shall notify the other Parties in writing of any material development in the negotiation with any Employee Body of Bezeq and its subsidiaries, promptly upon obtaining knowledge thereof.
- Notwithstanding anything in this Agreement to the contrary, nothing in this Agreement shall be deemed to require the Purchasers to agree to any condition, or take any action, that constitutes a Burdensome Condition, and neither the Seller nor the BComm Companies shall, without the prior written consent of the Purchaser A, agree to any Burdensome Condition. For purposes of this Agreement, a "Burdensome Condition" means any conditions imposed by, commitment or undertaking made to, or any Order of, any Governmental Body in connection with or related to the Control Permit or the transactions contemplated by this Agreement, that (i) individually or in the aggregate, has, or would reasonably be expected to have, a material and adverse effect on either Purchaser, after giving effect to the transactions contemplated by this Agreement, (ii) would impose, grant or otherwise require that either Sponsor, with respect to the investment contemplated by this Agreement, exercises or possesses any rights or powers to direct or influence (or other means of control) that are, individually or in the aggregate, greater than, or otherwise disproportionate to, such Sponsor's Pro Rata Share, or (iii) would bind or impose any obligation on any of the SCP Entities or the Local Sponsor and its Affiliates (other than the Purchasers) or their respective businesses (other than solely with respect to the SCP Entities operation and business in the State of Israel); provided that conditions that are expressly included in the Current Control Permit, the Communication Law and the Communication Order, in each case, as each of them is in effect as of the date of this Agreement (expressly disregarding any subsequent amendments, supplements, changes or other modifications, however effected or enacted), shall not be deemed Burdensome Conditions.
- (g) Each Party shall take all reasonable and necessary actions and use its commercially reasonable efforts in good faith, to file, without any undue delay all notices, reports

and other documents required to be filed by such Party with any Governmental Body in order to allow the pledge of the Bezeq Shares for the benefit of the Company's bondholders pursuant and subject to Company Debt Modifications (the "Share Pledge"). Such efforts shall require the Purchasers to (i) file, following the date of this Agreement (without undue delays), a formal request with the Israeli Ministry of Communication to authorize the Share Pledge (and to timely respond to subsequent requests for information by the Israeli Ministry of Communication in connection therewith), and (ii) coordinate with the Company and representatives of its bondholders in respect of matters related to the Share Pledge. For the avoidance of doubt, nothing in this Section 5.1(g) shall in any way limit or restrict the rights of any Party pursuant to Section 8.1(d). For the avoidance of doubt, in any event, the Company Series C Debentures and the Series D Debentures shall include negative pledge covenants, in accordance with the terms thereof and the Company Debt Modifications.

5.2 Conduct of Business.

- (a) From and after the date hereof and until the earlier of the Closing or the termination of this Agreement in accordance with Section 8.1 (the "Interim Period"), without limiting anything contained in Section 5.2(c), and unless otherwise agreed to in advance and in writing by Purchaser A (such agreement not to be unreasonably withheld) each of the Seller and the BComm Companies shall use (subject to applicable fiduciary duties under applicable Legal Requirements) its entire voting rights in the Company and Bezeq, respectively (in case such matters are brought to a shareholders' vote) so that the BComm Companies and Bezeq and its subsidiaries shall, not take, do, effect or allow (or commit or agree to take, do, effect or allow) any of the Restricted Matters (except as set forth in Schedule 5.2(a) hereof).
- (b) During the Interim Period, without limiting anything contained in <u>Section</u> <u>5.2(c)</u>, the Company shall, and the Company shall cause the other BComm Companies to:
 - (i) conduct its business in the ordinary course and in compliance with applicable Legal Requirements; and
 - (ii) except as set forth in <u>Schedule 5.2(b)(ii)</u> hereof, not take, do, effect or allow (or commit or agree to take, do, effect or allow) any of the Restricted Matters, unless otherwise consented to in advance and in writing by Purchaser A (such consent not to be unreasonably withheld).
- (c) Until the end of the Interim Period, without the prior written consent of Purchaser A (such consent not to be unreasonably withheld) (i) the Seller shall not, directly or indirectly, sell, transfer, convey, exchange, assign, gift, Encumber, or otherwise dispose of any of the Purchased Shares or any rights therein or thereto, (ii) the Company shall not, directly or indirectly, sell, transfer, convey, exchange, assign, gift, Encumber, or otherwise dispose of any of the Company Owned Securities or any rights therein or thereto, (iii) the Company shall cause SP1 not to (and the Company shall be responsible if SP1 does), directly or indirectly, sell, transfer, convey, exchange, assign, gift, Encumber, or otherwise dispose of any of the SP2 Securities or the SP1 Bezeq Shares, and (iv) the Company shall cause SP2 not to (and the Company shall be responsible if SP2 does), directly or indirectly, sell, transfer, convey, exchange, assign, gift, Encumber, or otherwise dispose of any of the SP2 Bezeq Shares.

5.3 Exclusivity; Acquisition Proposal.

(a) During the Interim Period, neither the Seller nor the Company shall, and the Company shall cause each of the BComm Companies not to, and each of them shall use its

reasonable best efforts to cause their respective directors, officers, employees, and advisors (including, without limitation, financial advisors, bankers, attorneys, accountants, consultants) and other representatives (it being understood that the representatives of the bondholders of the Seller and of the Company, respectively, shall not be considered their representatives unless acting at the direction of, the Seller or the BComm Companies, respectively) (collectively, "Representatives") not to, (i) solicit, initiate, knowingly encourage or knowingly facilitate any Acquisition Proposal or any inquiry, proposal or offer that could reasonably be expected to lead to any Acquisition Proposal or the making or consummation thereof, (ii) other than to inform any Person of the existence of the provisions contained in this Section 5.3(a), enter into, continue or otherwise participate in any discussions or negotiations regarding, or furnish to any Person any information in connection with, or enter into any agreement with respect to, any Acquisition Proposal or any inquiry, proposal or offer that could reasonably be expected to lead to any Acquisition Proposal, or (iii) enter into any Contract with respect to any Acquisition Proposal. Without limiting the generality of the foregoing, to the extent not already done, each of the Seller and the Company shall, and shall use its reasonable best efforts to cause its Representatives to, immediately cease and cause to be terminated any existing activities, including discussions or negotiations with any Person (other than the Purchasers and their Affiliates) with respect to any Acquisition Proposal.

- (b) Without limiting the forgoing, promptly, and in no event later than twenty four (24) hours after its receipt (including receipt by any of its Representatives) of any Acquisition Proposal, the Seller or the Company, as applicable, shall advise Purchaser A in writing of such Acquisition Proposal (including the material terms and conditions of such Acquisition Proposal and the identity of the Person making or submitting such Acquisition Proposal) and shall provide Purchaser A a copy of such Acquisition Proposal and any related draft agreements or other documentation or materials delivered in connection therewith. The Seller or the Company, as applicable, shall keep Purchaser A informed on a reasonably prompt basis with respect to any material change to the material terms of any such Acquisition Proposal (and in no event later than twenty four (24) hours following any such change).
- (c) The Parties acknowledge that either the Seller or the Company may be subject to disclosure requirements under applicable Legal Requirements and no Party shall have any claim with respect to such disclosure.
- (d) A breach of this <u>Section 5.3</u> by any Representative of the Seller or the Company shall be deemed a breach by the Seller or the Company, as applicable, and the Seller or the Company, as the case may be, shall be liable therefor.
- (e) Nothing in this Agreement shall be deemed to prevent, restrict or limit in any way, any arrangement among the Seller and its creditors, as long as such arrangement (i) does not prevent, conflict or is otherwise inconsistent with, directly or indirectly, the Seller's obligations hereunder or thereunder, and (ii) does not constitute an alternative to the transactions contemplated by this Agreement (and such arrangement shall not be considered an "Acquisition Proposal").

5.4 [RESERVED].

5.5 Directors.

(a) At least ten (10) Business Days prior to the Closing Date, Purchaser A shall deliver to the Seller and the Company a list of individuals to be appointed as members of the board of directors of the Company as of the Closing (the "Company Designated Directors"). The Company Designated Directors shall satisfy the qualification requirements under Legal Requirements applicable to board members of the Company.

- (b) At least ten (10) Business Days prior to the Closing Date, Purchaser A shall deliver to the Seller and the Company a list of two (2) individuals to be appointed as members of the board of directors of Bezeq as of the Closing (the "Bezeq Designated Directors"). The Bezeq Designated Directors shall satisfy the qualification requirements under Legal Requirements applicable to board members of Bezeq.
- 5.6 Notification. During the Interim Period, (a) each of the Parties shall promptly notify the other Parties if it becomes aware of any event that would make the satisfaction of any of the conditions in Section 6 impossible or unlikely to be satisfied as of the End Date, and (b) each of the Seller and the Company shall, upon reasonable request of Purchaser A, and subject to applicable fiduciary duties under Legal Requirements, update Purchaser A promptly following the occurrence of any material developments in any of the BComm Companies, Bezeq or any of its subsidiaries which are within the Knowledge of the Seller or the Knowledge of the Company, as the case may be, including with respect to regulatory matters or developments, all material matters related to the Israeli Ministry of Communications and any other material commercial matters. This Section 5.6 and any notification provided pursuant hereto shall not modify or limit in any way the Parties' representations, warranties, covenants, obligations, rights and remedies under this Agreement. No breach of this Section 5.6 shall be deemed a breach of this Agreement, including for the purposes of Section 6.2(c), unless such breach is made knowingly, willfully and intentionally.

5.7 Equity Offering.

- (a) As soon as practicable after the date of this Agreement, subject to the provisions of this Section 5.7, the Company shall offer the shareholders of the Company (other than the Seller) (the "Eligible Shareholders") the right to subscribe for a total of 8,383,234 newly issued Ordinary Shares of the Company (the "Additional Subscription Shares") at a price per share equal to the Subscription Price Per Share (the "Additional Equity Offering").
- Requirements in order to effect the Additional Equity Offering in accordance with the terms of this Agreement, including the filing of a registration statement with the SEC and a prospectus with the ISA, in each case to the extent required by under applicable Legal Requirements, and to receive all permits, approvals and orders required for the consummation of the Additional Equity Offering, including the receipt of all permits and approvals in order to list the Additional Subscribed Shares on the TASE and Nasdaq. The Company shall take all necessary actions and use its reasonable best efforts in good faith, to file, without any undue delay, all required filings with any Governmental Body in order to effect the Additional Equity Offering, to submit, without any undue delay, any additional filings requested by any such Governmental Body to effect the Additional Equity Offering and take all other actions required in order to effect as promptly as possible the Additional Equity Offering.
- (c) The Additional Equity Offering shall expire no later than twenty (20) Business Days prior to the Closing Date, and any Additional Subscription Shares not irrevocably subscribed for by the Eligible Shareholder in accordance with the foregoing prior to or on such date (such shares, the "<u>Unsubscribed Shares</u>") shall no longer be available for subscription by the Eligible Shareholders.
- (d) Upon the terms and subject to the conditions of this Agreement, effective as of, and conditioned upon, the Closing, Purchaser A subscribes for all of the Unsubscribed Shares (if any, including any Unsubscribed Shares arising due to the failure of the Company to complete its undertaking under Section 5.7(b) at a price per share equal to the Subscription Price Per Share, to

be issued thereto in accordance with the terms of this Agreement at, and conditioned upon, the Closing.

- (e) For the avoidance of doubt, nothing in the <u>Section 5.7</u> shall in any way limit, modify or otherwise affect the obligations of the Seller pursuant to <u>Section 2.4</u>.
- Court Approval. (a) As soon as practicable after the date hereof (and in any event no later than ten (10) Business Days following the date of this Agreement), the Seller and the Company shall jointly approach the Court, in order to obtain the Court Approval, and subject to and in accordance with the approval of the Court, each of the Seller and the Company shall establish a record date for, call, publicize the convening of, convene and hold a meeting of the bondholders of the Company and the Seller, respectively (the "Stakeholders' Meetings"), (b) the Seller and the Company, respectively, shall timely prepare and file all notices required by applicable Legal Requirements in connection with the Stakeholders' Meetings, it being agreed and understood that this Agreement and all of the Transaction Documents and the Company Debt Modifications and the amended deed of trust of the Company Series C Debenture and the deed of trust for the Series D Debentures, should be brought to the approval of the Stakeholders' Meetings of the Company, (c) each of the Seller and the Company shall take all necessary actions and make all reasonable efforts in order to receive the Court Approval and shall respond to and comply with, without undue delay, any queries, requests or instructions of the Court within the Court Approval process, (d) each of the Seller and the Company shall allow Purchaser A (and its representative, including its legal counsel) to review and comment on all material materials and filings submitted to the Court (or any other Governmental Body) only in connection with its respective Court Approval prior to the submission thereof, and shall incorporate into and reflect in any such submissions or filings any and all reasonable comments made by Purchaser A or its representatives, provided such implementation shall not cause a delay in the relevant submission, and (e) each of the Seller and the Company shall provide Purchaser A (and its legal counsel) with a copy of any material letters and/or other material correspondence and decisions (including any interim decisions) of the Court as part of the Court Approval process, and shall share, in cooperation with Purchaser A, all correspondence with, and decisions of, the Court or any bondholders (and in the case of the Company also shareholders) and shall keep Purchaser A (and its legal counsel) reasonably fully updated of the status of matters relating to the Court Approval (including the discussions with its bondholders and shareholders and any other creditors or relevant parties), including, if reasonably requested by Purchaser A, on weekly basis, a telephonic update on the status and progress of obtaining its respective Court Approval, and bondholders and shareholders approvals, as applicable. For the avoidance of doubt, nothing in this Section 5.8 shall in any way limit or restrict the rights of Purchaser A pursuant to Section 8.1(f). Without limiting anything herein contained, each of the Parties shall use its commercially reasonable efforts to obtain through the Court Approval a release of any lock-up restrictions (הסימה וטפטוף) that might otherwise be applicable to the securities being issued hereunder by the Company at the Closing.
- 5.9 Special Shareholder Meeting of the Company. As soon as practicable following the date of this Agreement (and in any event no later than ten (10) Business Days following the date of this Agreement), the Company shall file a motion with the Court requesting that the court issue an order to convene a special meeting of the shareholders of the Company (the "Company Shareholders Meeting") for the purpose of approving this Agreement and the transactions contemplated hereby (including the sale and issuance of the Adjustment Subscribed Shares (if any), the Purchaser Subscribed Shares, the Additional Purchaser Subscribed Shares (if any), the Seller Subscribed Shares (if any), the Election Subscribed Shares (if any) and the Seller Subscribed Debentures hereunder) (the "Shareholders Approval"). Promptly following the Court issuing such order, and in any event within three (3) Business Days following the date of such order, the Company shall

publish a notice calling the Company Shareholders Meeting to be held on the date that is no later than thirty five (35) days following such notice for the purpose of obtaining the Shareholders Approval. The Company shall solicit from the shareholders of the Company written ballots and proxies in favor of the Shareholders Approval in accordance with applicable Legal Requirements, and shall use its reasonable best efforts to secure the Shareholders Approval at the Company Shareholders' Meeting. The Company shall prepare and file all notices required by applicable Legal Requirements in connection with the Company Shareholders' Meeting, and shall respond promptly to all inquiries of any Governmental Body in connection therewith. For the avoidance of doubt, nothing in this Section 5.9 shall in any way limit or restrict the rights of Purchaser A pursuant to Section 8.1(f).

- 5.10 <u>TASE Approval</u>. Each of the Parties shall use its reasonable best efforts to obtain, as promptly as practicable after the date of this Agreement, the approval of the TASE to the issuance, at the Closing, of the Adjustment Subscribed Shares (if any), the Purchaser Subscribed Shares, the Additional Purchaser Subscribed Shares (if any), the Seller Subscribed Shares (if any), the Additional Subscription Shares and the Seller Subscribed Debentures, and to the adoption of the Company Debt Modifications.
- 5.11 <u>Debt Covenants</u>. Following the Closing, the Purchasers shall not cause the Company to (a) enter into any transaction with the Purchasers or any Affiliate of the Purchasers including payment of management fees other than expenses reimbursement and standard directors compensation, or (b) effect any dividend distributions, in either case, to the extent such transaction or dividend distributions would violate, and cause the Company to be in default under, the terms of its Series C Debentures (as modified by the Company Debt Modifications).
- 5.12 <u>Liquidity Information</u>. At least five (5) Business Days prior to the Closing, the Company shall deliver to the Seller and Purchaser A a reasonably detailed calculation of the anticipated Available Cash, along with reasonable documentation supporting such calculation, and allow representatives of the Seller or Purchaser A to discuss such calculation and supporting documentation with an authorized officer of the Company; <u>provided</u> that notwithstanding anything herein or otherwise to the contrary, except in the event of fraud (and then solely against the Person committing such fraud), neither the Seller nor any of its Non-Recourse Parties shall have any claims, rights or recourse against any Person (including any of the BComm Companies or the Purchasers or any of their respective Non-Recourse Parties) in respect of such calculation of Available Cash or any of the information or documentation provided pursuant to, or otherwise for the failure to comply with, this <u>Section 5.12</u>. For the avoidance of doubt, and notwithstanding anything to the contrary, nothing in this <u>Section 5.12</u> shall limit, modify or otherwise affect any Purchaser's rights under this Agreement (including under <u>Section 6.2</u>).

6. CONDITIONS PRECEDENT

- 6.1 <u>Conditions to the Obligations of the Seller, the Company and Purchasers.</u> The obligations of each of the Seller, the Company and the Purchasers to consummate the transactions contemplated by this Agreement is subject to the satisfaction as of the Closing of all of the following conditions precedent (any of which may be waived in whole or in part by a mutual agreement in writing of the Seller, the Company and the Purchasers):
- (a) <u>Control Permit</u>. A Control Permit was issued and is in full force and effect permitting the Purchasers to acquire the Purchased Shares, the Adjustment Subscribed Shares (if any), the Purchaser Subscribed Shares, the Additional Purchaser Subscribed Share (if any) and the

Election Subscribed Shares (if any) which Control Permit does not include any Burdensome Conditions

- (b) <u>Antitrust Commissioner's Approval</u>. The Antitrust Clearance shall have been obtained.
- (c) <u>No Injunction</u>. There shall not be in effect any injunction or other Order or Legal Requirement that (i) prohibits, enjoins or restrains the consummation of the transactions contemplated by this Agreement, and (ii) has been adopted or issued, or has otherwise become effective.
 - (d) <u>Court Approval</u>. The Court Approval shall have been obtained.
- (e) <u>TASE Approval</u>. The TASE has approved the issuance of the Adjustment Subscribed Shares (if any), the Purchaser Subscribed Shares, the Additional Purchaser Subscribed Shares (if any), the Seller Subscribed Shares (if any), the Election Subscribed Shares (if any), the Additional Subscription Shares and the Seller Subscribed Debentures and Series D Debentures, and the adoption of the Company Debt Modifications.
- 6.2 <u>Conditions to the Obligations of the Purchasers</u>. The obligations of the Purchasers to consummate the transactions contemplated by this Agreement is subject to the satisfaction as of the Closing of all of the following additional conditions precedent (any of which may be waived in whole or in part by Purchaser A in its sole discretion):
- (a) <u>Seller Representations and Warranties</u>. Each of the representations and warranties of the Seller contained in <u>Section 3.1</u> shall be true and correct in all respects, as of the date of this Agreement and as of the Closing Date, as if made at and as of the Closing Date, except for any representations and warranties specifically made as of a particular date, in which case such representations and warranties shall be true and correct in all respects as of such date.
- (b) <u>Company Representations and Warranties</u>. Each of the representations and warranties of the Company contained in <u>Section 3.2</u> shall be true and correct in all respects, as of the date of this Agreement and as of the Closing Date, as if made at and as of the Closing Date, except for any representations and warranties specifically made as of a particular date, in which case such representations and warranties shall be true and correct in all respects as of such date.
- (c) <u>Performance of the Obligations of the Seller and Company</u>. Each of the Seller and the Company shall have performed and complied in all material respects with all agreements and covenants required by this Agreement to be performed or complied with by the Seller or the Company, as applicable, at or prior to the Closing.
- (d) <u>No Material Adverse Effect; No Debt Instrument Default</u>. (i) As of the Closing Date there shall not exist any Material Adverse Effect, and (ii) there shall not be (for the purpose of this clause (ii), as of the Closing) any Effect that would reasonably be expected to constitute an event of default under the BComm Companies' debt instruments (including under the Company Series B Debentures or the Company Series C Debentures).
- (e) <u>Company Board of Directors</u>. Each of the Company Designated Directors shall have been appointed as a member of the board of directors of the Company effective as of the Closing, and (ii) effective as of the Closing, the Company Designated Directors constitute a majority of the directors on the board of directors of the Company.

- (f) <u>Bezeq Board of Directors</u>. (i) The total number of directors on the board of directors of Bezeq as of the Closing Date shall not exceed nine (9), and (ii) at least two (2) of the Bezeq Designated Directors shall have been appointed as member of the board of directors of Bezeq effective as of the Closing.
- (g) <u>Nasdaq and TASE Listing</u>. (i) The Company, Bezeq, Nasdaq or the TASE shall not have announced on or after the date of this Agreement a delisting or threatened (in writing by Nasdaq or the TASE) delisting of the Company's or Bezeq's Equity Securities from Nasdaq or the TASE, as applicable, (ii) (x) the Company is in compliance with all continued listing requirements of Nasdaq and the TASE, and (y) Bezeq is in compliance with all continued listing requirements of the TASE, and (iii) no suspension of trading specific to the Company's or Bezeq's Equity Securities on the Nasdaq or the TASE, as applicable, shall have occurred for more than two (2) trading days.
- (h) <u>Capitalization</u>. (i) None of the Bezeq Shares has been transferred, sold, granted, delivered or otherwise disposed of, or is subject to any Encumbrance, (ii) none of the Effects set forth in causes (a), (b), (c), (d) or (e) of the definition of Restricted Matters has occurred, from and after the date of this Agreement, with respect to any of the BComm Companies or their respective Equity Securities, and (iii) the Company Owned Securities shall represent, as of the Closing, direct and indirect holding of at least twenty five and a tenth percent (25.1%) of Bezeq's issued and outstanding share capital.
- (i) <u>Cash; Indebtedness</u>. (i) The Company shall have, as of immediately prior to the Closing (without giving effect to the transactions contemplated by this Agreement, but taking into account all Transaction Expenses of the BComm Companies) at least NIS 680,000,000 of Available Cash, (ii) the consolidated amount of indebtedness (including for borrowed money or as evidenced by bonds, debentures, notes or other debt instruments) of the BComm Companies, determined in accordance with IFRS, shall not exceed two billion, four hundred and sixty six million, forty eight thousand and three hundred and thirty three New Israeli Shekels (NIS 2,466,048,333) (par value), and (iii) from and after the date of this Agreement, no BComm Company has entered into any transaction (including entering into any new Contract or amending or modifying any existing Contract) in connection with debt restructuring, debt refinancing, creditors arrangement or otherwise affecting its debt Contracts or securities, other than the Company Debt Modifications.
- (j) <u>BComm Debt Modifications</u>. The Company Debt Modifications have been duly approved in accordance with the terms of all applicable Contracts (including the Company Series B Debentures and Company Series C Debentures) and all applicable Legal Requirements, and are in full force and effect as of the Closing.
- (k) <u>Shareholders Approval</u>. The Shareholders Approval shall have been obtained; <u>provided</u> that the condition to Closing specified in this <u>Section 6.2(k)</u> shall be deemed satisfied if the Court Approval has been obtained and such Court Approval has become final and nonappealable (and, for the avoidance of doubt, no appeal is pending).
- 6.3 <u>Conditions to the Obligations of Seller and Company</u>. The obligation of the Seller and the Company to consummate the transactions contemplated by this Agreement is subject to the satisfaction as of the Closing of all of the following additional conditions precedent (any of which may be waived in whole or in part by notice in writing of the Seller and the Company in their sole discretion):
- (a) <u>Purchaser Representations and Warranties</u>. Each of the representations and warranties of the Purchasers contained in <u>Section 4</u> shall be true and correct in all material respects

as of the date of this Agreement and as of the Closing Date, as if made as of the Closing Date, except for any representations and warranties specifically made as of a particular date, in which case such representations and warranties shall be true and correct in all material respects as of such date.

- (b) <u>Performance of the Obligations of the Purchasers</u>. The Purchasers shall have performed and complied in all material respects with all agreements and covenants required by this Agreement to be performed or complied with by the Purchasers at or prior to the Closing.
- 6.4 <u>Material Adverse Effect</u>. To the extent Purchaser A wishes to claim that the condition to Closing set forth in <u>Section 6.2(d)</u> has not been satisfied as a result of the occurrence of a Material Adverse Effect, it shall provide the Seller and the Company a written notice to that effect, signed by a duly authorized representative of Purchaser A. For the avoidance of doubt, no failure or delay of Purchaser A to provide a notice pursuant to this <u>Section 6.4</u> shall in any way affect or otherwise impact the determination of whether a Material Adverse Effect has occurred or whether or not any of the conditions to the Closing has been satisfied.

7. CLOSING

- 7.1 <u>The Closing</u>. The consummation of the transactions contemplated under this Agreement shall take place at a closing (the "<u>Closing</u>") at the offices of GKH law firm, 1 Azrieli Center, Tel-Aviv, Israel, at 12:00 (Israel time) on the thirteenth (13th) Business Day following the date on which the last of the conditions set forth in <u>Section 5.9</u> is fulfilled or waived, or at such other location, time and date as shall be mutually agreed upon in writing between the Parties (the "<u>Closing Date</u>").
- 7.2 <u>Transactions at the Closing</u>. At the Closing, the following transactions shall occur, which transactions shall be deemed to occur simultaneously and no transaction shall be deemed to have been completed or any document delivered until all such transactions have been completed and all required documents delivered:
- (a) <u>Delivery of Securities, Documents and Actions by the Seller</u>. The Seller shall deliver to the Purchasers the following securities and documents, and shall take the following actions:
 - (i) Without limiting anything contained in <u>Section 2.1</u>, the Seller shall deliver to the Purchasers (allocated among them on a pro rata basis, based on their respective Pro Rata Share) all of the rights, title and interest in, and cause the transfer of, the Purchased Shares, free and clear of any and all Encumbrances, to such bank or brokerage account of each Purchaser maintained in its name at a bank or brokerage firm in Israel, details of which will be provided by each Purchaser at least two (2) Business Days prior to the Closing Date, against the payment of such Purchaser Pro Rata Share of the Seller Purchase Price as forth herein.
 - (ii) The Seller shall deliver to the Purchasers (with a copy to the Company) executed resignations, effective as of the Closing Date, of each director of the Company or Bezeq set forth on Schedule 7.2(a)(ii)(A), in the forms attached as Schedule 7.2(a)(ii)(B).
 - (iii) The Seller shall deliver to the Purchasers a certificate executed by a duly authorized officer of the Seller and dated as of the Closing Date, certifying that each of the conditions set forth in Sections 6.2(a), 6.2(c) (with respect to the Seller), 6.2(h) and 6.2(i) have been satisfied.

- (iv) The Seller shall pay, or cause to be paid, by wire transfer of immediately available funds, an amount equal to the Seller Contribution Amount, into a bank account maintained in the name of the Company in a bank in Israel, details of which will be provided by the Company at least two (2) Business Days prior to the Closing Date. It is understood and agreed that a portion of the Seller Contribution Amount equal to the Seller Purchase Price shall be paid directly to the Company by the Purchasers, on behalf of the Seller, pursuant to Section 7.3.
- (b) <u>Delivery of Securities, Documents and Actions by the Company</u>. The Company shall deliver the following securities and documents, and shall take the following actions:
 - (i) Without limiting anything contained in <u>Section 2.3</u>, the Company shall deliver to the Purchasers (allocated among them on a pro rata basis, based on their respective Pro Rata Share) all of the Adjustment Subscribed Shares (if any) and the Purchaser Subscribed Shares, free and clear of all Encumbrances, to such bank or brokerage account of each Purchaser maintained in its the name at a bank or brokerage firm in Israel, details of which will be provided by such Purchaser at least two (2) Business Days prior to the Closing Date.
 - (ii) Without limiting anything contained in <u>Section 2.3</u>, the Company shall deliver to Purchaser A all of the Additional Purchaser Subscribed Shares and Election Subscribed Share (if any), free and clear of all Encumbrances, to a bank or brokerage account maintained in the name of Purchaser A at a bank or brokerage firm in Israel, details of which will be provided by the Purchaser at least two (2) Business Days prior to the Closing Date.
 - (iii) The Company shall deliver to the Purchasers (with a copy to the Seller) executed resignations, effective as of the Closing Date, of each director of Bezeq set forth on Schedule 7.2(b)(iii)(A), in the forms attached as Schedule 7.2(b)(iii)(B).
 - (iv) Unless a Purchaser Election has been made, the Company shall deliver to the Seller all of the Seller Subscribed Shares, free and clear of all Encumbrances to a bank or brokerage account maintained in the name of the Seller at a bank or brokerage firm in Israel, details of which will be provided by the Seller at least two (2) Business Days prior to the Closing Date.
 - (v) The Company shall deliver to the Seller all of the Seller Subscribed Debentures, free and clear of all Encumbrances to a bank or brokerage account maintained in the name of the Seller at a bank or brokerage firm in Israel, details of which will be provided by the Seller at least two (2) Business Days prior to the Closing Date.
 - (vi) The Company shall repay in full all principal and interest (but without any default interest or any other penalty charges or payments) outstanding on the Company Series B Debentures as of such repayment date.
 - (vii) The Company shall deliver to the Purchasers a certificate executed by a duly authorized officer of the Company and dated as of the Closing Date, certifying that each of the conditions set forth in <u>Sections 6.2(b)</u>, <u>6.2(c)</u> (with respect to the Company), 6.2(h), 6.2(j) and 6.2(j) have been satisfied.
 - (viii) The Company shall issue to the Series C Bondholders existing on the record date of the Closing the Series D Debentures.

- (c) <u>Delivery of Documents and Actions by the Purchasers</u>. The Purchasers shall deliver, the following documents, and shall take the following actions:
 - (i) Each Purchaser shall pay, or cause to be paid, by wire transfer of immediately available funds, its Pro Rata Share of the Seller Purchase Price, in accordance with and subject to Section 7.3, into a bank account maintained in the name of the Company in a bank in Israel, details of which will be provided by the Company at least two (2) Business Days prior to the Closing Date.
 - (ii) Each Purchaser shall pay, or cause to be paid, by wire transfer of immediately available funds, its Pro Rata Share of the Company Purchase Price into a bank account maintained in the name of the Company in a bank in Israel, details of which will be provided by the Company at least two (2) Business Days prior to the Closing Date.
 - (iii) Purchaser A shall pay, or cause to be paid, by wire transfer of immediately available funds, the Additional Company Purchase Price (if any) and the Election Purchase Price (if any) into a bank account maintained in the name of the Company in a bank in Israel, details of which will be provided by the Company at least two (2) Business Days prior to the Closing Date.
 - (iv) Each Purchaser shall deliver to the Seller and the Company a certificate executed by such Purchaser and dated as of the Closing, certifying that the conditions set forth in <u>Section 6.3(a)</u> (with respect to itself only) and <u>Section 6.3(b)</u> (with respect to itself only) have been satisfied.
- 7.3 <u>Payment Directions</u>. The Seller hereby irrevocably directs and instructs each Purchaser, on the Seller's behalf, to transfer directly to the Company, the full amount of the Seller Purchase Price that the Seller would otherwise be entitled to receive at Closing pursuant to this Agreement (subject to <u>Section 2.2(e)</u>), towards satisfaction of the Seller's payment obligations to the Company pursuant to <u>Section 2.4</u> and <u>Section 7.2(a)(iv)</u>. All amounts so directly transferred to the Company pursuant to this <u>Section 7.3</u> shall be deemed, for all purposes, as having been paid to the Seller and immediately thereafter paid to the Company by the Seller.

8. TERMINATION

- 8.1 <u>Termination Events</u>. This Agreement may be terminated and the transactions contemplated hereby abandoned prior to the Closing:
 - (a) by written mutual consent of Purchaser A, the Company and the Seller;
- (b) by the Seller or the Company, if either Purchaser has breached any covenant or agreement contained in this Agreement, or if any representation or warranty of either Purchaser has become untrue, in each case, such that the conditions set forth in Section 6.1 or Section 6.3, as the case may be, would not be satisfied at a Closing on or prior to the End Date; provided, however, that neither the Seller nor the Company may terminate this Agreement pursuant to this Section 8.1(b) if any such breach or failure to be true has been cured within seven (7) Business Days after written notice by the Seller or the Company to Purchaser A informing Purchaser A of such breach or failure to be true, except that no cure period shall be required for a breach which by its nature cannot be cured prior to the End Date; provided, further, that neither the Seller nor the Company may terminate this Agreement pursuant to this Section 8.1(b) if either of them is then in breach of this Agreement in any material respect; further, provided, that, for the avoidance of doubt, nothing in this Section

8.1(b) shall in any way limit or otherwise modify the rights of Purchaser A pursuant to <u>Section</u> 8.1(f);

- (c) by Purchaser A, if the Seller or the Company has breached any covenant or agreement contained in this Agreement, or if any representation or warranty of the Seller or the Company has become untrue, in each case, such that the conditions set forth in Section 6.1 or Section 6.2, as the case may be, would not be satisfied as of the Closing on or prior to the End Date; provided, however, that Purchaser A may not terminate this Agreement pursuant to this Section 8.1(c) if any such breach or failure to be true has been cured within seven (7) Business Days after written notice by Purchaser A to the Seller and the Company informing the Seller and the Company of such breach or failure to be true, except that no cure period shall be required for a breach which by its nature cannot be cured prior to the End Date; and, provided, further, that Purchaser A may not terminate this Agreement pursuant to this Section 8.1(c) if the Purchasers are then in breach of this Agreement in any material respect;
- (d) by any of Purchaser A, the Company or the Seller if the Closing has not occurred on or before the End Date; <u>provided</u> that the End Date may be extended with the written consent of all the Parties by up to three (3) additional thirty (30)-day periods; and, <u>provided, further</u>, that the right to terminate this Agreement under this <u>Section 8.1(d)</u> shall not be available to (A) to Purchaser A if the Purchasers' breach of any provision of this Agreement causes the failure of the Closing to be consummated by the End Date, or (B) to the Seller or the Company if any breach by the Company or the Seller of any provision of this Agreement causes the failure of the Closing to be consummated by the End Date.
- (e) by any of Purchaser A, the Company or the Seller, if the consummation of the transactions contemplated hereby is permanently enjoined or prohibited by the terms of a final, non-appealable Order of a Governmental Body of competent jurisdiction.
- (f) By Purchaser A, if any of the Court Approval or the Shareholders Approval has not been obtained prior to or on the date that is sixty (60) calendar days after the date of this Agreement.

8.2 Notice of Termination; Effect of Termination.

- (a) A Party desiring to terminate this Agreement pursuant to <u>Section 8.1</u> shall give written notice of such termination to the other Parties, specifying the provision pursuant to which such termination is effective.
- (b) Notwithstanding anything in this Agreement or otherwise to the contrary, if this Agreement is validly terminated prior to Closing in accordance with Section 8.1, except in the event of fraud, the maximum aggregate liability of a Party, its Non-Recourse Parties and their respective successors and assigns for any and all Losses (whether at law, in equity, in contract, in tort or otherwise) in connection with, relating to or arising out of the Transaction Documents (including any breach or nonperformance thereof) or the transactions contemplated hereby and thereby, shall in no event exceed (i) in the case of the Company, the Company Liability Cap, (ii) in the case of the Seller, the Seller Liability Cap, or (iii) in the case of the Purchasers, the Purchaser Liability Cap (provided that no Purchaser shall be responsible in excess of its Purchaser's Pro Rata Share of the Purchaser Liability Cap).
- (c) Each Party acknowledges that the agreements contained in this <u>Section 8.2</u> are an integral part of the transactions contemplated by this Agreement, and that without these agreements, neither Party would enter into this Agreement.

- (d) Each Party's right of termination under <u>Section 8.1</u> is in addition to the specific performance rights as set forth in <u>Section 9.13</u>.
- (e) If this Agreement is validly terminated pursuant to <u>Section 8.1</u>, all further obligations of the Parties under this Agreement will terminate and there shall be no liability on the part of any Party, <u>except</u> that (i) the provisions of <u>Section 8.2</u> and <u>Section 9</u> hereof (to the extent applicable) shall survive the termination of this Agreement, and (ii) subject to and without limiting <u>Section 8.2(b)</u>, nothing herein shall relieve or release any Party from any liability to the other Party for any breach of any of its representations, warranties, covenants or agreements set forth in this Agreement or for fraud, in each case, occurring prior to such termination.
- (f) For the avoidance of doubt: (i) neither the Seller or the Company nor any of their Affiliates or Representatives is making any representations or warranties with respect to, and does not guarantee that, and nothing herein shall be construed or interpreted as (or deemed) a representation or warranty with respect to, or a guarantee that, the Court Approval will be obtained or that the Agreement will be approved by the Seller's or the Company's bondholders or shareholders and any deemed representations, warranties or other promises or commitments to the contrary are hereby expressly disclaimed; and (ii) without derogating from the obligations of the Seller or the Company hereunder, the occurrence of a Material Adverse Effect in and of itself shall not entitle the Purchasers to any remedy other than not to consummate transactions contemplated hereby; <u>provided</u> that nothing in this clause (ii) shall relieve or release any Party from any breach or noncompliance with any of its obligations under this Agreement.

9. MISCELLANEOUS

- 9.1 <u>Expenses; Taxes</u>. Each Party will bear its own expenses incurred in connection with the preparation, execution, and performance of this Agreement and the transactions contemplated hereunder, including all court proceedings, fees and expenses of agents, representatives, counsel, and accountants, and applicable value added tax. Except as expressly provided in this Agreement, all transfer, documentary, sales, use, stamp, registration and other such taxes and mandatory payments imposed by any Governmental Body, and all fees and charges (including any penalties and interest) incurred in connection with the consummation of the transactions contemplated hereunder shall be borne by each respective Party to the extent such Party is liable for such taxes, fees and charges, in accordance with applicable Legal Requirement.
- 9.2 <u>Further Assurances</u>. Each of the Parties shall perform such further acts and execute such further documents as may reasonably be necessary to carry out and give full effect to the provisions of this Agreement and the intentions of the Parties as reflected thereby.

9.3 Survival; Limitation on Liability; Indemnification.

- (a) <u>Survival</u>. The representations and warranties contained in this Agreement or in any certificates delivered pursuant to this Agreement shall survive the Closing for a period of four (4) months following the Closing. Each covenant and agreement of the Parties contained in this Agreement shall survive the Closing in accordance with its terms. Notwithstanding the foregoing, any breach of representation, warranty, covenant or agreement in respect of which indemnity may be sought under this Agreement shall survive the time at which it would otherwise terminate pursuant to the foregoing, if written notice with respect thereto has been given by Purchaser A to the Seller or the Company, as applicable, prior to such time.
- (b) <u>Indemnification</u>. Subject to the limitations set forth in this <u>Section 9.3</u>, from and after the Closing, the Seller shall indemnify the Purchasers and their Affiliates and their

respective Representatives, successors and assigns against, and shall defend and hold each of them harmless from, any and all liabilities, losses, damages, fines, penalties, deficiencies, taxes, judgments, interest, awards, and other costs and expenses of whatever kind, including reasonable attorneys' and accountants' fees ("Losses"), to the extent arising out of, relating to or resulting from (i) a breach by the Seller of its representations or warranties set forth in this Agreement or in the certificate delivered thereby pursuant to Section 7.2(a)(iii(), or (ii)) any breach by the Seller of any covenant or agreement contained in this Agreement required to be performed before, at or after the Closing.

- (c) <u>Limitation on Liability</u>. Except in the case of fraud, (i) the Seller's maximum aggregate liability for any and all Losses in connection with, relating to or arising out of this Agreement or the transactions contemplated hereby shall in no event exceed the Seller Liability Cap, (ii) the Company's maximum aggregate liability for any Losses in connection with, relating to or arising out of this Agreement or the transactions contemplated hereby shall in no event exceed the Company Liability Cap, and (iii) each Purchaser's maximum aggregate liability for any Losses in connection with, relating to or arising out of this Agreement or the transactions contemplated hereby shall in no event exceed such Purchaser's Pro Rata Share of the Purchaser Liability Cap.
- (d) <u>No Consequential Damages</u>. In no event, whether before of following Closing, shall any of the Parties be responsible for any indirect or punitive damages of the other Parties, including consequential or special damages of any nature whatsoever, the loss of any profits, revenues, opportunities or goodwill, even if the other Party has been advised of the possibility of such damages, except to the extent such damages arise out of fraud. It is clarified, without limiting from anything contained herein or any other rights under applicable Legal Requirements, that payment by a Party to any third party (other than, with respect to the Purchasers only, the Sponsors or any investor therein) pursuant to a judgement of a competent court, shall be considered indirect damages.
- (e) It is clarified that any Restricted Matters taken by Bezeq or any subsidiary thereof (without any breach of any of Seller's or the Company's obligations hereunder), shall not, in and of itself, constitute a breach of the Seller's or the Company's representations and warranties hereunder, without limiting, modifying or otherwise affecting Purchaser's rights under Section 6.2 (including Section 6.2(c)).
- 9.4 <u>Investigations</u>. The representations, warranties, covenants and agreements of the Parties (and the conditions relating thereto) shall not be affected or deemed waived by reason of any investigation made (or not made) by or on behalf of any Party or its Affiliates (or their respective Representatives), or by reason of the fact that a Party or any of its Representatives knew or should have known of any facts or matters, including that any such representation and warranty is or might be inaccurate or untrue, whether prior to the date hereof or the Closing, and including if notwithstanding such knowledge a Party entered into this Agreement or proceeds to Closing. Each of the Parties hereby acknowledges that, regardless of any investigation made (or not made) by or on behalf of any of the other Parties or any of its Affiliates (or their respective Representatives), and regardless of the results of any such investigation, such other Parties have entered into this Agreement (and will effect the Closing) in express reliance upon the representations, warranties covenants and agreements of each Party made herein and the conditions and indemnification obligations relating thereto.
- 9.5 <u>Assignments, Successors</u>. No Party may assign any of its rights under this Agreement without the prior consent of the other Parties; <u>provided</u>, <u>however</u>, that each Purchaser may assign its rights and obligations hereunder (in whole or in part) to any Affiliate of such

Purchaser without the consent of the Seller or the Company, except that such transfer or assignment will not relieve the assigning Purchaser of any of its obligations hereunder, except to the extent such assignment would have a material and adverse effect on the ability of such Purchaser to obtain the Control Permit in a timely manner. Subject to the preceding sentence, this Agreement will apply to, be binding in all respects upon, and inure to the benefit of the successors and permitted assigns of the Parties.

- 9.6 <u>No Third-Party Beneficiaries</u>. This Agreement, specifically excluding express provisions contained in the Commitment Letters, shall not, and is not intended to, confer any rights or remedies upon any Person other than the Parties hereto and their respective successors and permitted assigns.
- 9.7 <u>Notices</u>. All notices, consents, waivers and deliveries under this Agreement must be in writing and will be deemed to have been duly given when (i) delivered by hand (against receipt), (ii) sent by fax or email (with confirmation of receipt) (iii) when received by the addressee, if sent by a nationally recognized overnight delivery service (receipt requested), or (iv) five (5) days after being sent registered or certified mail, return receipt requested, in each case to the appropriate addresses and fax numbers set forth below (or to such other addresses and fax numbers as a Party may hereafter designate by similar notice in accordance with this <u>Section 9.7</u>):

If to the Seller, to: Internet Gold – Golden Lines Ltd.

2 Dov Friedman St., Ramat Gan, Israel Attention: Doron Turgeman, CEO

Email: Doron@igld.com

with a copy (which shall not constitute notice) to:

GKH Law Firm One Azrieli Center Tel Aviv 6701101

Israel

Attention: Dr. Eyal Diskin, Adv. And Yoav Friedman, Adv.

Fax No.: +972-3-6914177

Email: yoavf@gkh-law.com; eyald@gkh-law.com;

If to the Company, to: B Communication Ltd.

2 Dov Friedman St., Ramat Gan, Israel

Attention: Ami Barlev, CEO E-mail: Amib@eurocum.co.il

with a copy (which shall

Gissin & Co Adv.

not constitute notice) to: 38B Habarzel St. Tel Aviv

Guy Gissin and Yael Hershkovitz

Guy@gissinlaw.co.il Yael@gissinlaw.co.il

If to Purchaser A, to: c/o Sea

c/o Searchlight Capital Partners 745 Fifth Avenue, 27th Floor

New York, NY 10151

Attention: Darren Glatt; Nadir Nurmohamed

Email: <u>dglatt@searchlightcap.com</u>; <u>nnurmohamed@searchlightcap.com</u>

with a copy (which shall not constitute notice) to:

Latham & Watkins LLP 885 Third Avenue

New York, New York 10022-4834

Attention: David A. Kurzweil; Eyal N. Orgad

Email: david.kurzweil@lw.com; eyal.orgad@lw.com

with a copy (which shall Meitar Liquornik Geva Leshem Tal

not constitute notice) to: 16 Abba Hillel Silver Rd., Ramat Gan 52506, Israel

Attention: Cliff Felig, Advocate

Telephone No.: +972-3-6103100 Facsimile No.: +972-3-6103111

E-mail: cfelig@meitar.com

with a copy (which shall Shibolet & Co.

not constitute notice) to: 4 Berkowitz St., Tel Aviv 6423806, Israel

Attention: Adi Zaltzman

Telephone No.: +972-3-7778384 E-mail: A.Zaltzman@shibolet.com

If to Purchaser B, to: [An entity wholly-owned by the Fuhrer Family]

[address]
Attention: [-]
Fax No.: [-]
Email: [-]

with a copy (which shall Herzog Fox & Neeman

not constitute notice) to: Asia House, 4 Weizmann St., Tel Aviv 6423904, Israel

Attention: Niv Sivan

Telephone No.: +972-3-6927442

E-mail: sivann@hfn.co.il

Adv. Cliff Felig, of Meitar Liquornik Geva Leshem Tal (16 Abba Hillel Silver Rd. Ramat Gan 52506, Israel) is hereby appointed and authorized, such appointment shall not be terminated until such other Person in the State of Israel is appointed by Purchaser A, by Purchaser A to receive any and all judicial documents from the Seller or the Company pursuant to regulation 478 of the Israeli Civil Procedure Regulations 5744-1984, and any such judicial documents served to Adv. Cliff Felig, of Meitar Liquornik Geva Leshem Tal shall be considered duly served to Purchaser A for all purposes.

- 9.8 <u>Governing Law</u>. This agreement shall be governed by, and construed in accordance with, the laws of the State of Israel without giving effect to its conflict of laws principles.
- 9.9 <u>Jurisdiction</u>. Any Proceeding brought with respect to this Agreement must be brought in any court of competent jurisdiction in Tel Aviv-Jaffa, Israel and, by execution and delivery of this Agreement, each Party (i) accepts, generally and unconditionally, the exclusive jurisdiction of such courts and any related appellate court, and irrevocably agrees to be bound by any judgment rendered thereby in connection with this Agreement and (ii) irrevocably waives any

objection it may now or hereafter have as to the venue of any such suit, action or Proceeding brought in such a court or that such court is an inconvenient forum.

- 9.10 <u>Amendments and Waivers</u>. Any term of this Agreement may be amended only with the written consent of all Parties. The observance of any term hereof may be waived (either prospectively or retroactively and either generally or in a particular instance) only with the written consent of the waiving Party(ies).
- 9.11 <u>Delays or Omissions</u>. No delay or omission to exercise any right, power, or remedy accruing to any of the Parties upon any breach or default by the other Party or Parties under this Agreement shall impair any such right or remedy nor shall it be construed to be a waiver of any such breach or default, or any acquiescence therein in any similar breach or default thereafter occurring.
- 9.12 Entire Agreement. This Agreement (together with the other Transaction Documents) supersedes all prior agreements among the Parties with respect to its subject matter and constitutes a complete and exclusive statement of the terms of the agreement between the Parties with respect to its subject matter, except for the Non-Disclosure Agreement, dated as of October 21, 2018, by and between the Seller and Searchlight Capital Partners L.P. The exhibits and schedules identified in and attached to this Agreement are incorporated herein by reference and shall be deemed as a part hereof as if set forth herein in full.
- 9.13 <u>Severability</u>. If any provision of this Agreement is held by a court of competent jurisdiction to be invalid, illegal or unenforceable under applicable law, then such provision shall be excluded from this Agreement and the validity, legality and enforceability of the remainder of this Agreement and the remaining provisions shall not in any way be affected or impaired thereby (unless the exclusion of such provision materially undermines the purpose and intent of the Parties, in which case this Agreement shall be null and void); <u>provided</u>, <u>however</u>, that in such event, this Agreement shall be interpreted so as to give effect, to the greatest extent consistent with and permitted by applicable Legal Requirements, to the meaning and intention of the excluded provision as determined by such court of competent jurisdiction.
- 9.14 <u>Enforcement</u>. Each Party agrees that the other Parties shall have the right to enforce its rights and the obligations hereunder by an action or actions for specific performance, injunctive or other equitable relief.

9.15 Headings; Construction.

(a) The captions, titles and headings used in this Agreement are for convenience of reference only, shall not be deemed part of this Agreement and shall not affect its construction or interpretation. Except where otherwise expressly provided: (i) all references to "Sections", "Exhibits" or "Schedules" refer to the corresponding Sections, Exhibits or Schedules of or to this Agreement; (ii) all words used in this Agreement will be construed to be of such gender or number as the circumstances require; (iii) the words "include", "includes" and "including" do not limit the preceding words or terms and shall be deemed followed by the phrase "without limitation" whether or not so specified; (iv) the words "hereof," "herein," and "hereunder" and words of similar import, when used in this Agreement, shall refer to this Agreement as a whole and not to any particular provision of this Agreement; (v) terms defined in the singular have a comparable meaning when used in the plural, and *vice versa*; (vi) references to "NIS" and "New Israeli Shekels" are to the currency of the State of Israel; (vii) the term "or" is not exclusive and has the meaning represented by the phrase "and/or" whether or not specified; (viii) the phrase "to the extent" means the degree to which a subject or other theory extends and such phrase shall not mean "if"; (ix) whenever this Agreement refers to a number of days, such number shall refer to calendar days unless Business

Days are specified; (x) any reference in this agreement to "a day" or "days" shall mean calendar days, unless otherwise expressly specified; and (xi) all references herein to the subsidiaries of a Person shall be deemed to include all direct and indirect subsidiaries of such Person unless otherwise indicated or the context otherwise requires. If any action is to be taken or given on or by a particular calendar day, and such calendar day is not a Business Day, then such action may be deferred until the next Business Day.

- (b) The Parties have participated jointly in the drafting of this Agreement and the other Transaction Documents, and each party was represented by counsel in the negotiation and execution of the Transaction Documents. The language used in this Agreement shall be deemed to be the language the Parties have chosen to express their mutual intent. In the event an ambiguity or question of intent or interpretation arises, the Transaction Documents shall be construed as if drafted jointly by the Parties and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any of the provisions of the Transaction Documents, and no rule of strict construction will be applied against any Party.
- 9.16 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original and all of which, taken together, shall constitute one and the same instrument.
- 9.17 Several and not Joint Liability. For the removal of doubt, each Purchaser hereby acknowledges and agrees that any and all of the representations, warranties and covenants of the Seller or the Company are made severally and not jointly, and, without limiting any express representations made by a Party in this Agreement, in no event shall any of the Company or the Seller be responsible or liable in any way for any breach of the representations, warranties and covenants of the other Party.
- Non-Parties. This Agreement may only be enforced against, and any claim or cause of action based upon, arising out of, or related to this Agreement or the transactions contemplated hereby may only be brought against, the Persons that are expressly named as parties hereto and then only with respect to, and to the extent of, the specific obligations set forth herein with respect to such party. Without limiting the generality of the foregoing, and notwithstanding anything that may be expressed or implied in this Agreement or any other Transaction Document or the Commitment Letters, and notwithstanding the fact that a Purchaser may be limited liability company, by entering into this Agreement, each of Parties acknowledges and agrees that: (a) no Person other than a Purchaser, the Company and the Seller shall have any obligations or liabilities under or in connection with this Agreement, and (b) no liability shall attach to, and no recourse shall be had by a Purchaser, the Company or the Seller, or any of their respective Affiliates or any Person purporting to claim by or through any of them or for the benefit of any of them under any theory of liability (including by attempting to pierce a corporate, limited liability company or partnership veil, by attempting to enforce any assessment, or by attempting to enforce any purported right at law or in equity, whether sounding in contract, tort, statute or otherwise) against, any Non-Recourse Party (as defined below) in any way under or in connection with this Agreement, the other Transaction Documents, the Commitment Letters or any other agreement or instrument delivered in connection with this Agreement or the Commitment Letters, or the transactions contemplated hereby or thereby (whether at law or in equity, whether sounding in contract, tort, statute or otherwise), including in the event a Purchaser breaches its obligations under this Agreement (whether willfully, intentionally, unintentionally or otherwise); except that: (i) the Seller or the Company may assert claims solely against the Purchasers, to cause the Purchasers to seek specific performance of each Sponsor's obligations under the Commitment Letters (in accordance with its terms and subject to its conditions), (ii) the Seller or the Company may assert claims solely against the Sponsors, under

and subject to the terms and conditions of the Commitment Letters, and (iii) each of the Seller, the Company and the Purchasers may assert claims solely against the other Party, solely under, in accordance with and subject to the terms and conditions of this Agreement and the other Transaction Documents. As used herein, "Non-Recourse Parties" means, collectively, (1) each Party's direct or indirect former, current or future equity holders, stockholders, members, officers, directors, employees, investment professionals, managers, management companies, general or limited partners, co-investors, controlling persons, advisors, agents, representatives, affiliates, creditors (including their trustees, representatives, advisors and attorneys), assignees or successors, (2) any and all direct or indirect former, current or future equity holders, stockholders, members, officers, directors, employees, investment professionals, managers, management companies, general or limited partners, co-investors, controlling persons, advisors, agents, representatives, affiliates, assignees or successors of any of the foregoing, (3) to the extent not already included in clauses (1) or (2) of this definition of Non-Recourse Parties, Searchlight Capital Partners, L.P., a Delaware limited partnership and any of its affiliates or any direct or indirect former, current or future equity holders, stockholders, members, officers, directors, employees, investment professionals, managers, management companies, general or limited partners, co-investors, controlling persons, advisors, agents, representatives, affiliates, assignees or successors of any of the foregoing, (4) to the extent not already included in clauses (1), (2) or (3) of this definition of Non-Recourse Parties, Local Sponsor and any of its affiliates or any direct or indirect former, current or future equity holders, stockholders, members, officers, directors, employees, investment professionals, managers, management companies, general or limited partners, co-investors, controlling persons, advisors, agents, representatives, affiliates, assignees or successors of any of the foregoing, (5) any and all former, current or future estates, heirs, executors, administrators, trustees, successors or assigns of any of the foregoing, and (6) any financial institution or other Person (other than the Sponsors pursuant to the foregoing clause (i)) which provided, provides or is committed to or will provide financing in connection with the transactions contemplated by this Agreement or the other Transaction Documents.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, the parties have executed this Agreement as of the date first written above.

Interr	Internet Gold – Golden Lines Ltd.		
Ву			
	Name:		
	Title:		
B Co.	B Communications Ltd.		
Ву			
	Name:		
	Title:		

Searchlight II BZQ, L.P.

By: Searchlight II BZQ GP, Ltd., its general partner

By:

Name: Darren Glatt

Title: Authorized Person

T.N.R. Investments Ltd.			
Name:			
Title:			

Exhibit A-1 –Commitment Letter

Exhibit A-2 –Commitment Letter

Exhibit A-3 – Limited Guarantee

Exhibit A-4 – Limited Guarantee

Exhibit B – Company Debt Modifications [See attached.]

c/o Searchlight Capital Partners, L.P. 745 Fifth Avenue, 27th Floor New York, New York 10151

June 24, 2019

Searchlight II BZQ, L.P. c/o Searchlight Capital Partners, L.P. 745 Fifth Avenue, 27th Floor New York, New York 10151

Re: Financing Commitment Letter

Ladies and Gentlemen:

Reference is made to that certain Share Purchase Agreement, dated as of even date herewith (the "Purchase Agreement"), by and among Searchlight II BZQ, L.P., a Cayman Islands exempt limited partnership ("Purchaser A"), T.N.R. Investments Ltd., an Israeli company, Internet Gold - Golden Lines Ltd. (the "Seller") and B Communications Ltd. (the "Company"). Capitalized terms used but not otherwise defined herein shall have the meanings ascribed to them in the Purchase Agreement.

This letter agreement is being delivered to Purchaser A in connection with the transactions contemplated by the Purchase Agreement (the "<u>Transactions</u>").

Commitment. This letter agreement confirms the commitment of each entity listed on Exhibit A attached hereto (each, an "Investor" and collectively, the "Investors"), severally and not jointly, subject to the terms and conditions set forth herein, to purchase, or to cause the purchase of, directly or indirectly, through one or more intermediate entities, its pro rata percentage (set forth on Exhibit A, and such amount of the Aggregate Commitment (as defined below) with respect to each Investor, such Investor's "Maximum Investor Commitment") of equity and/or debt securities of Purchaser A, such that all such securities purchased by all such Investors shall equal an aggregate amount of Purchaser A's Pro Rata Share of NIS 485,000,000 (the "Aggregate Commitment"), solely to the extent necessary for Purchaser A to fund its payment obligations under Section 7.2(c) and to pay all fees and expenses of Purchaser A related to the Transactions; provided that, notwithstanding anything herein or otherwise to the contrary, no Investor shall, under any circumstances, be obligated to contribute to, purchase securities of, or otherwise provide any funds to, Purchaser A in an amount exceeding the amount of such Investor's Maximum Investor Commitment and the Investors, collectively, shall not, under any circumstances, be obligated to contribute to, purchase securities of, or otherwise provide any funds to, Purchaser A in an amount exceeding the Aggregate Commitment. This letter agreement constitutes all of the obligations and liabilities of each Investor in relation to such Investor's portion of the Aggregate Commitment and shall not give rise to any other obligations or liabilities on any of the Investors. The proceeds from such purchase of debt and/or equity securities shall be used by Purchaser A solely to pay its obligations under Section 7.2(c) and fees and expenses of Purchaser A related to the Transactions and for no other purpose.

- 2. Funding; Termination. Each Investor's obligation to fund its Maximum Investor Commitment is subject to the terms of this letter agreement and to (a) the execution and delivery of the Purchase Agreement by all parties thereto as of the date hereof, (b) the satisfaction or written waiver by Purchaser A of each of the conditions set forth in Section 6.1 and Section 6.2 of the Purchase Agreement (other than those conditions which by their nature are to be satisfied at the Closing, but subject to the satisfaction thereof at the Closing), (c) the substantially simultaneous closing of the investment contemplated by the Financing Commitment Letter (substantially on the same terms as this letter agreement (other than NIS amounts stated therein)) delivered by T.N.R. Real Estate Properties Ltd. (the "Co-Sponsor Financing Commitment Letter"), and (d) the substantially simultaneous consummation of the Closing in accordance with the terms of the Purchase Agreement. For the avoidance of doubt, and notwithstanding anything herein or otherwise to the contrary, no Investor shall be obligated to fund any portion of its Maximum Investor Commitment unless the Closing occurs substantially simultaneously with such funding. This letter agreement and the obligation of each Investor to fund its Maximum Investor Commitment will terminate automatically and immediately upon the earliest to occur of (i) the occurrence of Closing and payment of Purchaser A's payments obligations under Section 7.2(c) (at which time all such obligations shall be immediately discharged), (ii) the termination of the Purchase Agreement pursuant to Section 8.1 thereof, (iii) the termination of the Co-Sponsor Financing Commitment Letter, or (iv) Seller, the Company, Purchaser A or any Person claiming by, through or for the benefit of Seller, the Company, or Purchaser A, asserting a claim in any suit, action, litigation, claim, charge, complaint, grievance, arbitration proceeding, at law or in equity, or by, in or before any court, tribunal, commission, agency or other governmental authority, or in any other proceeding against (i) any Investor or any other Recourse Party (as defined below) under or in connection with the Purchase Agreement or any of the transactions contemplated thereby other than Seller or the Company asserting any Retained Claim (as defined below) against the Recourse Parties against which such Retained Claim may be asserted in accordance with Section 3 hereof, or (ii) any Non-Recourse Party (as defined below). Section 3 hereof shall survive any such termination.
- 3. No Recourse. Notwithstanding anything that may be expressed or implied in this letter agreement or any document or instrument delivered in connection herewith or otherwise, and notwithstanding the fact that the Investors may be limited partnerships or limited liability companies, by its acceptance of the direct or indirect benefits of this letter agreement, each of Purchaser A, the Company and Seller acknowledges and agrees that: (a) no Person other than the Investors and Purchaser A shall have any obligations or liabilities under or in connection with this letter agreement, (b) none of the Investors shall have any obligations or liabilities under or in connection with this letter agreement or the Transactions except as expressly provided by this letter agreement, and (c) no liability shall attach to, and no recourse shall be had by Purchaser A, the Company, Seller or any of their respective affiliates or any Person purporting to claim by or through any of them or for the benefit of any of them under any

theory of liability (including, without limitation, by attempting to pierce a corporate, limited liability company or partnership veil, by attempting to compel Purchaser A to enforce any rights that it may have against any Person, by attempting to enforce any assessment, or by attempting to enforce any purported right at law or in equity, whether sounding in contract, tort, statute or otherwise) against, any Recourse Party or any Non-Recourse Party in any way under or in connection with this letter agreement, the Purchase Agreement or any other agreement or instrument delivered in connection with this letter agreement or the Purchase Agreement, or the transactions contemplated hereby or thereby (whether at law or in equity, whether sounding in contract, tort, statute or otherwise), including, without limitation, in the event Purchaser A breaches its obligations under the Purchase Agreement (whether willfully, intentionally, unintentionally or otherwise) and including whether or not Purchaser A's breach is caused by the breach by any Investor of its obligations under this letter agreement; except that: (i) Seller may assert claims solely against Searchlight Capital Partners, L.P., a Delaware limited partnership ("SCP LP") solely under the Non-Disclosure and Confidentiality Agreement, dated as of October 21, 2018, (ii) Seller or the Company may assert claims solely against Purchaser A, to cause Purchaser A to seek specific performance of each Investor's obligations under this letter agreement to fund its Maximum Investor Commitment in accordance with the terms and subject to the conditions of this letter agreement, (iii) Seller or the Company may assert claims solely against the Investors, under and subject to the terms and conditions of the last sentence of this Section 3 of this letter agreement, and (iv) Seller or the Company may assert claims solely against Purchaser A, in accordance with and subject to the terms and conditions of the Purchase Agreement (the claims described in clauses (i) through (iv) collectively, the "Retained Claims"). As used in this letter agreement, (x) the term "Recourse Parties" means, collectively, Purchaser A, the Investors and the other Persons against which Retained Claims may expressly be asserted pursuant to clauses (i) through (iv) of the definition of Retained Claims set forth above (and solely to the extent of such Retained Claims), and (y) the term "Non-Recourse Parties" means, collectively, (1) the Recourse Parties' respective direct or indirect former, current or future equity holders, stockholders, members, officers, directors, employees, investment professionals, managers, management companies, general or limited partners, co-investors, controlling persons, advisors, agents, representatives, affiliates, assignees or successors, (2) any and all direct or indirect former, current or future equity holders, stockholders, members, officers, directors, employees, investment professionals, managers, management companies, general or limited partners, co-investors, controlling persons, advisors, agents, representatives, affiliates, assignees or successors of any of the foregoing, (3) to the extent not already included in clauses (1) or (2) of this definition of Non-Recourse Parties, SCP LP and any of its affiliates or any direct or indirect former, current or future equity holders, stockholders, members, officers, directors, employees, investment professionals, managers, management companies, general or limited partners, co-investors, controlling persons, advisors, agents, representatives, affiliates, assignees or successors of any of the foregoing, (4) any and all former, current or future estates, heirs, executors, administrators, trustees, successors or assigns of any of the foregoing, and (5) any financial institution or other Person which provided, provides or is committed to or will provide financing in connection with the transactions contemplated by the Purchase Agreement; provided that none of the Recourse Parties (solely with respect to the Retained Claims) shall be Non-Recourse Parties. This letter agreement may only be enforced by (I) Purchaser A at the direction of the Investors, or (II) Seller or the Company, pursuant to Seller's or the Company's right to seek specific performance of Purchaser A's obligation to enforce the Investors' obligation to fund the Aggregate Commitment in accordance with the terms of this letter agreement, pursuant to, and subject to, and solely in accordance with, Section 9.14 of the Purchase Agreement and the terms and conditions set forth therein.

- 4. <u>Assignment</u>. This letter agreement and each Investor's commitment hereunder shall not be assignable, directly or indirectly, to any Person without the prior written consent of the Investors, and any attempted assignment without such consent shall be null and void and of no force and effect. Each Investor may assign its commitments hereunder to its affiliates or co-investors or funds or investment vehicles affiliated with such Investor; <u>provided</u>, <u>however</u>, that notwithstanding any such assignment, each Investor shall remain liable to perform all of its obligations hereunder in accordance with and subject to the terms of this letter agreement.
- 5. <u>Binding Effect.</u> (a) This letter agreement shall be binding on the Investors solely for the benefit of Purchaser A, and nothing in this letter agreement, express or implied, shall be construed to confer upon or give any Person other than Purchaser A any benefits, rights or remedies of any nature whatsoever under or by reason of, or any rights to enforce or cause Purchaser A to enforce, the Aggregate Commitment or any Maximum Investor Commitment or any provision of this letter agreement, (b) nothing set forth in this letter agreement shall be construed to confer upon or give any Person other than Purchaser A any benefits, rights or remedies under or by reason of, or any rights to enforce or cause Purchaser A to enforce, the Aggregate Commitment or any Maximum Investor Commitment or any provisions of this letter agreement; <u>provided</u>, <u>however</u>, that, notwithstanding the foregoing in this <u>Section 5</u>, Seller and the Company have relied on this letter agreement and, accordingly, Seller and the Company are express third-party beneficiaries hereof and shall have the enforcement rights described in the last sentience of <u>Section 3</u> hereof and no others. Purchaser A's creditors shall have no right to enforce this letter agreement or to cause Purchaser A to enforce this letter agreement.
- Representations and Warranties. Each Investor hereby represents and warrants, on a several (not joint and several) basis as to itself only, to Purchaser A that, except as would not reasonably be expected to impair or delay such Investor's performance of its Maximum Investor Commitment obligations hereunder in any material respect, (a) it has all power and authority to execute, deliver and perform this letter agreement, (b) the execution, delivery and performance of this letter agreement by it has been duly and validly authorized and approved by all necessary action by it, (c) this letter agreement has been duly and validly executed and delivered by it and constitutes a valid and legally binding obligation of it, enforceable against it in accordance with the terms of this letter agreement, subject to bankruptcy, insolvency, reorganization and other laws of general applicability relating to or affecting creditors' rights and to general equity principles, and (d) the execution, delivery and performance by it of this letter agreement (i) do not violate its organizational documents, (ii) do not violate any law or judgment applicable to it, or (iii) would not result in any violation of, or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, cancellation or acceleration of any obligation or to the loss of any benefit under, any contract to which it is a party.

7. Miscellaneous.

- a. <u>Counterparts</u>. This letter agreement may be executed in multiple counterparts (including by facsimile or by PDF delivered via email), each such counterpart when executed being deemed to be an original instrument, and all such counterparts shall together constitute one and the same agreement. This letter agreement will become effective upon its acceptance by Purchaser A, as evidenced by the delivery to each of the Investors of a counterpart of this letter agreement executed by Purchaser A.
- b. <u>Amendments; Waivers</u>. No amendment or waiver of any provision of this letter agreement will be valid and binding unless it is in writing and signed by Purchaser A and each of the Investors.
- c. <u>Entire Agreement</u>. This letter agreement constitutes the sole agreement, and supersedes all prior agreements, understandings and statements, written or oral, between the Investors or any of their respective affiliates (other than Purchaser A), on the one hand, and Purchaser A or any of its affiliates (other than the Investors), on the other, with respect to the subject matter hereof.
- Severability. If any term, condition or other provision of this letter agreement is determined by a court of competent jurisdiction to be invalid, illegal or incapable of being enforced by any rule of law or public policy, all other terms, provisions and conditions of this letter agreement shall nevertheless remain in full force and effect. Upon such determination that any term or other provision hereof is invalid, illegal or incapable of being enforced, (i) a suitable and equitable term or provision determined by a court of competent jurisdiction shall be substituted therefor in order to carry out, so far as may be valid, legal and enforceable under law, the intent and purpose of such invalid, illegal or unenforceable term or provision, and (ii) the remainder of this letter agreement and the application of such terms and other provision to other Persons or circumstances shall not be affected by such invalidity, illegality or unenforceability, nor shall such invalidity, illegality or unenforceability affect the validity, legality or enforceability of such term or provision, or the application of such term or provision, in any other jurisdiction. Notwithstanding anything in this letter agreement or otherwise to the contrary, (I) this letter agreement may not be enforced against any Investor without giving effect to the limitations represented by the Maximum Investor Commitment and the Aggregate Commitment, and (II) for the avoidance of doubt, the obligations under this letter agreement are several (and not joint and several) from the obligations of the parties to the Co-Sponsor Financing Commitment Letter thereunder, and the Investors shall not be liable or otherwise responsible for, directly or indirectly, for any breach of the Co-Sponsor Financing Commitment Letter by the parties thereto.
- e. <u>Governing Law; Submission to Jurisdiction.</u> This letter agreement shall be governed by, and construed in accordance with, the laws of the State of Israel without giving effect to its conflict of laws principles. Any Proceeding brought with respect to this letter agreement must be brought in any court of competent jurisdiction in Tel Aviv-Jaffa, Israel and, by execution and

delivery of this letter agreement, each party (i) accepts, generally and unconditionally, the exclusive jurisdiction of such courts and any related appellate court, and irrevocably agrees to be bound by any judgment rendered thereby in connection with this letter agreement and (ii) irrevocably waives any objection it may now or hereafter have as to the venue of any such suit, action or Proceeding brought in such a court or that such court is an inconvenient forum.

[Remainder of page intentionally left blank.]

Very truly yours,

Searchlight Capital II, L.P.
By: Searchlight Capital Partners II GP, L.P., its general partner
By: Searchlight Capital Partners II GP, LLC, its general partner
By:
Name: Darren Glatt
Title: Authorized Person

Searchlight Capital II PV, L.P.

By: Searchlight Capital Partners II GP, L.P., its general partner By: Searchlight Capital Partners II GP, LLC, its general partner

By: ________Name: Darren Glatt
Title: Authorized Person

Accepted and Ag	Accepted and Agreed:		
Searchlight II E	BZQ, L.P.,		
By: Searchlight	II BZQ GP, Ltd., its general	partner	
Ву:			
Name: I	Darren Glatt		
Title: A	uthorized Person		
Acknowledgeme	ent of express intended third	party beneficiaries:	
INTERNET GO	OLD - GOLDEN LINES L	ΓD.	
Ву:			
Name: Fitle:			
B COMMUNIC	CATIONS LTD.		
Ву:		-	
Name: Fitle:			

Exhibit A

Investor	Percentage of Aggregate Commitment
Searchlight Capital II, L.P.	49.737%
Searchlight Capital II PV, L.P.	50.263%

c/o T.N.R. Real Estate Properties Ltd. HaShiloah Street 6, Petah Tiqva 4951439, Israel

June 24, 2019

T.N.R. Investments LTD. 48 Bustanai St., Ramat Hasharon 4722431, Israel

Re: Financing Commitment Letter

Ladies and Gentlemen:

Reference is made to that certain Share Purchase Agreement, dated as of even date herewith (the "<u>Purchase Agreement</u>"), by and among Searchlight II BZQ, L.P., a Cayman Islands exempt limited partnership, T.N.R. Investments LTD., a limited liability company organized and existing under the laws of the State of Israel ("<u>Purchaser B</u>"), and Internet Gold - Golden Lines Ltd. (the "<u>Seller</u>") and B Communications Ltd. (the "<u>Company</u>"). Capitalized terms used but not otherwise defined herein shall have the meanings ascribed to them in the Purchase Agreement.

This letter agreement is being delivered to Purchaser B in connection with the transactions contemplated by the Purchase Agreement (the "<u>Transactions</u>").

Commitment. This letter agreement confirms the commitment of the entity listed on Exhibit A attached hereto, an entity fully owned by the Fuhrer family (the "Investor"), subject to the terms and conditions set forth herein, to purchase, or to cause the purchase of, directly or indirectly, through one or more intermediate entities, its Aggregate Commitment (as defined below) with respect to the Investor's "Maximum Investor Commitment") of equity and/or debt securities of Purchaser B, such that all such securities purchased by the Investor shall equal an aggregate amount of Purchaser B's Pro Rata Share of NIS 485,000,000 (the "Aggregate Commitment"), solely to the extent necessary for Purchaser B to fund its payment obligations under Section 7.2(c) and to pay all fees and expenses of Purchaser B related to the Transactions; <u>provided</u> that, notwithstanding anything herein or otherwise to the contrary, the Investor shall not, under any circumstances, be obligated to contribute to, purchase securities of, or otherwise provide any funds to, Purchaser B in an amount exceeding the amount of the Investor's Maximum Investor Commitment and the Investor shall not, under any circumstances, be obligated to contribute to, purchase securities of, or otherwise provide any funds to, Purchaser B in an amount exceeding the Aggregate Commitment. This letter agreement constitutes all of the obligations and liabilities of the Investor in relation to the Investor's Aggregate Commitment and shall not give rise to any other obligations or liabilities on the Investor. The proceeds from such purchase of debt and/or

equity securities shall be used by Purchaser B solely to pay its obligations under Section 7.2(c) and fees and expenses of Purchaser B related to the Transactions and for no other purpose.

- Funding; Termination. The Investor's obligation to fund its Maximum Investor Commitment is subject to the terms of this letter agreement and to (a) the execution and delivery of the Purchase Agreement by all parties thereto as of the date hereof, (b) the satisfaction or written waiver by Purchaser B of each of the conditions set forth in Section 6.1 and Section 6.2 of the Purchase Agreement (other than those conditions which by their nature are to be satisfied at the Closing, but subject to the satisfaction thereof at the Closing), (c) the substantially simultaneous closing of the investment contemplated by the Financing Commitment Letter (substantially on the same terms as this letter agreement (other than NIS amounts stated therein)) delivered by Searchlight Capital Partners, L.P. (the "Co-Sponsor Financing Commitment Letter"), and (d) the substantially simultaneous consummation of the Closing in accordance with the terms of the Purchase Agreement. For the avoidance of doubt, and notwithstanding anything herein or otherwise to the contrary, the Investor shall not be obligated to fund any portion of its Maximum Investor Commitment unless the Closing occurs substantially simultaneously with such funding. This letter agreement and the obligation of the Investor to fund its Maximum Investor Commitment will terminate automatically and immediately upon the earliest to occur of (i) the occurrence of Closing and payment of Purchaser B's payments obligations under Section 7.2(c) (at which time all such obligations shall be immediately discharged), (ii) the termination of the Purchase Agreement pursuant to Section 8.1 thereof, (iii) the termination of the Co-Sponsor Financing Commitment Letter, or (iv) Seller, the Company, Purchaser B or any Person claiming by, through or for the benefit of Seller, the Company, or Purchaser B, asserting a claim in any suit, action, litigation, claim, charge, complaint, grievance, arbitration proceeding, at law or in equity, or by, in or before any court, tribunal, commission, agency or other governmental authority, or in any other proceeding against (i) the Investor or any other Recourse Party (as defined below) under or in connection with the Purchase Agreement or any of the transactions contemplated thereby other than Seller or the Company asserting any Retained Claim (as defined below) against the Recourse Parties against which such Retained Claim may be asserted in accordance with Section 3 hereof, or (ii) any Non-Recourse Party (as defined below). Section 3 hereof shall survive any such termination.
- No Recourse. Notwithstanding anything that may be expressed or implied 3. in this letter agreement or any document or instrument delivered in connection herewith or otherwise, and notwithstanding the fact that the Investor may be a limited partnership or a limited liability company, by its acceptance of the direct or indirect benefits of this letter agreement, each of Purchaser B, the Company and Seller acknowledges and agrees that: (a) no Person other than the Investor and Purchaser B shall have any obligations or liabilities under or in connection with this letter agreement, (b) the Investor shall not have any obligations or liabilities under or in connection with this letter agreement or the Transactions except as expressly provided by this letter agreement, and (c) no liability shall attach to, and no recourse shall be had by Purchaser B, the Company, Seller or any of their respective affiliates or any Person purporting to claim by or through any of them or for the benefit of any of them under any theory of liability (including, without limitation, by attempting to pierce a corporate, limited liability company or partnership veil, by attempting to compel Purchaser B to enforce any rights that it may have against any Person, by attempting to enforce any assessment, or by attempting to enforce any purported right at law or in equity, whether sounding in contract, tort, statute or otherwise) against, any Recourse Party or

any Non-Recourse Party in any way under or in connection with this letter agreement, the Purchase Agreement or any other agreement or instrument delivered in connection with this letter agreement or the Purchase Agreement, or the transactions contemplated hereby or thereby (whether at law or in equity, whether sounding in contract, tort, statute or otherwise), including, without limitation, in the event Purchaser B breaches its obligations under the Purchase Agreement (whether willfully, intentionally, unintentionally or otherwise) and including whether or not Purchaser B's breach is caused by the breach by the Investor of its obligations under this letter agreement; except that: (i) Seller may assert claims solely against Searchlight Capital Partners, L.P., a Delaware limited partnership ("SCP LP") solely under the Non-Disclosure and Confidentiality Agreement, dated as of October 21, 2018, (ii) Seller or the Company may assert claims solely against Purchaser B, to cause Purchaser B to seek specific performance of the Investor's obligations under this letter agreement to fund its Maximum Investor Commitment in accordance with the terms and subject to the conditions of this letter agreement, (iii) Seller or the Company may assert claims solely against the Investor, under and subject to the terms and conditions of the last sentence of this Section 3 of this letter agreement, and (iv) Seller or the Company may assert claims solely against Purchaser B, in accordance with and subject to the terms and conditions of the Purchase Agreement (the claims described in clauses (i) through (iv) collectively, the "Retained Claims"). As used in this letter agreement, (x) the term "Recourse Parties" means, collectively, Purchaser B, the Investor and the other Persons against which Retained Claims may expressly be asserted pursuant to clauses (i) through (iv) of the definition of Retained Claims set forth above (and solely to the extent of such Retained Claims), and (y) the term "Non-Recourse Parties" means, collectively, (1) the Recourse Parties' respective direct or indirect former, current or future equity holders, stockholders, members, officers, directors, employees, investment professionals, managers, management companies, general or limited partners, co-investors, controlling persons, advisors, agents, representatives, affiliates, assignees or successors, (2) any and all direct or indirect former, current or future equity holders, stockholders, members, officers, directors, employees, investment professionals, managers, management companies, general or limited partners, co-investors, controlling persons, advisors, agents, representatives, affiliates, assignees or successors of any of the foregoing, (3) to the extent not already included in clauses (1) or (2) of this definition of Non-Recourse Parties, SCP LP and any of its affiliates or any direct or indirect former, current or future equity holders, stockholders, members, officers, directors, employees, investment professionals, managers, management companies, general or limited partners, co-investors, controlling persons, advisors, agents, representatives, affiliates, assignees or successors of any of the foregoing, (4) any and all former, current or future estates, heirs, executors, administrators, trustees, successors or assigns of any of the foregoing, and (5) any financial institution or other Person which provided, provides or is committed to or will provide financing in connection with the transactions contemplated by the Purchase Agreement; provided that none of the Recourse Parties (solely with respect to the Retained Claims) shall be Non-Recourse Parties. This letter agreement may only be enforced by (I) Purchaser B at the direction of the Investor, or (II) Seller or the Company, pursuant to Seller's or the Company's right to seek specific performance of Purchaser B's obligation to enforce the Investor's obligation to fund the Aggregate Commitment in accordance with the terms of this letter agreement, pursuant to, and subject to, and solely in accordance with, Section 9.14 of the Purchase Agreement and the terms and conditions set forth therein.

4. <u>Assignment</u>. This letter agreement and the Investor's commitment hereunder shall not be assignable, directly or indirectly, to any Person without the prior written consent of the Investor, and any attempted assignment without such consent shall be null and void

and of no force and effect. The Investor may assign its commitments hereunder to its affiliates or co-investors or funds or investment vehicles affiliated with the Investor; <u>provided</u>, <u>however</u>, that notwithstanding any such assignment, the Investor shall remain liable to perform all of its obligations hereunder in accordance with and subject to the terms of this letter agreement.

- 5. <u>Binding Effect.</u> (a) This letter agreement shall be binding on the Investor solely for the benefit of Purchaser B, and nothing in this letter agreement, express or implied, shall be construed to confer upon or give any Person other than Purchaser B any benefits, rights or remedies of any nature whatsoever under or by reason of, or any rights to enforce or cause Purchaser B to enforce, the Aggregate Commitment or any Maximum Investor Commitment or any provision of this letter agreement, (b) nothing set forth in this letter agreement shall be construed to confer upon or give any Person other than Purchaser B any benefits, rights or remedies under or by reason of, or any rights to enforce or cause Purchaser B to enforce, the Aggregate Commitment or any Maximum Investor Commitment or any provisions of this letter agreement; provided, however, that, notwithstanding the foregoing in this Section 5, Seller and the Company have relied on this letter agreement and, accordingly, Seller and the Company are express third-party beneficiaries hereof and shall have the enforcement rights described in the last sentience of Section 3 hereof and no others. Purchaser B's creditors shall have no right to enforce this letter agreement or to cause Purchaser B to enforce this letter agreement.
- Representations and Warranties. The Investor hereby represents and warrants, on a several (not joint and several) basis as to itself only, to Purchaser B that, except as would not reasonably be expected to impair or delay the Investor's performance of its Maximum Investor Commitment obligations hereunder in any material respect, (a) it has all power and authority to execute, deliver and perform this letter agreement, (b) the execution, delivery and performance of this letter agreement by it has been duly and validly authorized and approved by all necessary action by it, (c) this letter agreement has been duly and validly executed and delivered by it and constitutes a valid and legally binding obligation of it, enforceable against it in accordance with the terms of this letter agreement, subject to bankruptcy, insolvency, reorganization and other laws of general applicability relating to or affecting creditors' rights and to general equity principles, and (d) the execution, delivery and performance by it of this letter agreement (i) do not violate its organizational documents, (ii) do not violate any law or judgment applicable to it, or (iii) would not result in any violation of, or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination, cancellation or acceleration of any obligation or to the loss of any benefit under, any contract to which it is a party.

7. <u>Miscellaneous</u>.

a. <u>Counterparts</u>. This letter agreement may be executed in multiple counterparts (including by facsimile or by PDF delivered via email), each such counterpart when executed being deemed to be an original instrument, and all such counterparts shall together constitute one and the same agreement. This letter agreement will become effective upon its acceptance by Purchaser B, as evidenced by the delivery to each of the Investor of a counterpart of this letter agreement executed by Purchaser B.

- b. <u>Amendments; Waivers</u>. No amendment or waiver of any provision of this letter agreement will be valid and binding unless it is in writing and signed by Purchaser B and the Investor.
- c. <u>Entire Agreement</u>. This letter agreement constitutes the sole agreement, and supersedes all prior agreements, understandings and statements, written or oral, between the Investor or any of their respective affiliates (other than Purchaser B), on the one hand, and Purchaser B or any of its affiliates (other than the Investor), on the other, with respect to the subject matter hereof.
- Severability. If any term, condition or other provision of this letter agreement is determined by a court of competent jurisdiction to be invalid, illegal or incapable of being enforced by any rule of law or public policy, all other terms, provisions and conditions of this letter agreement shall nevertheless remain in full force and effect. Upon such determination that any term or other provision hereof is invalid, illegal or incapable of being enforced, (i) a suitable and equitable term or provision determined by a court of competent jurisdiction shall be substituted therefor in order to carry out, so far as may be valid, legal and enforceable under law, the intent and purpose of such invalid, illegal or unenforceable term or provision, and (ii) the remainder of this letter agreement and the application of such terms and other provision to other Persons or circumstances shall not be affected by such invalidity, illegality or unenforceability, nor shall such invalidity, illegality or unenforceability affect the validity, legality or enforceability of such term or provision, or the application of such term or provision, in any other jurisdiction. Notwithstanding anything in this letter agreement or otherwise to the contrary, (I) this letter agreement may not be enforced against the Investor without giving effect to the limitations represented by the Maximum Investor Commitment and the Aggregate Commitment, and (II) for the avoidance of doubt, the obligations under this letter agreement are several (and not joint and several) from the obligations of the parties to the Co-Sponsor Financing Commitment Letter thereunder, and the Investor shall not be liable or otherwise responsible for, directly or indirectly, for any breach of the Co-Sponsor Financing Commitment Letter by the parties thereto.
- e. Governing Law; Submission to Jurisdiction. This letter agreement shall be governed by, and construed in accordance with, the laws of the State of Israel without giving effect to its conflict of laws principles. Any Proceeding brought with respect to this letter agreement must be brought in any court of competent jurisdiction in Tel Aviv-Jaffa, Israel and, by execution and delivery of this letter agreement, each party (i) accepts, generally and unconditionally, the exclusive jurisdiction of such courts and any related appellate court, and irrevocably agrees to be bound by any judgment rendered thereby in connection with this letter agreement and (ii) irrevocably waives any objection it may now or hereafter have as to the venue of any such suit, action or Proceeding brought in such a court or that such court is an inconvenient forum.

[Remainder of page intentionally left blank.]

Very truly yours,
T.N.R. Real Estate Properties Ltd.
By:
Name: David Fuhrer Title: Chairman

Accepted and Agreed:	
T.N.R. Investments LTD. By:	
Name: David Fuhrer Title: Chairman	
Acknowledgement of express intended third party ber INTERNET GOLD - GOLDEN LINES LTD.	neficiaries:
By:	
Name: Title:	
B COMMUNICATIONS LTD.	
By:	
Name:	

Exhibit A

Investor	Maximum Investor Commitment
T.N.R. Real Estate Properties Ltd.	\$

LIMITED GUARANTEE

Limited Guarantee, dated as of June 24, 2019 (this "Limited Guarantee"), by each of the parties listed on Exhibit A hereto (each, a "Guarantor" and collectively, the "Guarantors"), in favor of Internet Gold - Golden Lines Ltd. and B Communications Ltd. (collectively, the "Guaranteed Party"). Reference is made to that certain Share Purchase Agreement, dated as of even date herewith (the "Purchase Agreement"), by and among Searchlight II BZQ, L.P., a Cayman Islands exempt limited partnership ("Purchaser A"), T.N.R. Investments Ltd., and the Guaranteed Party. Except as otherwise specified herein, capitalized terms used herein but not otherwise defined have the respective meanings ascribed to them in the Purchase Agreement.

<u>Limited Guarantee</u>. To induce the Guaranteed Party to enter into the Purchase 1. Agreement, each Guarantor hereby guarantees, severally and not jointly, and not jointly and severally, to the Guaranteed Party, on the terms and subject to the conditions set forth herein, payment of a portion (in accordance with the amount set forth opposite such Guarantor's name on Exhibit A hereto) of (i) the payment obligations of Purchaser A under the Purchase Agreement, solely to the extent exist and payable following the valid termination of the Purchase Agreement, in each case, on the terms and subject to the conditions set forth in the Purchase Agreement (including Section 8.2(b) thereof) and herein (such amounts collectively, the "Guaranteed Obligations"), in an amount equal to the percentage of the Maximum Aggregate Amount (as defined below) set forth opposite such Guarantor's name on Exhibit A hereto (such amount with respect to each Guarantor is such Guarantor's "Maximum Guarantor Amount"); provided that (a) the maximum liability of each Guarantor hereunder shall not exceed such Guarantor's Maximum Guarantor Amount, (b) the maximum aggregate liability of all Guarantors hereunder shall not exceed twenty six million and hundred thousand New Israeli Shekels (NIS 26,100,000) (the "Maximum Aggregate Amount"). It being understood and agreed that this Limited Guarantee may not be enforced without giving full and absolute effect to the Maximum Aggregate Amount and each Maximum Guarantor Amount. The Guaranteed Party hereby agrees that the Guarantors shall in no event be required to pay to any Person or Persons in the aggregate more than the Maximum Aggregate Amount (and that each Guarantor shall in no event be required to pay to any Person or Persons in the aggregate more than such Guarantor's Maximum Guarantor Amount) under, in respect of, or in connection with this Limited Guarantee or the Purchase Agreement, and no Guarantor shall have any obligation or liability to any Person under, in respect of or in connection with this Limited Guarantee or the Purchase Agreement other than under this Guarantee as specifically and expressly set forth herein. The guarantee by the Guaranters of the Guaranteed Obligations under this Limited Guarantee may be enforced for the payment of money only. Notwithstanding anything to the contrary contained in this Limited Guarantee, in the Purchase Agreement or otherwise, the Guaranteed Party hereby agrees that to the extent Purchaser is relieved of all or any portion of its payment obligations under the Purchase Agreement by satisfaction thereof or pursuant to any other agreement with the Guaranteed Party, each Guarantor shall be similarly relieved, to such extent, of its respective obligations under this Limited Guarantee.

2. Terms of Limited Guarantee; Recovery Claim.

(a) This Limited Guarantee is one of payment, not collection, and a separate action or actions may be brought and prosecuted against the Guarantors to enforce this Limited

Guarantee (solely in accordance with and subject to its terms, including the limitations contained herein), irrespective of whether any action is brought against Purchaser A or whether Purchaser A is joined in any such action or actions. Each Guarantor reserves the right, notwithstanding anything to the contrary provided herein or otherwise, to (i) set off any amount owed hereunder by such Guarantor against any payment owing by the Guaranteed Party to Purchaser A or any of the Guarantors, and (ii) assert any and all defenses which any Guarantor Purchaser A or any of their affiliates may have against payment of the Guaranteed Obligations.

- (b) The liability of the Guarantors under this Limited Guarantee shall, to the fullest extent permitted under applicable Legal Requirements, be absolute and unconditional, irrespective of:
 - (i) any change in the corporate existence, structure or ownership of Purchaser A or any insolvency, bankruptcy, reorganization, liquidation or other similar proceeding of Purchaser A;
 - (ii) any change in the manner, place or terms of payment or performance, or any change or extension of the time of payment or performance of, renewal or alteration of, the Guaranteed Obligations, or any liability incurred directly or indirectly in respect thereof; or
 - (iii) the existence of any claim, set-off or other right that the Guarantors may have at any time against Purchaser A, whether in connection with any of the Guaranteed Obligations or otherwise.

Notwithstanding the foregoing or anything to the contrary in this Limited Guarantee, each of the Guaranters shall be fully released and discharged hereunder if the Guaranteed Obligations are paid in full by Purchaser A or any other Person.

3. <u>Sole Remedy</u>.

- (a) The Guaranteed Party acknowledges and agrees that:
- (i) the sole cash assets of Purchaser A are cash in a *de minimis* amount (if any), and that no additional funds are expected to be contributed to Purchaser A unless and until the Closing occurs;
- (ii) that the Guaranteed Party is bound by and shall comply with the applicable terms and conditions of Sections 2, 3, 6 and 6 of the Equity Commitment Letter, dated as of the date hereof, by and between Purchaser A and the Guarantors (the "Equity Commitment Letter");
- (iii) the Guarantors shall not have any obligation or liability to any Person relating to, arising out of or in connection with the Purchase Agreement, this Limited Guarantee, the Equity Commitment Letter or the transactions contemplated thereby or hereby, other than as expressly set forth herein or in the Equity Commitment Letter; and

- (iv) notwithstanding anything to the contrary in this Limited Guarantee, the Equity Commitment Letter, the Purchase Agreement or otherwise, it has no and shall have no right of recovery against the Guarantors or any of any of the Non-Recourse Parties (as defined in the Equity Commitment Letter), through any Guarantor, Purchaser A or otherwise, whether by or through attempted piercing of the corporate, limited liability company or limited partnership veil, by or through a claim by or on behalf of Purchaser A against the Guarantors or any of the Non-Recourse Parties, or otherwise, except for its rights against the Guarantors under this Limited Guarantee pursuant to the terms and subject to the conditions hereof and except for the Retained Claims (as defined in the Equity Commitment Letter).
- (b) Recourse against the Guarantors under this Limited Guarantee shall be the sole and exclusive remedy (whether at law, in equity, in contract, in tort or otherwise) of the Guaranteed Party and its Non-Recourse Parties against the Guarantors and their respective Non-Recourse Parties in respect of any breaches, Losses or other damages relating to, arising out of or in connection with, the Purchase Agreement or the transactions contemplated thereby, including in respect of any oral representations made or alleged to be made in connection therewith, other than any Retained Claims (as defined in the Equity Commitment Letter). The Guaranteed Party hereby unconditionally and irrevocably covenants and agrees that it shall not institute, and shall cause its Non-Recourse Party not to institute, any Proceeding or bring any other claim (whether at law, in equity, in contract, in tort or otherwise) relating to, arising out of or in connection with, the Purchase Agreement, the Equity Commitment Letter or the transactions contemplated by the Purchase Agreement or the Equity Commitment Letter, or in respect of any oral representations made or alleged to be made in connection therewith or herewith, against any Guarantor or its Non-Recourse Parties, except for claims of the Guaranteed Party against the Guarantors under this Limited Guarantee and Retained Claims (as defined in the Equity Commitment Letter).
- (c) The Guaranteed Party further covenants and agrees that it shall not have the right to recover, and shall not recover, and shall not institute, directly or indirectly, and shall cause its Non-Recourse Parties not to institute, any Proceeding or bring any other claim to recover, more than the Maximum Aggregate Amount in respect of any liabilities or obligations of the Guarantors, their permitted assigns or Purchaser A, or the applicable Maximum Guarantor Amount from each Guarantor and its permitted assigns in respect of any liabilities or obligations of the Guarantors, their permitted assigns or Purchaser A, relating to, arising out of or in connection with the Purchase Agreement, this Limited Guarantee or the transactions contemplated thereby or hereby, and the Guaranteed Party shall promptly return all monies paid to it or its Non-Recourse Parties in excess of the Maximum Aggregate Amount or applicable Maximum Guarantor Amount.
- (d) Nothing set forth in this Limited Guarantee shall confer or give to any Person other than the Guaranteed Party any rights or remedies against any Person, including the Guarantors, except as expressly and specifically set forth herein. The Guaranteed Party acknowledges that each Guarantor is agreeing to enter into this Limited Guarantee in reliance on the provisions set forth in this <u>Section 3</u>. This <u>Section 3</u> shall survive termination of this Limited Guarantee in perpetuity.

- 4. <u>Representations and Warranties</u>. Each Guarantor (on a several, and not joint and several, basis) hereby represents and warrants with respect to itself that:
- (a) this Limited Guarantee is a legal, valid and binding obligation of such Guarantor, enforceable against such Guarantor in accordance with its terms, subject to bankruptcy, insolvency, reorganization and other Legal Requirements of general applicability relating to or affecting creditors' rights and to general equity principles; and
- (b) such Guarantor has unfunded capital commitments in an amount not less than such Guarantor's Maximum Guarantor Amount or has such other financial means at its disposal to enable such Guarantor to pay such Guarantor's Maximum Guarantor Amount when due pursuant to this Limited Guarantee.
- Termination. Notwithstanding anything to the contrary, this Limited Guarantee shall terminate and the Guarantors shall have no further obligation under or in connection with this Limited Guarantee as of the earliest to occur of: (i) the Closing; (ii) the payment of the Guaranteed Obligations; (iii) the thirty (30) day anniversary of the termination of the Purchase Agreement if no claim is brought hereunder in writing alleging such Guaranteed Obligations are due and owing prior to such thirty (30) day anniversary; provided that if such claim is brought, this Limited Guarantee shall not terminate with respect to the Guaranteed Obligations subject to such claim until either (x) a final resolution of such claim by a final, non-appealable judgment of a court of competent jurisdiction and payment of such Guaranteed Obligations, if applicable (or the date of such final resolution if such Guaranteed Obligations are determined by such court not to be payable), or (y) a written agreement signed by the Guaranters and the Guarantee terminating this Limited Guarantee; and (iv) the termination of this Limited Guarantee by mutual written agreement of the Guarantors and the Guaranteed Party. Upon any termination of this Limited Guarantee in accordance with and subject to the first sentence of this Section 5, no Person shall have any rights or claims against any of Purchaser A, the Guarantors or their respective Non-Recourse Parties under the Purchase Agreement, this Limited Guarantee, the Equity Commitment Letter or in respect of any oral representations made or alleged to be made in connection herewith, whether at law or equity, in contract, in tort or otherwise, and none of Purchaser A, the Guarantors or their respective Non-Recourse Parties shall have any further liability or obligation relating to or arising out of the Purchase Agreement, this Limited Guarantee or the Equity Commitment Letter, in respect of the transactions contemplated thereby or hereby or in respect of any oral representations made or alleged to be made in connection herewith or therewith. In the event that the Guaranteed Party or any of its Non-Recourse Parties asserts, directly or indirectly, in any litigation or any other proceeding (whether at law, in equity, in contract, in tort or otherwise) (1) that the provisions of Section 1 hereof limiting the each Guarantors' liability to the Maximum Aggregate Amount or any Guarantor's liability to such Guarantor's Maximum Guarantor Amount, (2) that the provisions of Section 3 hereof are illegal, invalid or unenforceable, in whole or in part, or (3) any theory of liability against any Guarantor or any of its Non-Recourse Parties with respect to the transactions contemplated by the Purchase Agreement (including in respect of any oral representations made or alleged to be made in connection therewith) other than a Retained Claim, (x) the obligations of each Guarantor under this Limited Guarantee shall terminate, ab initio, and shall be null and void, (y) if any Guarantor has previously made any payments under this Limited Guarantee, such Guarantor shall be entitled to recover such payments from the Guaranteed Party, and (z) none of Purchaser A, the

Guaranters nor any of its or their respective Non-Recourse Parties shall have any liability to the Guaranteed Party or any of its Non-Recourse Parties with respect to the transactions contemplated by the Purchase Agreement, the Equity Commitment Letter or this Limited Guarantee (including in respect of any oral representations made or alleged to be made in connection therewith or herewith) any other agreement or instrument delivered in connection with any of the foregoing or the transactions contemplated hereby or thereby.

- 6. <u>Continuing Guarantee</u>. Except to the extent terminated pursuant to the provisions of <u>Section 5</u> of this Limited Guarantee, this Limited Guarantee is a continuing one and shall remain in full force and effect until the payment and satisfaction in full of the Guaranteed Obligations, shall be binding upon the Guarantors, their successors and assigns. All obligations to which this Limited Guarantee applies or may apply under the terms hereof shall be conclusively presumed to have been created in reliance hereon.
- 7. <u>Entire Agreement</u>. This Limited Guarantee constitutes the entire agreement among the parties with respect to the subject matter hereof and supersedes any and all prior discussions, negotiations, proposals, undertakings, understandings and agreements, whether written or oral, among Purchaser A and the Guarantors or any of their Non-Recourse Parties, on the one hand, and the Guaranteed Party or any of its Non-Recourse Parties, on the other hand. Except as provided in this Limited Guarantee, no representation or warranty has been made or relied upon by any of the parties to this Limited Guarantee with respect to this Limited Guarantee or the transactions contemplated hereby.
- 8. Amendments and Waivers. No amendment or waiver of any provision of this Limited Guarantee will be valid and binding unless it is in writing and signed, in the case of an amendment, by each of the Guarantors and the Guaranteed Party or, in the case of a waiver, by the party or each of the parties against whom the waiver is to be effective. No waiver by any party of any breach or violation of, or default under, this Limited Guarantee, whether intentional or not, will be deemed to extend to any prior or subsequent breach, violation or default hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence. No delay or omission on the part of any party in exercising any right, power, or remedy under this Limited Guarantee will operate as a waiver thereof.
- 9. <u>Counterparts</u>. This Limited Guarantee may be executed in any number of counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. This Limited Guarantee will become effective when duly executed by each party hereto. A signature hereto sent or delivered by facsimile or other electronic transmission shall be as legally binding and enforceable as a signed original for all purposes.
- 10. <u>Notices</u>. Any notice or other communication required or permitted to be delivered to any party under this Agreement shall be in writing and shall be deemed properly given and made as follows: (a) if sent by registered or certified mail in the United States, return receipt requested, then such communication shall be deemed duly given and made upon receipt; (b) if sent by nationally recognized overnight air courier for next day delivery, then such communication shall be deemed duly given and made the next Business Day after being sent (or if sent by two day delivery, two (2) Business Days after being sent); (c) if sent by facsimile or email transmission before 5:00 p.m. (addressee's local time) on any Business Day, then such

communication shall be deemed duly given and made when receipt is confirmed; (d) if sent by facsimile or email transmission on a day other than a Business Day and receipt is confirmed, or if sent after 5:00 p.m. (addressee's local time) on any Business Day and receipt is confirmed, then such communication shall be deemed duly given and made on the Business Day following the date on which receipt is confirmed; and (e) if otherwise actually Personally delivered to a duly authorized representative of the recipient, then such communication shall be deemed duly given and made when delivered to such authorized representative; <u>provided</u> that such notices, requests, demands and other communications are delivered to the address set forth below, or to such other address as any party shall provide by like notice to the other parties to this Agreement.

If to any Guarantor, to:

c/o Searchlight Capital Partners, L.P. 745 Fifth Avenue, 27th Floor New York, New York 10151

Attention: Darren Glatt; Nadir Nurmohamed

Email: <u>dglatt@searchlightcap.com</u>; <u>nnurmohamed@searchlightcap.com</u>

with a copy (which shall not constitute notice) to:

Latham & Watkins LLP 885 Third Avenue New York, New York 10022-4834

Attention: David A. Kurzweil; Eyal N. Orgad

Email: david.kurzweil@lw.com; eyal.orgad@lw.com

If to the Guaranteed Party, to:

 $[\bullet]$

with a copy (which shall not constitute notice) to:

[•] [•]

Attention: $[\bullet]$ Fax: $[\bullet]$ Email: $[\bullet]$

or such other address or facsimile number as such party may hereafter specify by like notice to the other parties hereto.

11. <u>No Assignment</u>. This Limited Guarantee and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns. Neither this Limited Guarantee nor any of the rights, interests or obligations hereunder may be assigned (by operation of law or otherwise) or delegated by either the

Guarantors or the Guaranteed Party to any other Person without the prior written consent of the Guaranteed Party (in the case of an assignment by any Guarantor) or each of the Guarantors (in the case of an assignment by the Guaranteed Party) and any purported assignment without such consent shall be null and void and of no force and effect.

- 12. <u>No Third Party Beneficiaries</u>. This Limited Guarantee is not intended to, and does not, confer upon any Person other than the parties hereto any rights or remedies hereunder, except that each Non-Recourse Party of any Guarantor shall be considered a third party beneficiary of the provisions of <u>Section 3</u> and <u>Section 5</u> hereof.
- 13. <u>Severability</u>. Any term or provision of this Limited Guarantee that is invalid or unenforceable in any situation in any jurisdiction will not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction; <u>provided</u>, <u>however</u>, that this Limited Guarantee may not be enforced without giving full and absolute effect to the limitation of the amount payable by the Guarantors hereunder to the Maximum Aggregate Amount and by each Guarantor to its Maximum Guarantor Amount provided in <u>Section 1</u> hereof and to the provisions of <u>Section 3</u> hereof.
- 14. <u>Interpretation</u>. The headings and titles contained in this Limited Guarantee are for convenience purposes only and will not in any way affect the meaning or interpretation hereof. The parties have participated jointly in negotiating and drafting this Limited Guarantee. If an ambiguity or a question of intent or interpretation arises, this Limited Guarantee shall be construed as if drafted jointly by the parties, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Limited Guarantee.
- 15. Governing Law; Forum. This Limited Guarantee shall be governed by, and construed in accordance with, the laws of the State of Israel without giving effect to its conflict of laws principles. Any Proceeding brought with respect to this Limited Guarantee must be brought in any court of competent jurisdiction in Tel Aviv-Jaffa, Israel and, by execution and delivery of this Limited Guarantee, each party (i) accepts, generally and unconditionally, the exclusive jurisdiction of such courts and any related appellate court, and irrevocably agrees to be bound by any judgment rendered thereby in connection with this Limited Guarantee and (ii) irrevocably waives any objection it may now or hereafter have as to the venue of any such suit, action or Proceeding brought in such a court or that such court is an inconvenient forum

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, the undersigned have duly executed and delivered this Limited Guarantee as of the date first set forth above.

Guarantors:

Searchlight Capital II, L.P. By: Searchlight Capital Partners II GP, L.P., its general partner By: Searchlight Capital Partners II GP, LLC, its general partner		
By: Name: Darren Glatt Title: Authorized Person		
Searchlight Capital II PV, L.P. By: Searchlight Capital Partners II GP, L.P., its general partner By: Searchlight Capital Partners II GP, LLC, its general partner		
By: Name: Darren Glatt Title: Authorized Person		

Accepted and Agreed,

INTERNET GOLD - GOLDEN LINES LTD.

By:	
Name:	
Title:	
B COMMUNICATIONS LTD.	
D	
By:	
Name:	
Title:	

Exhibit A

Guarantor	Maximum Guarantor Amount (% of Maximum Aggregate Amount)
Searchlight Capital II, L.P.	NIS 12,981,357 (49.737%)
Searchlight Capital II PV, L.P.	NIS 13,118,643 (50.263%)
Total	NIS 26,100,000 (100.000%)

LIMITED GUARANTEE

Limited Guarantee, dated as of June 24, 2019 (this "Limited Guarantee"), by the party listed on Exhibit A hereto (the "Guarantor"), an entity fully owned by the Fuhrer family, in favor of Internet Gold - Golden Lines Ltd. and B Communications Ltd. (collectively, the "Guaranteed Party"). Reference is made to that certain Share Purchase Agreement, dated as of even date herewith (the "Purchase Agreement"), by and among Searchlight II BZQ, L.P., a Cayman Islands exempt limited partnership ("Purchaser A"), T.N.R. Investments LTD., a limited liability company organized and existing under the laws of the State of Israel ("Purchaser B"), and the Guaranteed Party. Except as otherwise specified herein, capitalized terms used herein but not otherwise defined have the respective meanings ascribed to them in the Purchase Agreement.

Limited Guarantee. To induce the Guaranteed Party to enter into the Purchase Agreement, The Guarantor hereby guarantees to the Guaranteed Party, on the terms and subject to the conditions set forth herein, payment of the payment obligations of Purchaser B under the Purchase Agreement, solely to the extent exist and payable following the valid termination of the Purchase Agreement, in each case, on the terms and subject to the conditions set forth in the Purchase Agreement (including Section 8.2(b) thereof) and herein (the "Guaranteed Obligations"), in an amount equal to the Maximum Aggregate Amount (as defined below) set on Exhibit A hereto (The "Maximum Guarantor Amount"); provided that (a) the maximum liability of the Guarantor hereunder shall not exceed the Guarantor's Maximum Guarantor Amount, (b) the maximum aggregate liability of the Guarantor hereunder shall not exceed three million and nine hundred thousand New Israeli Shekels (NIS 3,900,000) (the "Maximum Aggregate Amount"). It being understood and agreed that this Limited Guarantee may not be enforced without giving full and absolute effect to the Maximum Aggregate Amount. The Guaranteed Party hereby agrees that the Guarantor shall in no event be required to pay to any Person or Persons in the aggregate more than the Maximum Aggregate Amount under, in respect of, or in connection with this Limited Guarantee or the Purchase Agreement, and that the Guarantor shall not have any obligation or liability to any Person under, in respect of or in connection with this Limited Guarantee or the Purchase Agreement other than under this Guarantee as specifically and expressly set forth herein. The guarantee by the Guaranter of the Guaranteed Obligations under this Limited Guarantee may be enforced for the payment of money only. Notwithstanding anything to the contrary contained in this Limited Guarantee, in the Purchase Agreement or otherwise, the Guaranteed Party hereby agrees that to the extent Purchaser is relieved of all or any portion of its payment obligations under the Purchase Agreement by satisfaction thereof or pursuant to any other agreement with the Guaranteed Party, the Guarantor shall be similarly relieved, to such extent, of its respective obligations under this Limited Guarantee.

2. Terms of Limited Guarantee; Recovery Claim.

(a) This Limited Guarantee is one of payment, not collection, and a separate action or actions may be brought and prosecuted against the Guarantor to enforce this Limited Guarantee (solely in accordance with and subject to its terms, including the limitations contained herein), irrespective of whether any action is brought against Purchaser B or whether Purchaser B is joined in any such action or actions. The Guarantor reserves the right, notwithstanding anything to the contrary provided herein or otherwise, to (i) set off any amount owed hereunder by it against any payment owing by the Guaranteed Party to Purchaser B or the Guarantor, and (ii) assert any

and all defenses which the Guarantor, Purchaser B or any of their affiliates may have against payment of the Guaranteed Obligations.

- (b) The liability of the Guarantor under this Limited Guarantee shall, to the fullest extent permitted under applicable Legal Requirements, be absolute and unconditional, irrespective of:
 - (i) any change in the corporate existence, structure or ownership of Purchaser B or any insolvency, bankruptcy, reorganization, liquidation or other similar proceeding of Purchaser B;
 - (ii) any change in the manner, place or terms of payment or performance, or any change or extension of the time of payment or performance of, renewal or alteration of, the Guaranteed Obligations, or any liability incurred directly or indirectly in respect thereof; or
 - (iii) the existence of any claim, set-off or other right that the Guarantors may have at any time against Purchaser B, whether in connection with any of the Guaranteed Obligations or otherwise.

Notwithstanding the foregoing or anything to the contrary in this Limited Guarantee, the Guarantor shall be fully released and discharged hereunder if the Guaranteed Obligations are paid in full by Purchaser B or any other Person.

3. Sole Remedy.

- (a) The Guaranteed Party acknowledges and agrees that:
- (i) the sole cash assets of Purchaser B are cash in a *de minimis* amount (if any), and that no additional funds are expected to be contributed to Purchaser B unless and until the Closing occurs;
- (ii) that the Guaranteed Party is bound by and shall comply with the applicable terms and conditions of Sections 2, 3, 6 and 6 of the Equity Commitment Letter, dated as of the date hereof, by and between Purchaser B and the Guarantor (the "Equity Commitment Letter");
- (iii) the Guarantor shall not have any obligation or liability to any Person relating to, arising out of or in connection with the Purchase Agreement, this Limited Guarantee, the Equity Commitment Letter or the transactions contemplated thereby or hereby, other than as expressly set forth herein or in the Equity Commitment Letter; and
- (iv) notwithstanding anything to the contrary in this Limited Guarantee, the Equity Commitment Letter, the Purchase Agreement or otherwise, it has no and shall have no right of recovery against the Guarantor or any of the Non-Recourse Parties (as defined in the Equity Commitment Letter), through any Guarantor, Purchaser B or otherwise, whether by or through attempted piercing of the corporate, limited liability company or limited partnership veil, by or through a claim by or on behalf of Purchaser B

against the Guarantor or any of the Non-Recourse Parties, or otherwise, except for its rights against the Guarantor under this Limited Guarantee pursuant to the terms and subject to the conditions hereof and except for the Retained Claims (as defined in the Equity Commitment Letter).

- Recourse against the Guarantor under this Limited Guarantee shall be the (b) sole and exclusive remedy (whether at law, in equity, in contract, in tort or otherwise) of the Guaranteed Party and its Non-Recourse Parties against the Guarantor and its respective Non-Recourse Parties in respect of any breaches, Losses or other damages relating to, arising out of or in connection with, the Purchase Agreement or the transactions contemplated thereby, including in respect of any oral representations made or alleged to be made in connection therewith, other than any Retained Claims (as defined in the Equity Commitment Letter). The Guaranteed Party hereby unconditionally and irrevocably covenants and agrees that it shall not institute, and shall cause its Non-Recourse Party not to institute, any Proceeding or bring any other claim (whether at law, in equity, in contract, in tort or otherwise) relating to, arising out of or in connection with, the Purchase Agreement, the Equity Commitment Letter or the transactions contemplated by the Purchase Agreement or the Equity Commitment Letter, or in respect of any oral representations made or alleged to be made in connection therewith or herewith, against the Guarantor or its Non-Recourse Parties, except for claims of the Guaranteed Party against the Guarantor under this Limited Guarantee and Retained Claims (as defined in the Equity Commitment Letter).
- (c) The Guaranteed Party further covenants and agrees that it shall not have the right to recover, and shall not recover, and shall not institute, directly or indirectly, and shall cause its Non-Recourse Parties not to institute, any Proceeding or bring any other claim to recover, more than the Maximum Aggregate Amount in respect of any liabilities or obligations of the Guarantor, their permitted assigns or Purchaser B, or the applicable Maximum Guarantor Amount from the Guarantor and its permitted assigns in respect of any liabilities or obligations of the Guarantor, its permitted assigns or Purchaser B, relating to, arising out of or in connection with the Purchase Agreement, this Limited Guarantee or the transactions contemplated thereby or hereby, and the Guaranteed Party shall promptly return all monies paid to it or its Non-Recourse Parties in excess of the Maximum Aggregate Amount or applicable Maximum Guarantor Amount.
- (d) Nothing set forth in this Limited Guarantee shall confer or give to any Person other than the Guaranteed Party any rights or remedies against any Person, including the Guarantor, except as expressly and specifically set forth herein. The Guaranteed Party acknowledges that the Guarantor is agreeing to enter into this Limited Guarantee in reliance on the provisions set forth in this Section 3. This Section 3 shall survive termination of this Limited Guarantee in perpetuity.
- 4. <u>Representations and Warranties</u>. The Guarantor (on a several, and not joint and several, basis) hereby represents and warrants with respect to itself that:
- (a) this Limited Guarantee is a legal, valid and binding obligation of such Guarantor, enforceable against such Guarantor in accordance with its terms, subject to bankruptcy, insolvency, reorganization and other Legal Requirements of general applicability relating to or affecting creditors' rights and to general equity principles; and

- (b) such Guarantor has unfunded capital commitments in an amount not less than such Guarantor's Maximum Guarantor Amount or has such other financial means at its disposal to enable such Guarantor to pay such Guarantor's Maximum Guarantor Amount when due pursuant to this Limited Guarantee.
- <u>Termination</u>. Notwithstanding anything to the contrary, this Limited Guarantee shall terminate and the Guarantor shall have no further obligation under or in connection with this Limited Guarantee as of the earliest to occur of: (i) the Closing; (ii) the payment of the Guaranteed Obligations; (iii) the thirty (30) day anniversary of the termination of the Purchase Agreement if no claim is brought hereunder in writing alleging such Guaranteed Obligations are due and owing prior to such thirty (30) day anniversary; provided that if such claim is brought, this Limited Guarantee shall not terminate with respect to the Guaranteed Obligations subject to such claim until either (x) a final resolution of such claim by a final, non-appealable judgment of a court of competent jurisdiction and payment of such Guaranteed Obligations, if applicable (or the date of such final resolution if such Guaranteed Obligations are determined by such court not to be payable), or (y) a written agreement signed by the Guarantor and the Guarantee terminating this Limited Guarantee; and (iv) the termination of this Limited Guarantee by mutual written agreement of the Guarantor and the Guaranteed Party. Upon any termination of this Limited Guarantee in accordance with and subject to the first sentence of this Section 5, no Person shall have any rights or claims against any of Purchaser B, the Guarantor or their respective Non-Recourse Parties under the Purchase Agreement, this Limited Guarantee, the Equity Commitment Letter or in respect of any oral representations made or alleged to be made in connection herewith, whether at law or equity, in contract, in tort or otherwise, and none of Purchaser B, the Guarantor or their respective Non-Recourse Parties shall have any further liability or obligation relating to or arising out of the Purchase Agreement, this Limited Guarantee or the Equity Commitment Letter, in respect of the transactions contemplated thereby or hereby or in respect of any oral representations made or alleged to be made in connection herewith or therewith. In the event that the Guaranteed Party or any of its Non-Recourse Parties asserts, directly or indirectly, in any litigation or any other proceeding (whether at law, in equity, in contract, in tort or otherwise) (1) that the provisions of Section 1 hereof limiting the Guarantor's liability to the Maximum Aggregate Amount or any Guarantor's liability to such Guarantor's Maximum Guarantor Amount, (2) that the provisions of Section 3 hereof are illegal, invalid or unenforceable, in whole or in part, or (3) any theory of liability against the Guarantor or any of its Non-Recourse Parties with respect to the transactions contemplated by the Purchase Agreement (including in respect of any oral representations made or alleged to be made in connection therewith) other than a Retained Claim, (x) the obligations of the Guarantor under this Limited Guarantee shall terminate, ab initio, and shall be null and void, (y) if the Guarantor has previously made any payments under this Limited Guarantee, such Guarantor shall be entitled to recover such payments from the Guaranteed Party, and (z) none of Purchaser B, the Guarantor nor any of its or their respective Non-Recourse Parties shall have any liability to the Guaranteed Party or any of its Non-Recourse Parties with respect to the transactions contemplated by the Purchase Agreement, the Equity Commitment Letter or this Limited Guarantee (including in respect of any oral representations made or alleged to be made in connection therewith or herewith) any other agreement or instrument delivered in connection with any of the foregoing or the transactions contemplated hereby or thereby.
- 6. <u>Continuing Guarantee</u>. Except to the extent terminated pursuant to the provisions of <u>Section 5</u> of this Limited Guarantee, this Limited Guarantee is a continuing one and shall remain

in full force and effect until the payment and satisfaction in full of the Guaranteed Obligations, shall be binding upon the Guarantor, its successors and assigns. All obligations to which this Limited Guarantee applies or may apply under the terms hereof shall be conclusively presumed to have been created in reliance hereon.

- 7. Entire Agreement. This Limited Guarantee constitutes the entire agreement among the parties with respect to the subject matter hereof and supersedes any and all prior discussions, negotiations, proposals, undertakings, understandings and agreements, whether written or oral, among Purchaser B and the Guaranter or any of their Non-Recourse Parties, on the one hand, and the Guaranteed Party or any of its Non-Recourse Parties, on the other hand. Except as provided in this Limited Guarantee, no representation or warranty has been made or relied upon by any of the parties to this Limited Guarantee with respect to this Limited Guarantee or the transactions contemplated hereby.
- 8. Amendments and Waivers. No amendment or waiver of any provision of this Limited Guarantee will be valid and binding unless it is in writing and signed, in the case of an amendment, by the Guarantor and the Guaranteed Party or, in the case of a waiver, by the party or each of the parties against whom the waiver is to be effective. No waiver by any party of any breach or violation of, or default under, this Limited Guarantee, whether intentional or not, will be deemed to extend to any prior or subsequent breach, violation or default hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence. No delay or omission on the part of any party in exercising any right, power, or remedy under this Limited Guarantee will operate as a waiver thereof.
- 9. <u>Counterparts</u>. This Limited Guarantee may be executed in any number of counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. This Limited Guarantee will become effective when duly executed by each party hereto. A signature hereto sent or delivered by facsimile or other electronic transmission shall be as legally binding and enforceable as a signed original for all purposes.
- Notices. Any notice or other communication required or permitted to be delivered to any party under this Agreement shall be in writing and shall be deemed properly given and made as follows: (a) if sent by registered or certified mail in the United States, return receipt requested, then such communication shall be deemed duly given and made upon receipt; (b) if sent by nationally recognized overnight air courier for next day delivery, then such communication shall be deemed duly given and made the next Business Day after being sent (or if sent by two day delivery, two (2) Business Days after being sent); (c) if sent by facsimile or email transmission before 5:00 p.m. (addressee's local time) on any Business Day, then such communication shall be deemed duly given and made when receipt is confirmed; (d) if sent by facsimile or email transmission on a day other than a Business Day and receipt is confirmed, or if sent after 5:00 p.m. (addressee's local time) on any Business Day and receipt is confirmed, then such communication shall be deemed duly given and made on the Business Day following the date on which receipt is confirmed; and (e) if otherwise actually Personally delivered to a duly authorized representative of the recipient, then such communication shall be deemed duly given and made when delivered to such authorized representative; provided that such notices, requests, demands and other communications are delivered to the address set forth below, or to such other address as any party shall provide by like notice to the other parties to this Agreement.

If to Guarantor, to:

c/o T.N.R. Real Estate Properties Ltd. HaShiloah Street 6, Petah Tiqva 4951439, Israel

Attention: Tal Fuhrer

Email: Tal.F@NeopharmGroup.com

with a copy (which shall not constitute notice) to:

Herzog Fox & Neeman Asia House, 4 Weizmann St., Tel Aviv 6423904, Israel Attention: Niv Sivan E-mail: sivann@hfn.co.il

If to the Guaranteed Party, to:

 $[\bullet]$

with a copy (which shall not constitute notice) to:

[●] [●] [●]

Attention: $[\bullet]$ Fax: $[\bullet]$ Email: $[\bullet]$

or such other address or facsimile number as such party may hereafter specify by like notice to the other parties hereto.

- 11. <u>No Assignment</u>. This Limited Guarantee and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns. Neither this Limited Guarantee nor any of the rights, interests or obligations hereunder may be assigned (by operation of law or otherwise) or delegated by either the Guarantor or the Guaranteed Party to any other Person without the prior written consent of the Guaranteed Party (in the case of an assignment by the Guarantor) or the Guarantor (in the case of an assignment by the Guaranteed Party) and any purported assignment without such consent shall be null and void and of no force and effect.
- 12. <u>No Third Party Beneficiaries</u>. This Limited Guarantee is not intended to, and does not, confer upon any Person other than the parties hereto any rights or remedies hereunder, except that each Non-Recourse Party of the Guarantor shall be considered a third party beneficiary of the provisions of Section 3 and Section 5 hereof.

- 13. <u>Severability</u>. Any term or provision of this Limited Guarantee that is invalid or unenforceable in any situation in any jurisdiction will not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction; <u>provided</u>, <u>however</u>, that this Limited Guarantee may not be enforced without giving full and absolute effect to the limitation of the amount payable by the Guarantor hereunder to the Maximum Aggregate Amount and by the Guarantor to its Maximum Guarantor Amount provided in <u>Section 1</u> hereof and to the provisions of Section 3 hereof.
- 14. <u>Interpretation</u>. The headings and titles contained in this Limited Guarantee are for convenience purposes only and will not in any way affect the meaning or interpretation hereof. The parties have participated jointly in negotiating and drafting this Limited Guarantee. If an ambiguity or a question of intent or interpretation arises, this Limited Guarantee shall be construed as if drafted jointly by the parties, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Limited Guarantee.
- 15. Governing Law; Forum. This Limited Guarantee shall be governed by, and construed in accordance with, the laws of the State of Israel without giving effect to its conflict of laws principles. Any Proceeding brought with respect to this Limited Guarantee must be brought in any court of competent jurisdiction in Tel Aviv-Jaffa, Israel and, by execution and delivery of this Limited Guarantee, each party (i) accepts, generally and unconditionally, the exclusive jurisdiction of such courts and any related appellate court, and irrevocably agrees to be bound by any judgment rendered thereby in connection with this Limited Guarantee and (ii) irrevocably waives any objection it may now or hereafter have as to the venue of any such suit, action or Proceeding brought in such a court or that such court is an inconvenient forum

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, the undersigned have duly executed and delivered this Limited Guarantee as of the date first set forth above.

Guarantor:
T.N.R. REAL ESTATE PROPERTIES LTD.
By: Name: David Fuhrer Title: Chairman

Accepted and Agreed,

INTERNET GOLD - GOLDEN LINES LTD.

By:	
Nam	ie:
Title	:
в сом	MUNICATIONS LTD.
D	
By:	
Nam	ie:
Title	:

Exhibit A

Guarantor	Maximum Guarantor Amount
T.N.R. REAL ESTATE PROPERTIES LTD	NIS 3,900,000

ANNUAL AND EXTRAORDINARY GENERAL MEETING OF SHAREHOLDERS OF

INTERNET GOLD-GOLDEN LINES LTD.

August 8, 2019

GO GREEN

e-Consent makes it easy to go paperless. With e-Consent, you can quickly access your proxy material, statements and other eligible documents online, while reducing costs, clutter and paper waste. Enroll today via www.astfinancial.com to enjoy online access.

Please sign, date and mail your proxy card in the envelope provided as soon as possible.

Please detach along perforated line and mail in the envelope provided.

00003330300030200000 0

080819

	THE ELECTION OF THE DIRECTORS IN ITEM 1 AND ITEMS 2, 3 AND 4. VELOPE. PLEASE MARK YOUR VOTE IN BLUE OR BLACK INK AS SHOWN HERE X
	To elect two directors for terms expiring at the Company's 2020 Annual General Meeting of Shareholders. MOSHE LUSKY FOR AGAINST ABSTAIN TO elect two directors for terms expiring at the Company's 2020 Annual FOR AGAINST ABSTAIN TO elect two directors for terms expiring at the Company's 2020 Annual FOR AGAINST ABSTAIN TO elect two directors for terms expiring at the Company's 2020 Annual FOR AGAINST ABSTAIN TO elect two directors for terms expiring at the Company's 2020 Annual FOR AGAINST ABSTAIN MOSHE LUSKY
	YAHEL SHACHAR
	To authorize the Company's Board of Directors to effect a reverse share split of the Company's ordinary shares at a ratio not to exceed one-for-one hundred and to approve related amendments to the Company's Memorandum and Articles of Association.
	3. To ratify and approve the reappointment of Somekh Chaikin, registered public accounting firm, a member of KPMG International, as the Company's independent registered public accountants for the year ending December 31, 2019, and to authorize the Company's Board of Directors and its Audit Committee (under their authority in accordance with the Israeli Companies Law), to fix the compensation of such independent registered public accountants in accordance with the volume and nature of their services.
	To approve an arrangement pursuant to Section 350 of the Israeli Companies Law.
	5. Do your holdings in the Company or vote on the Proposals above require the approval of the Prime Minister of Israel and Israeli Minister of Communications pursuant to the Communications Law and Communications Order, as defined in the Proxy Statement?
To change the address on your account, please check the box at right and indicate your new address in the address space above. Please note that changes to the registered name(s) on the account may not be submitted via this method.	VOTES WILL NOT BE COUNTED FOR ANY PROPOSAL UNLESS "YES" OR "NO" HAS BEEN SPECIFIED IN ITEM 5 AS TO WHETHER THE HOLDINGS IN THE COMPANY OR THE VOTE REQUIRES THE APPROVAL OF THE PRIME MINISTER OF ISRAEL AND ISRAELI MINISTER OF COMMUNICATIONS (AS DESCRIBED IN THE PROXY STATEMENT).
Signature of Shareholder Date:	Signature of Shareholder Date:
Note: Please sign exactly as your name or names appear on this Proxy. When shares are held joint title as such. If the signer is a corporation, please sign full corporate name by duly authorized	ly, each holder should sign. When signing as executor, administrator, attorney, trustee or guardian, please give full officer, giving full title as such. If signer is a partnership, please sign in partnership name by authorized person.

INTERNET GOLD-GOLDEN LINES LTD.

THIS PROXY IS SOLICITED ON BEHALF OF THE BOARD OF DIRECTORS

The undersigned hereby appoint(s) Doron Turgeman and Ran Nakel, attorneys or attorney of the undersigned, for and in the name(s) of the undersigned, with power of substitution and revocation in each to vote any and all ordinary shares, par value NIS 0.01 per share, of Internet Gold-Golden Lines Ltd. (the "Company"), which the undersigned would be entitled to vote as fully as the undersigned could if personally present at the Annual and Extraordinary General Meeting of Shareholders of the Company to be held on August 8, 2019 at 10:30 a.m. (Israel time) at the offices of the Company, 2 Dov Friedman Street, Ramat Gan 5250301, Israel and at any adjournment or adjournments thereof, and hereby revoking any prior proxies to vote said shares, upon the following items of business more fully described in the Notice of and Proxy Statement for such Annual General Meeting, or the Proxy Statement, (receipt of which is hereby acknowledged):

THIS PROXY WILL BE VOTED AS SPECIFIED ON THE REVERSE. IN THE ABSENCE OF SUCH SPECIFICATION, THE SHARES REPRESENTED BY THIS PROXY CARD WILL BE VOTED FOR THE ELECTION OF THE DIRECTORS IN ITEM 1 AND ITEMS 2, 3 AND 4 SET FORTH ON THE REVERSE. ON ANY OTHER BUSINESS THAT MAY PROPERLY COME BEFORE THE MEETING, THIS PROXY WILL BE VOTED IN ACCORDANCE WITH THE JUDGMENT OF THE PERSONS NAMED ABOVE AS PROXIES.

VOTES FOR ANY PROPOSAL WILL NOT BE COUNTED UNLESS "YES" OR "NO" HAS BEEN SPECIFIED IN ITEM 5 AS TO WHETHER THE HOLDINGS IN THE COMPANY OR THE VOTE REQUIRES THE APPROVAL OF THE PRIME MINISTER OF ISRAEL AND ISRAELI MINISTER OF COMMUNICATIONS (AS DESCRIBED IN THE PROXY STATEMENT).